UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

		FORM 10-Q
×	QUARTERLY REPORT PURSUANT TO SECTION 13 OR 1	.5(d) OF THE SECURITIES EXCHANGE ACT OF 1934
	For the quarterly period ended June 30, 2008	
		OR
	TRANSITION REPORT PURSUANT TO SECTION 13 OR 1	15(d) OF THE SECURITIES EXCHANGE ACT OF 1934
	For the transition period from to	
	Commiss	sion file number: 1-13782
		AKE TECHNOLOGIES CORPORATION registrant as specified in its charter)
	Delaware (State or other jurisdiction of incorporation or organization)	25-1615902 (I.R.S. Employer Identification No.)
	1001 Air Brake Avenue Wilmerding, PA (Address of principal executive offices)	15148 (Zip Code)
	(Registrant's te	412-825-1000 lephone number, including area code)
	(Former name, former address	N/A s and former fiscal year, if changed since last report)
	Indicate by check mark whether the registrant (1) has filed all repo	orts required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 trant was required to file such reports), and (2) has been subject to such filing
the c	Indicate by check mark whether the registrant is a large accelerate lefinitions of "large accelerated filer," "accelerated filer" and "small	ed filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See er reporting company" in Rule 12b-2 of the Exchange Act.
Larg	e accelerated filer $oxtimes$ Accelerated filer $oxtimes$ Non-accelerated file	er Smaller reporting company
	Indicate by check mark whether the registrant is a shell company	(as defined in Rule 12b-2 of the Exchange Act). Yes \square No \boxtimes
	Indicate the number of shares outstanding of each of the issuer's c	classes of common stock, as of the latest practicable date.
	<u>Class</u> Common Stock, \$.01 par value per share	Outstanding at August 4, 2008 48,434,741 shares

WESTINGHOUSE AIR BRAKE TECHNOLOGIES CORPORATION

June 30, 2008 FORM 10-Q

TABLE OF CONTENTS

	PART I—FINANCIAL INFORMATION	Page
Item 1.	Financial Statements	
	Condensed Consolidated Balance Sheets as of June 30, 2008 and December 31, 2007	3
	Condensed Consolidated Statements of Operations for the three and six months ended June 30, 2008 and 2007	4
	Condensed Consolidated Statements of Cash Flows for the six months ended June 30, 2008 and 2007	5
	Notes to Condensed Consolidated Financial Statements	6
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	29
Item 3.	Quantitative and Qualitative Disclosures about Market Risk	38
Item 4.	Controls and Procedures	39
	PART II—OTHER INFORMATION	
Item 1.	<u>Legal Proceedings</u>	40
Item 1A.	Risk Factors	40
Item 2.	<u>Unregistered Sales of Equity Securities and Use of Proceeds</u>	40
Item 4.	Submission of Matters to a Vote of Security Holders	40
Item 6.	<u>Exhibits</u>	41
	<u>Signatures</u>	42

PART I—FINANCIAL INFORMATION

Item 1. Financial Statements

WESTINGHOUSE AIR BRAKE TECHNOLOGIES CORPORATION CONDENSED CONSOLIDATED BALANCE SHEETS

In thousands, except shares and par value	Unaudited June 30, 2008	December 31, 2007
Assets		
Current Assets		
Cash and cash equivalents	\$ 241,680	\$ 234,689
Accounts receivable	284,145	222,235
Inventories	196,331	175,977
Deferred income taxes	24,471	24,766
Other current assets	10,857	8,100
Total current assets	757,484	665,767
Property, plant and equipment	416,857	417,157
Accumulated depreciation	(237,701)	(234,720)
Property, plant and equipment, net	179,156	182,437
Other Assets		
Goodwill	237,842	232,593
Other intangibles, net	57,710	58,673
Deferred income taxes	4,536	4,316
Other noncurrent assets	19,372	14,916
Total other assets	319,460	310,498
Total Assets	\$ 1,256,100	\$ 1,158,702
Liabilities and Shareholders' Equity		
Current Liabilities		
Accounts payable	\$ 142,925	\$ 137,226
Accrued income taxes	1,255	3,625
Customer deposits	94,761	67,291
Accrued compensation	30,400	30,519
Accrued warranty	16,399	14,390
Other accrued liabilities	39,484	42,184
Total current liabilities	325,224	295,235
Long-term debt	150,135	150,177
Reserve for postretirement and pension benefits	54,375	53,539
Deferred income taxes	10,781	9,834
Accrued warranty	10,362	7,924
Other long term liabilities	26,572	24,725
Total liabilities	577,449	541,434
Shareholders' Equity		
Preferred stock, 1,000,000 shares authorized, no shares issued	_	_
Common stock, \$.01 par value; 100,000,000 shares authorized: 66,174,767 shares issued and 48,426,541 and 48,698,344		
outstanding at June 30, 2008 and December 31, 2007, respectively	662	662
Additional paid-in capital	322,488	320,928
Treasury stock, at cost, 17,748,226 and 17,476,423 shares, at June 30, 2008 and December 31, 2007, respectively	(256,455)	(238,131)
Retained earnings	589,769	524,538
Accumulated other comprehensive income	22,187	9,271
Total shareholders' equity	678,651	617,268
Total Liabilities and Shareholders' Equity	\$ 1,256,100	\$ 1,158,702

The accompanying notes are an integral part of these statements.

WESTINGHOUSE AIR BRAKE TECHNOLOGIES CORPORATION CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS

	Unau Three Mon Jun	ths Ended	Unau Six Montl June	ns Ended
In thousands, except per share data	2008	2007	2008	2007
Net sales	\$ 390,194	\$ 325,722	\$ 773,521	\$ 639,986
Cost of sales	(281,693)	(234,872)	(559,805)	(462,570)
Gross profit	108,501	90,850	213,716	177,416
Selling, general and administrative expense	(42,036)	(34,915)	(82,481)	(69,860)
Engineering expense	(9,631)	(9,026)	(19,612)	(17,842)
Amortization expense	(912)	(992)	(1,815)	(1,533)
Total operating expenses	(52,579)	(44,933)	(103,908)	(89,235)
Income from operations	55,922	45,917	109,808	88,181
Other income and expenses				
Interest expense, net	(1,293)	(686)	(2,774)	(1,469)
Other expense, net	(696)	(1,637)	(1,079)	(2,446)
Income from continuing operations before income taxes	53,933	43,594	105,955	84,266
Income tax expense	(20,171)	(15,469)	(39,680)	(30,587)
Income from continuing operations	33,762	28,125	66,275	53,679
Discontinued operations				
Income (loss) from discontinued operations (net of tax)		5	(3)	(27)
Net income	\$ 33,762	\$ 28,130	\$ 66,272	\$ 53,652
Earnings Per Common Share				
Basic				
Income from continuing operations	\$ 0.70	\$ 0.58	\$ 1.37	\$ 1.11
Loss from discontinued operations	_	_	_	_
Net income	\$ 0.70	\$ 0.58	\$ 1.37	\$ 1.11
Diluted				
Income from continuing operations	\$ 0.69	\$ 0.57	\$ 1.35	\$ 1.09
Loss from discontinued operations	_	_	_	_
Net income	\$ 0.69	\$ 0.57	\$ 1.35	\$ 1.09
Weighted average shares outstanding				
Basic	48,095	48,666	48,306	48,413
Diluted	48,655	49,294	48,924	49,022

The accompanying notes are an integral part of these statements.

WESTINGHOUSE AIR BRAKE TECHNOLOGIES CORPORATION CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS

	Unaudited Six Months Ended June 30,	
<u>In thousands</u>	2008	2007
Operating Activities	_	
Net income	\$ 66,272	\$ 53,652
Stock-based compensation expense	5,024	4,645
Adjustments to reconcile net income to net cash provided by operations:	(0.0)	
Discontinued operations	(38)	114
Depreciation and amortization	14,468	13,676
Excess income tax benefits from exercise of stock options	(1,188)	(1,402)
Changes in operating assets and liabilities		
Accounts receivable	(58,422)	(12,265)
Inventories	(18,873)	(22,067)
Accounts payable	4,360	7,639
Accrued income taxes	(459)	6,181
Accrued liabilities and customer deposits	26,781	(21,379)
Other assets and liabilities	(7,748)	(2,571)
Net cash provided by operating activities	30,177	26,223
Investing Activities		
Purchase of property, plant and equipment and other	(7,852)	(8,342)
Proceeds from disposal of property, plant and equipment	64	139
Acquisitions of business, net of cash acquired		(73,264)
Net cash used for investing activities	(7,788)	(81,467)
Financing Activities		
Repayments of other borrowings	(47)	
Proceeds from the issuance of treasury stock for stock options and other benefit plans	1,677	4,310
Stock repurchase	(24,653)	(3,421)
Excess income tax benefits from exercise of stock options	1,188	1,402
Cash dividends (\$0.02 per share for the six months ended June 30, 2008 and 2007)	(969)	(986)
Net cash (used for) provided by financing activities	(22,804)	1,305
Effect of changes in currency exchange rates	7,406	7,186
Increase (decrease) in cash	6,991	(46,753)
Cash, beginning of year	234,689	187,979
Cash, end of period	\$241,680	\$141,226

The accompanying notes are an integral part of these statements.

WESTINGHOUSE AIR BRAKE TECHNOLOGIES CORPORATION NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS FOR THE QUARTERLY PERIOD ENDED JUNE 30, 2008 (UNAUDITED)

1. BUSINESS

Wabtec is one of the world's largest providers of value-added, technology-based products and services for the global rail industry. Our products are found on virtually all U.S. locomotives, freight cars and passenger transit vehicles, as well as in more than 100 countries throughout the world. Our products enhance safety, improve productivity and reduce maintenance costs for customers, and many of our core products and services are essential in the safe and efficient operation of freight rail and passenger transit vehicles. Wabtec is a global company with operations in 12 countries. In the first six months of 2008, about 42% of the Company's revenues came from customers outside the U.S.

2. ACCOUNTING POLICIES

Basis of Presentation The unaudited condensed consolidated interim financial statements have been prepared in accordance with generally accepted accounting principles and the rules and regulations of the Securities and Exchange Commission and include the accounts of Wabtec and its majority owned subsidiaries. These condensed interim financial statements do not include all of the information and footnotes required for complete financial statements. In Management's opinion, these financial statements reflect all adjustments of a normal, recurring nature necessary for a fair presentation of the results for the interim periods presented. Results for these interim periods are not necessarily indicative of results to be expected for the full year.

The Company operates on a four-four-five week accounting quarter, and accordingly, the quarters end on or about March 31, June 30, September 30 and December 31. The acquisition of POLI S.p.A. ("POLI") occurred on June 30, 2008, subsequent to the close of our four-four-five week accounting quarter.

The notes included herein should be read in conjunction with the audited consolidated financial statements included in Wabtec's Annual Report on Form 10-K for the year ended December 31, 2007. The December 31, 2007 information has been derived from the Company's December 31, 2007 Annual Report on Form 10-K.

Revenue Recognition Revenue is recognized in accordance with Staff Accounting Bulletins (SABs) 101, "Revenue Recognition in Financial Statements" and 104 "Revision of Topic 13." Revenue is recognized when products have been shipped to the respective customers, title has passed and the price for the product has been determined.

The Company recognizes revenues on long-term contracts based on the percentage of completion method of accounting. The units-of-delivery method or other input-based or output-based measures, as appropriate, are used to measure the progress toward completion of individual contracts. Contract revenues and cost estimates are reviewed and revised at a minimum quarterly and adjustments are reflected in the accounting period as such amounts are determined. Provisions are made currently for estimated losses on uncompleted contracts.

Certain pre-production costs relating to long-term production and supply contracts have been deferred and will be recognized over the life of the contracts. Deferred pre-production costs were \$10.2 million and \$9.5 million at June 30, 2008 and December 31, 2007, respectively.

Use of Estimates The preparation of financial statements in conformity with generally accepted accounting principles in the United States requires the Company to make estimates and assumptions that affect the reported amounts of assets and liabilities at the date of the financial statements and reported amounts of revenues and expenses during the reporting period. Actual amounts could differ from the estimates. On an ongoing basis, management reviews its estimates based on currently available information. Changes in facts and circumstances may result in revised estimates.

WESTINGHOUSE AIR BRAKE TECHNOLOGIES CORPORATION

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS—(Continued) FOR THE QUARTERLY PERIOD ENDED JUNE 30, 2008 (UNAUDITED)

Stock-Based Compensation The Company recognizes compensation expense for stock-based compensation based on the grant date fair value ratably over the requisite service period following the date of grant.

Financial Derivatives and Hedging Activities The Company has entered into foreign currency forward contracts to reduce the impact of changes in currency exchange rates. Forward contracts are agreements with a counterparty to exchange two distinct currencies at a set exchange rate for delivery on a set date at some point in the future. There is no exchange of funds until the delivery date. At the delivery date the Company can either take delivery of the currency or settle on a net basis.

At June 30, 2008, the Company had forward contracts for the sale of USD and the purchase of Euro with a notional value of €1.2 million Euro (or \$1.6 million USD), with an average exchange rate of \$1.33 USD per €1 Euro. These forward contracts are used to mitigate the variability in cash flows from the payment of liabilities denominated in currencies other than the USD. Since the Company does not treat these derivatives as hedges, the change in fair value of both the forward contracts and the related liabilities are recorded in the income statement. For the six months ended June 30, 2008, the Company recorded a fair value gain in the amount of \$295,000.

At June 30, 2008, the Company had forward contracts for the sale of USD and the purchase of South African Rand (ZAR). The Company concluded that these foreign currency forward contracts qualify for cash flow hedge accounting which permits the recording of the fair value of the forward contract and corresponding adjustment to other comprehensive income (loss), net of tax, on the balance sheet. As of June 30, 2008, the Company had forward contracts with a notional value of R18.7 million ZAR (or \$2.4 million USD), with an average exchange rate of R7.78 ZAR per USD, resulting in the recording of a current asset of \$68,000 and a corresponding offset in accumulated other comprehensive loss of \$43,000, net of tax.

At December 31, 2006, the Company had forward contracts for the sale of U.S. Dollars (USD) and the purchase of Canadian Dollars (CAD). The Company concluded that these foreign currency forward contracts qualify for cash flow hedge accounting which permits the recording of the fair value of the forward contract and corresponding adjustment to other comprehensive income (loss), net of tax, on the balance sheet. As of December 31, 2006, the Company had forward contracts with a notional value of \$48 million CAD (or \$42.7 million U.S.) with an average exchange rate of \$0.89 USD per \$1 CAD, resulting in the recording of a current liability of \$1.3 million and a corresponding offset in accumulated other comprehensive loss of \$825,000, net of tax. During 2007, these foreign currency forward contracts were settled.

Foreign Currency Translation Assets and liabilities of foreign subsidiaries, except for the Company's Mexican operations whose functional currency is the U.S. Dollar, are translated at the rate of exchange in effect on the balance sheet date while income and expenses are translated at the average rates of exchange prevailing during the year. Foreign currency gains and losses resulting from transactions, and the translation of financial statements are recorded in the Company's consolidated financial statements based upon the provisions of SFAS No. 52, "Foreign Currency Translation." The effects of currency exchange rate changes on intercompany transactions and balances of a long-term investment nature are accumulated and carried as a component of shareholders' equity. The effects of currency exchange rate changes on transactions that are denominated in a currency other than an entity's functional currency are charged or credited to earnings. Foreign exchange transaction losses recognized as other expense were \$854,000 and \$1.3 million for the three months ended June 30, 2008 and 2007, respectively, and \$1.2 million and \$2.1 million for the six months ended June 30, 2008 and 2007, respectively.

Other Comprehensive Income Comprehensive income is defined as net income and all other non-owner changes in shareholders' equity. The Company's accumulated other comprehensive income consists of foreign

WESTINGHOUSE AIR BRAKE TECHNOLOGIES CORPORATION

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS—(Continued) FOR THE QUARTERLY PERIOD ENDED JUNE 30, 2008 (UNAUDITED)

currency translation adjustments, foreign currency hedges, foreign exchange contracts, and pension and post retirement related adjustments. Changes in the table below adjust components of accumulated other comprehensive income. Total comprehensive income was:

	Three Mor			ths Ended e 30,
In thousands	2008	2007	2008	2007
Net income	\$33,762	\$28,130	\$66,272	\$53,652
Foreign currency translation gain	4,010	9,937	12,391	12,963
Unrealized (loss) gain on foreign exchange contracts, net of tax	(107)	1,232	43	1,581
Change in pension and post retirement benefit plans, net of tax	227	_	482	_
Total comprehensive income	\$37,892	\$39,299	\$79,188	\$68,196

The components of accumulated other comprehensive loss were:

In thousands	2008	2007
Foreign currency translation gain	\$ 52,973	\$ 40,582
Unrealized gain on foreign exchange contracts, net of tax	43	_
Pension benefit plans and post retirement benefit plans, net of tax	(30,829)	(31,311)
Total accumulated comprehensive income	\$ 22,187	\$ 9,271

Reclassifications Certain prior year amounts have been reclassified where necessary to conform to the current year presentation.

Recent Accounting Pronouncements In September 2006, the FASB issued SFAS No. 157, "Fair Value Measurements," (SFAS 157). SFAS 157 defines fair value, establishes a framework for measuring fair value in generally accepted accounting principles, and expands disclosures about fair value measurements. The provisions of this standard apply to other accounting pronouncements that require or permit fair value measurements. The statement indicates, among other things, that a fair value measurement assumes that the transaction to sell an asset or transfer a liability occurs in the principal market for the asset or liability or, in the absence of a principal market, the most advantageous market for the asset or liability. SFAS 157 defines fair value based upon an exit price model. Relative to SFAS 157, the FASB issued FASB Staff Positions (FSP) 157-1, 157-2, and proposed 157-c. FSP 157-1 amends SFAS 157 to exclude SFAS 13 and its related interpretive accounting pronouncements that address leasing transactions, while FSP 157-2 delays the effective date of SFAS 157 for all nonfinancial assets and nonfinancial liabilities, except those that are recognized or disclosed at fair value in the financial statements on a recurring basis. FSP 157-c clarifies the principles in SFAS 157 on the fair value measurement of liabilities. Public comments on FSP 157-c were due in February 2008, and responses and recommendations were presented to the board on April 9, 2008. The Company adopted SFAS 157 as of January 1, 2008, with the exception of the application of the statement to non-recurring nonfinancial assets and nonfinancial liabilities. Refer to Note 13 to the Condensed Consolidated Financial Statements for additional discussion on fair value measurements.

In February 2007, the FASB issued SFAS No. 159, "The Fair Value Option for Financial Assets and Financial Liabilities—Including an Amendment of FASB Statement No. 115" (SFAS 159). SFAS 159 permits entities to choose to measure eligible items at fair value at specified election dates and report unrealized gains

WESTINGHOUSE AIR BRAKE TECHNOLOGIES CORPORATION

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS—(Continued) FOR THE QUARTERLY PERIOD ENDED JUNE 30, 2008 (UNAUDITED)

and losses on items for which the fair value option has been elected in earnings at each subsequent reporting date. SFAS 159 is effective for fiscal years beginning after November 15, 2007. The Company adopted SFAS 159 and has elected not to measure any additional financial instruments and other items at fair value.

In December 2007, the FASB issued SFAS No. 141 (Revised), "Business Combinations" (SFAS 141(R)), replacing SFAS No. 141, "Business Combinations" (SFAS 141), and SFAS No. 160, "Noncontrolling Interests in Consolidated Financial Statements—an Amendment of ARB No. 51" (SFAS 160). SFAS 141(R) retains the fundamental requirements of SFAS 141, broadens its scope by applying the acquisition method to all transactions and other events in which one entity obtains control over one or more other businesses, and requires, among other things, that assets acquired and liabilities assumed be measured at fair value as of the acquisition date, that liabilities related to contingent consideration be recognized at the acquisition date and remeasured at fair value in each subsequent reporting period, that acquisition-related costs be expensed as incurred, and that income be recognized if the fair value of the net assets acquired exceeds the fair value of the consideration transferred. SFAS 160 establishes accounting and reporting standards for noncontrolling interests (i.e., minority interests) in a subsidiary, including changes in a parent's ownership interest in a subsidiary and requires, among other things, that noncontrolling interests in subsidiaries be classified as a separate component of equity. Except for the presentation and disclosure requirements of SFAS 160, which are to be applied retrospectively for all periods presented, SFAS 141(R) and SFAS 160 are to be applied prospectively in financial statements issued for fiscal years beginning after December 15, 2008. The Company is currently evaluating the impact of adopting these statements.

In March 2008, the FASB issued SFAS No. 161, "Disclosures about Derivative Instruments and Hedging Activities—an Amendment FASB Statement No. 133" (SFAS 161). SFAS 161 requires enhanced disclosures about an entity's derivative and hedging activities and thereby improves the transparency of financial reporting. SFAS 161 applies to all derivative instruments within the scope of SFAS No. 133, "Accounting for Derivative Instruments and Hedging Activities" (SFAS 133) as well as related hedged items, bifurcated derivatives, and nonderivative instruments that are designated and qualify as hedging instruments. Entities with instruments subject to SFAS 161 must provide more robust qualitative disclosures and expanded quantitative disclosures. SFAS 161 is effective prospectively for financial statements issued for fiscal years and interim periods beginning after November 15, 2008, with early application permitted. The Company is currently evaluating the disclosure implications of adopting this statement.

In June 2008, the FASB issued FSP Emerging Issues Task Force ("EITF") 03-6-1, "Determining Whether Instruments Granted in Share-Based Payment Transactions Are Participating Securities" (FSP EITF 03-6-1). FSP EITF 03-6-1 states that unvested share-based payment awards that contain nonforfeitable rights to dividends or dividend equivalents (whether paid or unpaid) are participating securities and shall be included in the computation of earnings per share pursuant to the two-class method. FSP ETIF 03-6-1 is effective for fiscal years beginning after December 15, 2008, and interim periods within those years. Upon adoption, a company is required to retrospectively adjust its earnings per share data (including any amounts related to interim periods, summaries of earnings and selected financial data) to conform with the provisions of FSP EITF 03-6-1. The Company is currently evaluating the impact of adopting FSP EITF 03-6-1.

3. ACQUISITIONS AND DISCONTINUED OPERATIONS

On June 30, 2008, subsequent to the close of our four-four-five accounting quarter, the Company acquired POLI S.p.A., a European based manufacturer of rail braking equipment, for a net price of €53 million (about \$80 million USD) in cash. POLI's products include brake discs for high speed applications, as well as tread brake units and pneumatic brake valves that meet International Union of Railways (UIC) standards. POLI will operate as a business of Wabtec's Transit Group.

WESTINGHOUSE AIR BRAKE TECHNOLOGIES CORPORATION

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS—(Continued) FOR THE QUARTERLY PERIOD ENDED JUNE 30, 2008 (UNAUDITED)

On June 8, 2007, the Company acquired 100% of the stock of Ricon Corporation (Ricon), a manufacturer of a variety of electro-mechanical wheelchair lifts and ramps and windows, primarily for passenger bus and rail applications. The purchase price was \$73.7 million resulting in additional goodwill of \$50.4 million, of which none will be deductible for tax purposes. Included in the purchase price is \$6.5 million related to an escrow deposit, which may be released to the Company for working capital adjustments or indemnity claims in accordance with the purchase and escrow agreements.

For the Ricon acquisition, the following table summarizes the fair values of the assets acquired and liabilities assumed at the date of the acquisition:

	Ricon
	June 8,
In thousands	
Current assets	\$ 21,100
Property, plant & equipment	3,000
Goodwill and other intangible assets	72,000
Other assets	100
Total assets acquired	96,200
Total liabilities assumed	(22,500)
Net assets acquired	\$ 73,700

Of the allocation of \$21.6 million of acquired intangible assets, exclusive of goodwill, \$9.7 million was assigned to the trade name, \$9.3 million was assigned to customer relationships, \$1.8 million was assigned to patents, and \$760,000 was assigned to customer backlog. The trade name is considered to have an indefinite useful life while the customer relationships and patents average useful life is 10 years.

The following unaudited pro forma financial information presents income statement results as if the acquisition of Ricon had occurred on January 1, 2007:

	Thr	ee Months Ended	Six	Months Ended	
In thousands		June 30, 2007	<u>_</u>	June 30, 2007	
Net sales	\$	337,219	\$	668,025	
Gross profit		94,362		185,492	
Net income		28,034		52,944	
Diluted earnings per share					
As Reported	\$	0.57	\$	1.09	
Pro forma		0.57		1.08	

At March 31, 2006, the sale of a non-core product division was completed for approximately \$1.4 million in cash, including a working capital adjustment of approximately \$600,000 which was established with the buyer in the fourth quarter of 2006. The assets sold primarily included transit car interior products and services for customers located in Europe. This sale resulted in a loss of approximately \$1.7 million including the working capital adjustment. This adjustment is subject to litigation and a resolution is expected sometime in late 2008.

 $At \ August \ 6, 2007, the \ sale \ of \ a \ joint \ venture \ in \ China \ was \ completed \ for \ approximately \$398,000 \ in \ cash.$

WESTINGHOUSE AIR BRAKE TECHNOLOGIES CORPORATION

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS—(Continued) FOR THE QUARTERLY PERIOD ENDED JUNE 30, 2008 (UNAUDITED)

4. INVENTORIES

The components of inventory, net of reserves, were:

In thousands	June 30, 2008	December 31, 2007
Raw materials	\$ 77,367	\$ 68,542
Work-in-process	80,072	71,282
Finished goods	38,892	36,153
Total inventory	\$ 196,331	\$ 175,977

5. RESTRUCTURING AND IMPAIRMENT CHARGES

Since 2006, Wabtec has downsized its Canadian operations by moving certain products to lower-cost facilities and outsourcing. In the Freight segment, Wabtec recorded charges of \$3.1 million and \$3.0 million for the three months ended June 30, 2008 and 2007, respectively. Wabtec recorded charges of \$3.1 million and \$4.0 million for the six months ended June 30, 2008 and 2007, respectively. Total expenses for restructuring and other expenses recorded since 2006 have been \$14.9 million, comprised of the \$4.8 million for employee severance costs for approximately 370 salaried and hourly employees; \$4.8 million of pension and postretirement benefit curtailment for those employees; \$4.8 million related to asset impairments for structures, machinery, and equipment; and \$541,000 for goodwill impairment specific to one location. The goodwill impairment was recorded as amortization expense and most of the other charges were recorded to cost of sales. Severance costs are contractual liabilities and payment is dependent on the waiver by or expiration of certain seniority rights of those employees. As of June 30, 2008, \$2.3 million of this amount had been paid. In the Transit segment, Wabtec recorded charges of \$1.1 million for the six months ended June 30, 2007. Total expenses for restructuring and other expenses recorded since 2006 have been \$1.5 million, comprised of severance, pension, and asset impairment. All charges were recorded to cost of sales. As of June 30, 2008, all but \$68,000 of these costs has been paid.

6. INTANGIBLES

Goodwill was \$237.8 million and \$232.6 million at June 30, 2008 and December 31, 2007, respectively. The change in the carrying amount of goodwill by segment for the three months ended June 30, 2008 is as follows:

	Freight	Transit	
<u>In thousands</u>	Group	Group	Total
Balance at December 31, 2007	\$ 114,829	\$ 117,764	\$ 232,593
Adjustment to preliminary purchase allocation	_	654	654
Foreign currency impact	1,005	3,590	4,595
Balance at June 30 2008	\$ 115,834	\$ 122,008	\$ 237,842

WESTINGHOUSE AIR BRAKE TECHNOLOGIES CORPORATION

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS—(Continued) FOR THE QUARTERLY PERIOD ENDED JUNE 30, 2008 (UNAUDITED)

As of June 30, 2008 and December 31, 2007, the Company's trademarks had a net carrying amount of \$34.8 million and \$34.5 million, respectively, and the Company believes these intangibles have an indefinite life. Intangible assets of the Company, other than goodwill and trademarks, consist of the following:

In thousands_	June 30, 	December 31, 2007
Patents and other, net of accumulated amortization of \$27,153 and \$25,620	\$ 7,681	\$ 8,702
Customer relationships, net of accumulated amortization of \$2,013 and \$1,320	15,191	15,450
Total	\$22,872	\$ 24,152

The weighted average remaining useful life of patents is 7 years and customer relationships is 13 years. Amortization expense for intangible assets was \$903,000 and \$1.8 million for the three and six months ended June 30, 2008, respectively, and \$993,000 and \$1.5 million for the three and six months ended June 30, 2007, respectively.

7. LONG-TERM DEBT

Long-term debt consisted of the following:

	June 30,	December 31,
In thousands	2008	2007
6.875% senior notes, due 2013	\$ 150,000	\$ 150,000
Capital Leases	203	250
Total	150,203	150,250
Less—current portion	68	73
Long-term portion	\$ 150,135	\$ 150,177

Refinancing Credit Agreement

In January 2004, the Company refinanced its existing unsecured revolving credit agreement with a consortium of commercial banks. This "Refinancing Credit Agreement" provided a \$175 million five-year revolving credit facility expiring in January 2009. In November 2005, the Company entered into an amendment to the Refinancing Credit Agreement which, among other things, extended the expiration of the agreement until December 2010. The Company entered into an amendment to its Refinancing Credit Agreement in February 2007 which permits the Company to complete any acquisitions without prior approval of the bank consortium as long as certain financial parameters and ratios are met. At June 30, 2008, the Company had available bank borrowing capacity, net of \$58.7 million of letters of credit, of approximately \$116.3 million, subject to certain financial covenant restrictions.

Refinancing Credit Agreement borrowings bear variable interest rates indexed to the indices described below. The Company did not borrow under the Refinancing Credit Agreement during the six months ended June 30, 2008 or during the year ended December 31, 2007.

Under the Refinancing Credit Agreement, the Company may elect a base interest rate or an interest rate based on the London Interbank Offered Rates of Interest ("LIBOR"). The base interest rate is the greater of

WESTINGHOUSE AIR BRAKE TECHNOLOGIES CORPORATION

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS—(Continued) FOR THE QUARTERLY PERIOD ENDED JUNE 30, 2008 (UNAUDITED)

LaSalle Bank National Association's prime rate or the federal funds effective rate plus 0.5% per annum. The LIBOR rate is based on LIBOR plus a margin that ranges from 62.5 to 175 basis points depending on the Company's consolidated total indebtedness to cash flow ratios. The current margin is 62.5 basis points.

The Refinancing Credit Agreement limits the Company's ability to declare or pay cash dividends and prohibits the Company from declaring or making other distributions, subject to certain exceptions. The Refinancing Credit Agreement contains various other covenants and restrictions including the following limitations: incurrence of additional indebtedness; mergers, consolidations and sales of assets and acquisitions; additional liens; sale and leasebacks; permissible investments, loans and advances; certain debt payments; capital expenditures; and imposes a minimum interest expense coverage ratio and a maximum debt to cash flow ratio.

The Refinancing Credit Agreement contains customary events of default, including payment defaults, failure of representations or warranties to be true in any material respect, covenant defaults, defaults with respect to other indebtedness of the Company, bankruptcy, certain judgments against the Company, ERISA defaults and "change of control" of the Company. The Refinancing Credit Agreement includes the following covenants: a minimum interest coverage ratio of 3, maximum debt to cash flow ratio of 3.25 and a minimum net worth of \$180 million plus 50% of consolidated net income since September 30, 2003. The Company is in compliance with these measurements and covenants and expects that these measurements will not be any type of limiting factor in executing our operating activities.

6.875% Senior Notes Due August 2013

In August 2003, the Company issued \$150 million of Senior Notes due in 2013 ("the Notes"). The Notes were issued at par. Interest on the Notes accrues at a rate of 6.875% per annum and is payable semi-annually on January 31 and July 31 of each year. The proceeds were used to repay debt outstanding under the Company's existing credit agreement, and for general corporate purposes. The principal balance is due in full at maturity.

The Notes are senior unsecured obligations of the Company and rank pari passu with all existing and future senior debt and senior to all our existing and future subordinated indebtedness of the Company. The indenture under which the Notes were issued contains covenants and restrictions which limit among other things, the following: the incurrence of indebtedness, payment of dividends and certain distributions, sale of assets, change in control, mergers and consolidations and the incurrence of liens.

During 2007, 509,800 shares were repurchased at an average price of \$35.06 per share. During the first quarter of 2008, the Company repurchased 712,900 shares of Wabtec stock at an average price of \$34.29 per share. During the second quarter of 2008, the Company repurchased 5,200 shares of Wabtec stock at an average price of \$35.97 per share. All purchases were on the open market.

WESTINGHOUSE AIR BRAKE TECHNOLOGIES CORPORATION

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS—(Continued) FOR THE QUARTERLY PERIOD ENDED JUNE 30, 2008 (UNAUDITED)

8. EMPLOYEE BENEFIT PLANS

Effective January 1, 2008, the Company early-adopted the measurement date (the date at which plan assets and the benefit obligation are measured) provisions of Statement of Financial Accounting Standards (SFAS) No. 158, "Employers' Accounting for Defined Benefit Pension and Other Postretirement Plans—an amendment of FASB Statements No. 87, 88, 106 and 132(R)" (SFAS 158). Under SFAS 158, the measurement date is required to be the Company's fiscal year-end. The Company's U.K. defined benefit pension plan previously used an October 31 measurement date. All plans are now measured as of December 31, consistent with the Company's fiscal year-end. The non-cash effect of the adoption of the measurement date provisions of SFAS 158 at January 1, 2008 decreased retained earnings by \$72,000, net of tax. There was no effect on the results of operations.

Defined Benefit Pension Plans

The Company sponsors defined benefit pension plans that cover certain U.S., Canadian, German and United Kingdom employees and which provide benefits of stated amounts for each year of service of the employee.

	U.: Three mon June	ths ended 30,	International Three months ended June 30, 2008 2007		
In thousands, except percentages	2008	2008 2007			
Net periodic benefit cost Service cost	\$ 77	\$ 88	\$ 887	\$ 962	
Interest cost	690	\$ 60 671	1,918	1,688	
Expected return on plan assets	(832)	(775)	(2,208)	(1,873)	
Net amortization/deferrals	338	404	366	408	
Curtailment loss recognized		404	2,122	400	
Settlement loss recognized			388		
Net periodic benefit cost	\$ 273	\$ 388	\$ 3,473	\$ 1,185	
•	\$ 2/3	<u>ъ 300</u>	\$ 3,473	\$ 1,105	
Assumptions	6.35%	5.80%	5.62%	5.11%	
Discount rate	8.00%	8.00%	7.17%	6.70%	
Expected long-term rate of return Rate of compensation increase	3.00%	3.00%	3.84%	3.62%	
	TT (T	.t1	
	U.: Six month June	ıs ended 30,	Interna Six month June	ns ended 30,	
In thousands, except percentages Not possible boundit cost	Six month	is ended	Six month	ns ended	
Net periodic benefit cost	Six month June 2008	s ended 30, 	Six montl June 2008	ns ended 30, 2007	
Net periodic benefit cost Service cost	Six month June 2008 \$ 153	s ended 30, 2007	Six month June 2008 \$ 1,775	2007 \$ 1,884	
Net periodic benefit cost Service cost Interest cost	Six month June 2008 \$ 153 1,381	\$ 176 1,343	Six month June 2008 \$ 1,775 3,839	\$ 1,884 33,314	
Net periodic benefit cost Service cost	Six month June 2008 \$ 153	\$ 176 1,343 (1,550)	Six month June 2008 \$ 1,775 3,839 (4,428)	\$ 1,884 33,14 (3,675)	
Net periodic benefit cost Service cost Interest cost Expected return on plan assets Net amortization/deferrals	Six month June 2008 \$ 153 1,381 (1,665)	\$ 176 1,343	Six month June 2008 \$ 1,775 3,839 (4,428) 735	\$ 1,884 33,314	
Net periodic benefit cost Service cost Interest cost Expected return on plan assets Net amortization/deferrals Curtailment loss recognized	Six month June 2008 \$ 153 1,381 (1,665)	\$ 176 1,343 (1,550)	Six month June 2008 \$ 1,775 3,839 (4,428)	\$ 1,884 33,14 (3,675)	
Net periodic benefit cost Service cost Interest cost Expected return on plan assets Net amortization/deferrals Curtailment loss recognized Settlement loss recognized	Six month June 2008 \$ 153 1,381 (1,665) 677 — —	\$ 176 1,343 (1,550) 807	\$ 1,775 3,839 (4,428) 735 2,122 520	\$ 1,884 3,314 (3,675) 799	
Net periodic benefit cost Service cost Interest cost Expected return on plan assets Net amortization/deferrals Curtailment loss recognized Settlement loss recognized Net periodic benefit cost	Six month June 2008 \$ 153 1,381 (1,665) 677 — —	\$ 176 1,343 (1,550) 807 —	\$ 1,775 3,839 (4,428) 735 2,122	\$ 1,884 33,14 (3,675)	
Net periodic benefit cost Service cost Interest cost Expected return on plan assets Net amortization/deferrals Curtailment loss recognized Settlement loss recognized	Six month June 2008 \$ 153 1,381 (1,665) 677 — —	\$ 176 1,343 (1,550) 807	\$ 1,775 3,839 (4,428) 735 2,122 520	\$ 1,884 3,314 (3,675) 799	
Net periodic benefit cost Service cost Interest cost Expected return on plan assets Net amortization/deferrals Curtailment loss recognized Settlement loss recognized Net periodic benefit cost Assumptions	Six month June 2008 \$ 153 1,381 (1,665) 677 — — 546	\$ 176 1,343 (1,550) 807 — — \$ 776	\$ 1,775 3,839 (4,428) 735 2,122 520 \$ 4,563	\$ 1,884 30, 2007 \$ 1,884 3,314 (3,675) 799 — \$ 2,322	

WESTINGHOUSE AIR BRAKE TECHNOLOGIES CORPORATION

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS—(Continued) FOR THE QUARTERLY PERIOD ENDED JUNE 30, 2008 (UNAUDITED)

The Company's funding methods are based on governmental requirements and differ from those methods used to recognize pension expense, which is primarily based on the projected unit credit method applied in the accompanying financial statements. The Company does not expect to contribute to the U.S. plan and expects to contribute \$6.7 million to the international plans during 2008.

Post Retirement Benefit Plans

In addition to providing pension benefits, the Company has provided certain unfunded postretirement health care and life insurance benefits for a portion of North American employees. The Company is not obligated to pay health care and life insurance benefits to individuals who had retired prior to 1990.

Tutamatianal

Three mon June	ths ended 30,	International Three months ended June 30,		
2008	2008 2007		2007	
	*	· -·	\$ 60	
			93	
(226)	(83)		61	
-	_	(954)	_	
		32		
\$ 319	\$ 503	\$ (958)	\$ 214	
	·			
6.35%	5.80%	5.50%	5.25%	
U. Six mont June	hs ended	Interna Six montl June	hs ended	
Six mont	hs ended	Six montl	hs ended	
Six mont June 2008	ns ended : 30, 	Six montl June 2008	hs ended 2 30, 2007	
Six monting June 2008 \$ 100	s ended 30, 2007	Six month June 2008 \$ 27	\$ 117	
Six mont June 2008	ns ended : 30, 	Six montl June 2008	hs ended 2 30, 2007	
Six monting June 2008 \$ 100	s ended 30, 2007	Six month June 2008 \$ 27	\$ 117	
Six month June 2008 \$ 100 990	\$ 117 1,055	Six month June 2008 \$ 27 105	\$ 117 181	
Six month June 2008 \$ 100 990	\$ 117 1,055	Six month June 2008 \$ 27 105 (205)	\$ 117 181	
Six month June 2008 \$ 100 990	\$ 117 1,055	Six month June 2008 \$ 27	\$ 117 181	
Six monting June 2008 \$ 100 990 (452) — — — —	\$ 117 1,055 (166)	\$ 27 105 (205) (954) 32	\$ 117 181 118 ————————————————————————————	
	\$ 50 495 (226) — — \$ 319	\$ 50 \$ 59 495 527 (226) (83) ————————————————————————————————————	Three months ended June 30, Three mon June 3008 2008 2007 2008 \$ 50 \$ 59 \$ 14 495 527 52 (226) (83) (102) — — (954) — — 32 \$ 319 \$ 503 \$ (958)	

9. STOCK-BASED COMPENSATION

As of June 30, 2008, the Company maintains employee stock-based compensation plans for stock options, restricted stock, and incentive stock awards as governed by the 2000 Stock Incentive Plan, as amended (the 2000 Plan). The Company also maintains a Non-Employee Directors' Fee and Stock Option Plan (Directors Plan).

Stock-based compensation expense was \$5.0 million and \$4.6 million for the six months ended June 30, 2008 and 2007, respectively. Included in the stock-based compensation expense for 2008 above is \$448,000 of

WESTINGHOUSE AIR BRAKE TECHNOLOGIES CORPORATION

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS—(Continued) FOR THE QUARTERLY PERIOD ENDED JUNE 30, 2008 (UNAUDITED)

expense related to stock options, \$1.5 million related to restricted stock, \$2.5 million related to incentive stock awards, and \$603,000 as compensation for Directors' fees. At June 30, 2008, unamortized compensation expense related to those stock options, restricted shares and incentive stock awards expected to vest totaled \$16.5 million and will be recognized over a weighted average period of 2.0 years.

Stock Options Under the 2000 Plan, stock options are granted to eligible employees at the fair market value, which is the average of the high and low Wabtec stock price on the date of grant. Generally, the options become exercisable over a three or four year vesting period and expire ten years from the date of grant. Options issued under the Directors Plan become exercisable over a three-year vesting period and expire ten years from the date of grant.

The following table summarizes the Company's stock option activity and related information for both the 2000 Plan and Directors Plan for the six months ended June 30, 2008:

	Options	Weighted Average Exercise Price	Weighted Average Remaining Contractual Life	Aggregate intrinsic value (in thousands)
Outstanding at December 31, 2007	1,009,494	\$ 14.61	5.2	\$ 20,022
Granted	253,500	34.78		3,310
Exercised	(132,384)	12.53		(4,675)
Canceled	(16,250)	19.49		(461)
Outstanding at June 30, 2008	1,114,360	\$ 19.37	5.9	\$ 31,724
Exercisable at June 30, 2008	821,852	\$ 14.08	4.7	\$ 27,747

The fair value of each option grant is estimated on the date of grant using the Black-Scholes option-pricing model with the following weighted-average assumptions:

	SIX IIIOIIUIS	ciiucu
	June 3	0,
	2008	2007
Dividend yield	.3%	.3%
Risk-free interest rate	3.5%	4.7%
Stock price volatility	33.9	40.7
Expected life (years)	5.0	5.0

Six months ended

The dividend yield is based on the Company's dividend rate and the current market price of the underlying common stock at the date of grant. Expected life in years is determined from historical stock option exercise data. Expected volatility is based on the historical volatility of the Company's stock. The risk-free interest rate is based on the U.S. Treasury bond rates for the expected life of the option.

Restricted Stock and Incentive Stock Awards Under the 2000 Plan, the Company adopted a restricted stock plan in 2006. Eligible employees are granted restricted stock that generally vests over three or four years from the date of grant.

In addition, the Company has issued incentive stock awards to eligible employees that vest upon attainment of certain cumulative three year performance goals. The incentive stock awards included in the table below represent the maximum number of shares that may ultimately vest. As of June 30, 2008, based on the Company's performance, we estimate that the majority of these stock awards will vest and have recorded compensation expense accordingly. If our estimate of the number of these stock awards expected to vest changes in a future accounting period, compensation expense could be reduced and will be recognized over the remaining vesting period.

WESTINGHOUSE AIR BRAKE TECHNOLOGIES CORPORATION

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS—(Continued) FOR THE QUARTERLY PERIOD ENDED JUNE 30, 2008 (UNAUDITED)

Compensation expense for the restricted stock and incentive stock awards is based on the closing price of the Company's common stock on the date of grant and recognized over the applicable vesting period.

The following table summarizes the restricted stock and incentive stock awards activity and related information for the three months ended June 30, 2008:

	Restricted Stock	Incentive Stock Awards	Aver Da	eighted age Grant ate Fair Value
Outstanding at December 31, 2007	301,500	694,049	\$	29.65
Granted	73,000	293,000		34.92
Vested	(82,958)	(243,913)		20.51
Canceled	(16,500)	(51,470)		27.16
Outstanding at June 30, 2008	275,042	691,666	\$	34.91

10. INCOME TAXES

The overall effective income tax rate was 37.4% for both the three and six months ended June 30, 2008 and 35.5% and 36.3% for the three and six months ended June 30, 2007, respectively.

The Company adopted the provisions of Financial Accounting Standards Board Interpretation No. 48 Accounting for Uncertainty in Income Taxes ("FIN 48")—an interpretation of FASB Statement No. 109 on January 1, 2007. FIN 48 clarifies the accounting for uncertainty in income taxes.

As of January 1, 2008, the liability for income taxes associated with uncertain tax positions was \$17.2 million, of which \$10.6 million, if recognized, would favorably affect the Company's effective tax rate. The total liability related to uncertain tax positions as of June 30, 2008 is \$17.7 million, of which \$10.9 million, if recognized, would favorably affect the Company's effective income tax rate.

The Company includes interest and penalties related to uncertain tax positions in income tax expense. As of January 1, 2008, the Company had accrued interest and penalties related to uncertain tax positions of approximately \$3.3 and \$1.8 million, respectively. The total interest and penalties accrued as of June 30, 2008 are approximately \$4.1 and \$2.2 million, respectively.

In April 2008, the Internal Revenue Service completed its audit of the tax year ended December 31, 2004 without change to the tax as reported by the Company. With limited exception, the Company is no longer subject to examination by various U.S. and foreign taxing authorities for years before 2003.

At this time, the Company believes that it is reasonably possible that unrecognized tax benefits may significantly change within the next twelve months. The approximate range is \$3.0 million to \$4.0 million based on the expiration of statutory review periods and current examinations.

WESTINGHOUSE AIR BRAKE TECHNOLOGIES CORPORATION

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS—(Continued) FOR THE QUARTERLY PERIOD ENDED JUNE 30, 2008 (UNAUDITED)

11. EARNINGS PER SHARE

The computation of earnings per share is as follows:

		nths Ended e 30,
In thousands, except per share	2008	2007
Basic earnings per share		
Income from continuing operations applicable to common shareholders	\$33,762	\$28,125
Divided by		
Weighted average shares outstanding	48,095	48,666
Basic earnings from continuing operations per share	\$ 0.70	\$ 0.58
Diluted earnings per share		
Income from continuing operations applicable to common shareholders	\$33,762	\$28,125
Divided by sum of the		
Weighted average shares outstanding	48,095	48,666
Conversion of dilutive stock options and non-vested stock	560	628
Diluted shares outstanding	48,655	49,294
Diluted earnings from continuing operations per share	\$ 0.69	\$ 0.57
	Jun	ths Ended te 30,
In thousands, except per share Resign correlings por share		
Basic earnings per share	2008	2007
Basic earnings per share Income from continuing operations applicable to common shareholders	Jun	e 30,
Basic earnings per share Income from continuing operations applicable to common shareholders Divided by	2008	2007
Basic earnings per share Income from continuing operations applicable to common shareholders	3008 \$66,275	2007 \$53,679
Basic earnings per share Income from continuing operations applicable to common shareholders Divided by Weighted average shares outstanding Basic earnings from continuing operations per share	2008 \$66,275 48,306	2007 \$53,679 48,413
Basic earnings per share Income from continuing operations applicable to common shareholders Divided by Weighted average shares outstanding	2008 \$66,275 48,306	2007 \$53,679 48,413
Basic earnings per share Income from continuing operations applicable to common shareholders Divided by Weighted average shares outstanding Basic earnings from continuing operations per share Diluted earnings per share	\$66,275 48,306 \$ 1.37	\$53,679 48,413 \$ 1.11
Basic earnings per share Income from continuing operations applicable to common shareholders Divided by Weighted average shares outstanding Basic earnings from continuing operations per share Diluted earnings per share Income from continuing operations applicable to common shareholders	\$66,275 48,306 \$ 1.37	\$53,679 48,413 \$ 1.11
Basic earnings per share Income from continuing operations applicable to common shareholders Divided by Weighted average shares outstanding Basic earnings from continuing operations per share Diluted earnings per share Income from continuing operations applicable to common shareholders Divided by sum of the	\$66,275 48,306 \$ 1.37 \$66,275	\$53,679 48,413 \$ 1.11 \$53,679
Basic earnings per share Income from continuing operations applicable to common shareholders Divided by Weighted average shares outstanding Basic earnings from continuing operations per share Diluted earnings per share Income from continuing operations applicable to common shareholders Divided by sum of the Weighted average shares outstanding	\$66,275 48,306 \$ 1.37 \$66,275 48,306	\$53,679 48,413 \$ 1.11 \$53,679

12. WARRANTIES

The following table reconciles the changes in the Company's product warranty reserve:

	Six Mont	
In thousands	2008	2007
Balance at December 31, 2007 and 2006, respectively	\$22,314	\$17,399
Warranty provision	9,752	5,329
Acquisition	_	1,399
Warranty claim payments	(5,305)	(5,489)
Balance at June 30, 2008 and 2007, respectively	\$26,761	\$18,638

WESTINGHOUSE AIR BRAKE TECHNOLOGIES CORPORATION

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS—(Continued) FOR THE QUARTERLY PERIOD ENDED JUNE 30, 2008 (UNAUDITED)

13. FAIR VALUE MEASUREMENT

In September 2006, the FASB issued SFAS No. 157, "Fair Value Measurements," (SFAS 157), which is effective for fiscal years beginning after November 15, 2007 and for interim periods within those years. This statement defines fair value, establishes a framework for measuring fair value and expands the related disclosure requirements. This statement is also applicable under other accounting pronouncements that require or permit fair value measurements. The statement indicates, among other things, that a fair value measurement assumes that the transaction to sell an asset or transfer a liability occurs in the principal market for the asset or liability or, in the absence of a principal market, the most advantageous market for the asset or liability. SFAS 157 defines fair value based upon an exit price model.

Relative to SFAS 157, the FASB issued FASB Staff Positions (FSP) 157-1 and 157-2. FSP 157-1 amends SFAS 157 to exclude SFAS No. 13, "Accounting for Leases," (SFAS 13) and its related interpretive accounting pronouncements that address leasing transactions, while FSP 157-2 delays the effective date of the application of SFAS 157 to fiscal years beginning after November 15, 2008 for all nonfinancial assets and nonfinancial liabilities that are recognized or disclosed at fair value in the financial statements on a nonrecurring basis.

The Company adopted SFAS 157 as of January 1, 2008, with the exception of the application of the statement to non-recurring nonfinancial assets and nonfinancial liabilities. Non-recurring nonfinancial assets and nonfinancial liabilities for which the Company has not applied the provisions of SFAS 157 include those measured at fair value in goodwill impairment testing, indefinite lived intangible assets measured at fair value for impairment testing, asset retirement obligations initially measured at fair value, and those initially measured at fair value in a business combination. The implementation of SFAS 157 for financial assets and financial liabilities did not have a material impact on the Company's consolidated financial position and results of operations. The Company is currently evaluating the impact of adopting SFAS 157 for nonfinancial assets and nonfinancial liabilities.

Valuation Hierarchy. SFAS 157 establishes a valuation hierarchy for disclosure of the inputs to valuation used to measure fair value. This hierarchy prioritizes the inputs into three broad levels as follows. Level 1 inputs are quoted prices (unadjusted) in active markets for identical assets or liabilities. Level 2 inputs are quoted prices for similar assets and liabilities in active markets or inputs that are observable for the asset or liability, either directly or indirectly through market corroboration, for substantially the full term of the financial instrument. Level 3 inputs are unobservable inputs based on the Company's assumptions used to measure assets and liabilities at fair value. A financial asset or liability's classification within the hierarchy is determined based on the lowest level input that is significant to the fair value measurement.

The following table provides the assets and liabilities carried at fair value measured on a recurring basis as of June 30, 2008:

		Fair Value Measurements at June 30, 2008 U			
	Total Carrying	Quoted Prices in		Significant	
	Value at	Active Markets for	Significant Other	Unobservable	
	June 30,	Identical Assets	Observable Inputs	Inputs	
In thousands	2008	(Level 1)	(Level 2)	(Level 3)	
Foreign currency forward contracts	\$ 349	<u>\$</u>	\$ 349	\$ —	
Total	\$ 349	\$	\$ 349	\$ —	

As a result of our global operating activities, the Company is exposed to market risks from changes in foreign currency exchange rates, which may adversely affect our operating results and financial position. When

WESTINGHOUSE AIR BRAKE TECHNOLOGIES CORPORATION

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS—(Continued) FOR THE QUARTERLY PERIOD ENDED JUNE 30, 2008 (UNAUDITED)

deemed appropriate, the Company minimizes these risks through entering into foreign currency forward contracts. The foreign currency forward contracts are valued using broker quotations, or market transactions in either the listed or over-the counter markets. As such, these derivative instruments are classified within level 2.

14. COMMITMENTS AND CONTINGENCIES

Claims have been filed against the Company and certain of its affiliates in various jurisdictions across the United States by persons alleging bodily injury as a result of exposure to asbestos-containing products. Since 2000, the number of such claims has increased and the resolution of these claims may take a significant period of time. Most of these claims have been made against our wholly owned subsidiary, Railroad Friction Products Corporation (RFPC), and are based on a product sold by RFPC prior to the time that the Company acquired any interest in RFPC. On April 17, 2005, a claim against the Company by a former stockholder of RFPC contending that the Company assumed that entity's liability for asbestos claims arising from exposure to RFPC's product was resolved in the Company's favor.

Most of these claims, including all of the RFPC claims, are submitted to insurance carriers for defense and indemnity or to non-affiliated companies that retain the liabilities for the asbestos-containing products at issue. We cannot, however, assure that all these claims will be fully covered by insurance or that the indemnitors will remain financially viable. Our ultimate legal and financial liability with respect to these claims, as is the case with other pending litigation, cannot be estimated.

It is Management's belief that the potential range of loss for asbestos-related bodily injury cases is not reasonably determinable at present for a variety of factors, including: (1) the limited asbestos case settlement history of the Company's wholly owned subsidiary, RFPC; (2) the unpredictable nature of personal injury litigation in general; and (3) the uncertainty of asbestos litigation in particular. Despite this uncertainty, and although the results of the Company's operations and cash flows for any given period could be adversely affected by asbestos-related lawsuits, Management believes that the final resolution of the Company's asbestos-related cases will not be material to the Company's overall financial position, results of operations and cash flows. In general, this belief is based upon: (1) Wabtec's and RFPC's limited history of settlements and dismissals of asbestos-related cases to date; (2) the inability of many plaintiffs to establish any exposure or causal relationship to RFPC's product; and (3) the inability of many plaintiffs to demonstrate any identifiable injury or compensable loss.

More specifically, as to RFPC, Management's belief that any losses due to asbestos-related cases would not be material is also based on the fact that RFPC owns insurance which provides coverage for asbestos-related bodily injury claims. To date, RFPC's insurers have provided RFPC with defense and indemnity in these actions. As to Wabtec and its divisions, Management's belief that asbestos-related cases will not have a material impact is also based on its position that it has no legal liability for asbestos-related bodily injury claims, and that the former owners of Wabtec's assets retained asbestos liabilities for the products at issue. To date, Wabtec has been able to successfully defend itself on this basis, including two arbitration decisions and a judicial opinion, all of which confirmed Wabtec's position that it did not assume any asbestos liabilities from the former owners of certain Wabtec assets. Although Wabtec has incurred defense and administrative costs in connection with asbestos bodily injury actions, these costs have not been material, and the Company has no information that would suggest these costs would become material in the foreseeable future.

In March 2006, Management began an internal investigation related to business transactions conducted by a subsidiary, Pioneer Friction Limited ("Pioneer"), in West Bengal, India. Through an internal compliance review,

WESTINGHOUSE AIR BRAKE TECHNOLOGIES CORPORATION

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS—(Continued) FOR THE QUARTERLY PERIOD ENDED JUNE 30, 2008 (UNAUDITED)

Management discovered that disbursements were made which may be in violation of applicable laws and regulations. Pioneer is a fourth-tier subsidiary of Wabtec; two of the intermediate subsidiaries are Australian companies which are, in turn, owned by a U.S. holding company. The Company has entered into a non-prosecution agreement with the United States Department of Justice and has paid a penalty of \$300,000. The Company has reached a civil settlement with the Securities and Exchange Commission and will pay a penalty of \$377,000.

On October 18, 2007, Faiveley Transport Malmo AB filed a request for arbitration with the International Chamber of Commerce alleging breach of contract and trade secret violations relating to the Company's manufacture and sale of certain components. The components at issue are limited in number and used in the transit industry. On that same day, Faiveley also filed a related proceeding against the Company in the United States District Court for the Southern District of New York, requesting a preliminary injunction in aid of the arbitration. In both forums, Faiveley seeks to prevent the Company from manufacturing and selling the subject components until the arbitration panel decides Faiveley's claim. In the arbitration, Faiveley also seeks an unspecified amount of damages. The Company's motion and subsequent appeal to dismiss the federal court action on jurisdictional grounds were denied. During the week of July 28, 2008, the federal court conducted a hearing on Faiveley's injunction request. A decision is expected in mid-August 2008. The arbitration panel expects to conduct its hearing on the underlying breach of contract issue in February 2009. The Company denies Faiveley's allegations and does not believe that it has any material legal liability in this matter; it will vigorously contest both proceedings.

The Company is subject to a number of other commitments and contingencies as described in its Annual Report on Form 10-K for the Year Ended December 31, 2007, filed on February 29, 2008. During the first six months of 2008, there were no material changes to the information described in Note 18 therein.

15. SEGMENT INFORMATION

Wabtec has two reportable segments—the Freight Group and the Transit Group. The key factors used to identify these reportable segments are the organization and alignment of the Company's internal operations, the nature of the products and services, and customer type. The business segments are:

Freight Group manufactures products and provides services geared primarily to the production and operation of freight cars and locomotives, including braking control equipment, on-board electronic components and train coupler equipment.

Transit Group consists of products for passenger transit vehicles and locomotives (typically subways, commuter rail and buses) that include braking, coupling, monitoring systems, climate control and door equipment engineered to meet individual customer specifications, as well as commuter rail locomotives.

The Company evaluates its business segments' operating results based on income from operations. Corporate activities include general corporate expenses, elimination of intersegment transactions, interest income and expense and other unallocated charges. Since certain administrative and other operating expenses and other items have not been allocated to business segments, the results in the following tables are not necessarily a measure computed in accordance with generally accepted accounting principles and may not be comparable to other companies.

Income (loss) from operations Interest expense and other, net

Income (loss) from continuing operations before income taxes

WESTINGHOUSE AIR BRAKE TECHNOLOGIES CORPORATION

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS—(Continued) FOR THE QUARTERLY PERIOD ENDED JUNE 30, 2008 (UNAUDITED)

Segment financial information for the three months ended June 30, 2008 is as follows:

In thousands	Freight Group	Transit Group	Corporate Activities and Elimination	Total
Sales to external customers	\$199,631	\$190,563	\$ —	\$390,194
Intersegment sales/(elimination)	5,354	557	(5,911)	
Total sales	\$204,985	\$191,120	\$ (5,911)	\$390,194
Income (loss) from operations	\$ 36,154	\$ 25,102	\$ (5,334)	\$ 55,922
Interest expense and other, net	_	_	(1,989)	(1,989)
Income (loss) from continuing operations before income taxes	\$ 36,154	\$ 25,102	\$ (7,323)	\$ 53,933
Segment financial information for the three months ended June 30, 2007 is as follows:				
In thousands	Freight Group	Transit Group	Corporate Activities and Elimination	Total
Sales to external customers	\$180,986	\$144,736	\$ —	\$325,722
Intersegment sales/(elimination)	3,703	231	(3,934)	
Total sales	\$184,689	\$144,967	\$ (3,934)	\$325,722
Income (loss) from operations	\$ 31,054	\$ 18,981	\$ (4,118)	\$ 45,917
Interest expense and other, net	_	_	(2,323)	(2,323)
Income (loss) from continuing operations before income taxes	\$ 31,054	\$ 18,981	\$ (6,441)	\$ 43,594
Segment financial information for the six months ended June 30, 2008 is as follows:				
In thousands_	Freight Group	Transit Group	Corporate Activities and Elimination	Total
Sales to external customers	\$391,397	\$382,124	\$ —	\$773,521
Intersegment sales/(elimination)	9,897	1,001	(10,898)	
Total sales	\$401,294	\$383,125	\$ (10,898)	\$773,521

\$ 72,163

\$ 72,163

\$ 47,534

\$ 47,534

\$109,808

\$105,955

(3,853)

(9,889)

(3,853)

(13,742)

WESTINGHOUSE AIR BRAKE TECHNOLOGIES CORPORATION

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS—(Continued) FOR THE QUARTERLY PERIOD ENDED JUNE 30, 2008 (UNAUDITED)

Segment financial information for the six months ended June 30, 2007 is as follows:

	Freight	Transit	Corporate Activities and	
In thousands	Group	Group	Elimination	Total
Sales to external customers	\$365,653	\$274,333	\$ —	\$639,986
Intersegment sales/(elimination)	7,407	453	(7,860)	
Total sales	\$373,060	\$274,786	\$ (7,860)	\$639,986
Income (loss) from operations	\$ 66,539	\$ 29,596	\$ (7,954)	\$ 88,181
Interest expense and other, net	_	_	(3,915)	(3,915)
Income (loss) from continuing operations before income taxes	\$ 66,539	\$ 29,596	\$ (11,869)	\$ 84,266

Sales by product is as follows:

	Three Months Ended June 30,			
In thousands_		2008		2007
Brake products	\$	131,852	\$	112,876
Freight electronics & specialty products		96,339		87,568
Remanufacturing, overhaul & build		84,519		75,903
Transit products		55,304		34,104
Other		22,180		15,271
Total sales	\$	390,194	\$	325,722

	Six Months Ended June 30,			
In thousands_		2008		2007
Brake products	\$	262,262	\$	229,027
Freight electronics & specialty products		190,353		175,234
Remanufacturing, overhaul & build		179,211		147,263
Transit products		103,379		59,980
Other		38,316		28,482
Total sales	\$	773,521	\$	639,986

WESTINGHOUSE AIR BRAKE TECHNOLOGIES CORPORATION

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS—(Continued) FOR THE QUARTERLY PERIOD ENDED JUNE 30, 2008 (UNAUDITED)

16. GUARANTOR SUBSIDIARIES FINANCIAL INFORMATION

Effective August 2003, the Company issued \$150 million of Senior Notes due in 2013 ("Notes"). The obligations under the Notes are fully and unconditionally guaranteed by all U.S. subsidiaries as guarantors. In accordance with positions established by the Securities and Exchange Commission, the following shows separate financial information with respect to the parent, the guarantor subsidiaries and the non-guarantor subsidiaries. The principal elimination entries eliminate investment in subsidiaries and certain intercompany balances and transactions.

Balance Sheet as of June 30, 2008:

<u>In thousands</u>	Parent	Guarantors	Non-Guarantors	Elimination	Consolidated
Cash and cash equivalents	\$ 51,255	\$ 3,152	\$ 187,273	\$ —	\$ 241,680
Accounts receivable	241	194,328	89,576	_	284,145
Inventories	_	122,360	73,971	_	196,331
Other current assets	26,135	3,852	5,341	_	35,328
Total current assets	77,631	323,692	356,161		757,484
Property, plant and equipment	1,996	98,217	78,943	_	179,156
Goodwill	7,980	150,926	78,936	_	237,842
Investment in subsidiaries	1,450,306	271,037	180,807	(1,902,150)	_
Other intangibles	958	45,476	11,276	_	57,710
Other long term assets	(948)	310	24,546	_	23,908
Total Assets	\$1,537,923	\$ 889,658	\$ 730,669	\$(1,902,150)	\$1,256,100
Current liabilities	\$ (2,809)	\$ 228,029	\$ 100,004	\$	\$ 325,224
Intercompany	658,425	(731,920)	73,495	_	_
Long-term debt	150,000	135	_	_	150,135
Other long term liabilities	53,656	15,494	32,940	_	102,090
Total liabilities	859,272	(488,262)	206,439		577,449
Stockholders' equity	678,651	1,377,920	524,230	(1,902,150)	678,651
Total Liabilities and Stockholders' Equity	\$1,537,923	\$ 889,658	\$ 730,669	\$(1,902,150)	\$1,256,100

WESTINGHOUSE AIR BRAKE TECHNOLOGIES CORPORATION

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS—(Continued) FOR THE QUARTERLY PERIOD ENDED JUNE 30, 2008 (UNAUDITED)

Balance Sheet as of December 31, 2007:

In thousands	Parent	Guarantors	Non-Guarantors	Elimination	Consolidated
Cash and cash equivalents	\$ 82,911	\$ 10,004	\$ 141,774	\$ —	\$ 234,689
Accounts receivable	135	150,662	71,438	_	222,235
Inventories	_	108,958	67,019	_	175,977
Other current assets	24,703	3,530	4,633	_	32,866
Total current assets	107,749	273,154	284,864		665,767
Property, plant and equipment	2,493	100,806	79,138	_	182,437
Goodwill	7,980	151,297	73,316	_	232,593
Investment in subsidiaries	1,311,343	223,145	59,850	(1,594,338)	_
Other intangibles	1,354	46,602	10,717	_	58,673
Other long term assets	(1,526)	693	20,065		19,232
Total Assets	\$1,429,393	\$ 795,697	\$ 527,950	\$(1,594,338)	\$1,158,702
Current liabilities	\$ 1,576	\$ 203,938	\$ 89,721	\$ —	\$ 295,235
Intercompany	608,282	(644,920)	36,638	_	_
Long-term debt	150,000	177	_	_	150,177
Other long term liabilities	52,267	13,445	30,310	_	96,022
Total liabilities	812,125	(427,360)	156,669		541,434
Stockholders' equity	617,268	1,223,057	371,281	(1,594,338)	617,268
Total Liabilities and Stockholders' Equity	\$1,429,393	\$ 795,697	\$ 527,950	\$(1,594,338)	\$1,158,702

Income Statement for the Three Months Ended June 30, 2008:

In thousands	Parent	Guarantors	Non-Guarantors	Elimination(1)	Consolidated
Net sales	\$ —	\$ 287,801	\$ 141,222	\$ (38,829)	\$ 390,194
Cost of sales	(151)	(192,171)	(119,059)	29,688	(281,693)
Gross (loss) profit	(151)	95,630	22,163	(9,141)	108,501
Operating expenses	(11,047)	(30,692)	(10,840)	<u> </u>	(52,579)
Operating (loss) profit	(11,198)	64,938	11,323	(9,141)	55,922
Interest (expense) income, net	(4,718)	2,103	1,322	_	(1,293)
Other (expense) income, net	(188)	346	(854)	_	(696)
Equity earnings	61,738	9,022	_	(70,760)	_
Income (loss) from continuing operations before income					
tax	45,634	76,409	11,791	(79,901)	53,933
Income tax expense	(11,872)	(3,589)	(4,710)		(20,171)
Income (loss) from continuing operations	33,762	72,820	7,081	(79,901)	33,762
Discontinued operations (net of tax)	_	_	_	_	_
Net income (loss)	\$ 33,762	\$ 72,820	\$ 7,081	\$ (79,901)	\$ 33,762

⁽¹⁾ Includes elimination of gross profit realized with certain intercompany transactions between Guarantor and Non-Guarantor subsidiaries.

WESTINGHOUSE AIR BRAKE TECHNOLOGIES CORPORATION

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS—(Continued) FOR THE QUARTERLY PERIOD ENDED JUNE 30, 2008 (UNAUDITED)

Income Statement for the Three Months Ended June 30, 2007:

In thousands	Parent	Guarantors	Non-Guarantors	Elimination(1)	Consolidated
Net sales	\$ —	\$ 245,363	\$ 109,024	\$ (28,665)	\$ 325,722
Cost of sales	(628)	(164,474)	(88,878)	19,108	(234,872)
Gross (loss) profit	(628)	80,889	20,146	(9,557)	90,850
Operating expenses	(12,078)	(21,835)	(11,020)	_	(44,933)
Operating (loss) profit	(12,706)	59,054	9,126	(9,557)	45,917
Interest (expense) income, net	(4,348)	2,791	871	_	(686)
Other (expense) income, net	(202)	601	(2,036)	_	(1,637)
Equity earnings	54,858	7,237		(62,095)	
Income (loss) from continuing operations before income					
tax	37,602	69,683	7,961	(71,652)	43,594
Income tax expense	(9,502)	(3,636)	(2,331)		(15,469)
Income (loss) from continuing operations	28,100	66,047	5,630	(71,652)	28,125
Income (loss) from discontinued operations (net of tax)	30	_	(25)	_	5
Net income (loss)	\$ 28,130	\$ 66,047	\$ 5,605	\$ (71,652)	\$ 28,130

(1) Includes elimination of gross profit realized with certain intercompany transactions between Guarantor and Non-Guarantor subsidiaries.

Income Statement for the Six Months Ended June 30, 2008:

Parent	Guarantors	Non-Guarantors	Elimination(1)	Consolidated
\$ —	\$ 575,032	\$ 267,423	\$ (68,934)	\$ 773,521
(370)	(388,948)	(223,292)	52,805	(559,805)
(370)	186,084	44,131	(16,129)	213,716
(26,992)	(55,643)	(21,273)		(103,908)
(27,362)	130,441	22,858	(16,129)	109,808
(9,543)	4,494	2,275	_	(2,774)
(411)	1,938	(2,606)	_	(1,079)
126,752	16,057	_	(142,809)	_
89,436	152,930	22,527	(158,938)	105,955
(23,164)	(7,333)	(9,183)		(39,680)
66,272	145,597	13,344	(158,938)	66,275
		(3)		(3)
\$ 66,272	\$ 145,597	\$ 13,341	\$ (158,938)	\$ 66,272
	\$ — (370) (370) (26,992) (27,362) (9,543) (411) 126,752 89,436 (23,164) 66,272	\$ — \$ 575,032 (370) (388,948) (370) 186,084 (26,992) (55,643) (27,362) 130,441 (9,543) 4,494 (411) 1,938 126,752 16,057 89,436 152,930 (23,164) (7,333) 66,272 145,597 —	\$ — \$ 575,032 \$ 267,423 (370) (388,948) (223,292) (370) 186,084 44,131 (26,992) (55,643) (21,273) (27,362) 130,441 22,858 (9,543) 4,494 2,275 (411) 1,938 (2,606) 126,752 16,057 — 89,436 152,930 22,527 (23,164) (7,333) (9,183) 66,272 145,597 13,344 — (3)	\$ — \$ 575,032 \$ 267,423 \$ (68,934) (370) (388,948) (223,292) 52,805 (370) 186,084 44,131 (16,129) (26,992) (55,643) (21,273) — (27,362) 130,441 22,858 (16,129) (9,543) 4,494 2,275 — (411) 1,938 (2,606) — 126,752 16,057 — (142,809) 89,436 152,930 22,527 (158,938) (23,164) (7,333) (9,183) — 66,272 145,597 13,344 (158,938) — — (3) —

⁽¹⁾ Includes elimination of gross profit realized with certain intercompany transactions between Guarantor and Non-Guarantor subsidiaries.

WESTINGHOUSE AIR BRAKE TECHNOLOGIES CORPORATION

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS—(Continued) FOR THE QUARTERLY PERIOD ENDED JUNE 30, 2008 (UNAUDITED)

Income Statement for the Six Months Ended June 30, 2007:

<u>In thousands</u>	Parent	Guarantors	Non-Guarantors	Elimination(1)	Consolidated
Net sales	\$ —	\$ 487,742	\$ 213,574	\$ (61,330)	\$ 639,986
Cost of sales	417	(326,524)	(178,737)	42,274	(462,570)
Gross profit (loss)	417	161,218	34,837	(19,056)	177,416
Operating expenses	(23,267)	(44,862)	(21,106)		(89,235)
Operating (loss) profit	(22,850)	116,356	13,731	(19,056)	88,181
Interest (expense) income, net	(8,415)	5,500	1,446	_	(1,469)
Other (expense) income, net	(601)	1,039	(2,884)	_	(2,446)
Equity earnings	103,439	8,943		(112,382)	
Income (loss) from continuing operations before income					
tax	71,573	131,838	12,293	(131,438)	84,266
Income tax expense	(17,951)	(7,686)	(4,950)		(30,587)
Income (loss) from continuing operations	53,622	124,152	7,343	(131,438)	53,679
Income (loss) from discontinued operations (net of tax)	30		(57)		(27)
Net income (loss)	\$ 53,652	\$ 124,152	\$ 7,286	\$ (131,438)	\$ 53,652

⁽¹⁾ Includes elimination of gross profit realized with certain intercompany transactions between Guarantor and Non-Guarantor subsidiaries.

Condensed Statement of Cash Flows for the Six Months Ended June 30, 2008:

			Non-		
In thousands	Parent	Guarantors	Guarantors	Elimination	Consolidated
Net cash (used for) provided by operating activities	\$ (4,416)	\$ 141,926	\$ 51,605	\$(158,938)	\$ 30,177
Net cash used for investing activities	(4,483)	(3,134)	(171)	_	(7,788)
Net cash (used for) provided by financing activities	(22,757)	(145,644)	(13,341)	158,938	(22,804)
Effect of changes in currency exchange rates			7,406		7,406
(Decrease) increase in cash	(31,656)	(6,852)	45,499	_	6,991
Cash, beginning of year	82,911	10,004	141,774		234,689
Cash, end of period	\$ 51,255	\$ 3,152	\$187,273	\$ —	\$ 241,680

WESTINGHOUSE AIR BRAKE TECHNOLOGIES CORPORATION

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS—(Continued) FOR THE QUARTERLY PERIOD ENDED JUNE 30, 2008 (UNAUDITED)

Condensed Statement of Cash Flows for the Six Months Ended June 30, 2007:

			Non-		
In thousands	Parent	Guarantors	Guarantors	Elimination	Consolidated
Net cash provided by (used for) operating activities	\$ 7,940	\$ 131,479	\$ 18,242	\$(131,438)	\$ 26,223
Net cash used for investing activities	(73,335)	(4,525)	(3,607)	_	(81,467)
Net cash provided by (used for) financing activities	1,305	(124,152)	(7,286)	131,438	1,305
Effect of changes in currency exchange rates			7,186		7,186
(Decrease) increase in cash	(64,090)	2,802	14,535	_	(46,753)
Cash, beginning of year	106,233	(231)	81,977		187,979
Cash, end of period	\$ 42,143	\$ 2,571	\$ 96,512	\$ —	\$ 141,226

17. OTHER EXPENSE, NET

The components of other expense are as follows:

	Three Months Ended June 30,		Six Months Ende June 30,	
In thousands_	2008	2007	2008	2007
Foreign currency loss	\$ 854	\$ 1,345	\$ 1,196	\$ 2,095
Other miscellaneous (income) expense	(158)	292	(117)	351
Total other expense	\$ 696	\$ 1,637	\$ 1,079	\$ 2,446

Item 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

The following discussion should be read in conjunction with the information in the unaudited condensed consolidated financial statements and notes thereto included herein and Westinghouse Air Brake Technologies Corporation's Financial Statements and Management's Discussion and Analysis of Financial Condition and Results of Operations included in its 2007 Annual Report on Form 10-K, filed February 29, 2008.

OVERVIEW

Wabtec is one of the world's largest providers of value-added, technology-based products and services for the global rail industry. Our products are found on virtually all U.S. locomotives, freight cars and passenger transit vehicles, as well as in more than 100 countries throughout the world. Our products enhance safety, improve productivity and reduce maintenance costs for customers, and many of our core products and services are essential in the safe and efficient operation of freight rail and passenger transit vehicles. Wabtec is a global company with operations in 12 countries. In the first six months of 2008, about 42% of the Company's revenues came from customers outside the U.S.

Management Review and Future Outlook

Wabtec's long-term financial goals are to generate free cash flow in excess of net income, maintain a strong credit profile while minimizing our overall cost of capital, increase margins through strict attention to cost controls, and increase revenues through a focused growth strategy, including global and market expansion, new products and technologies, aftermarket products and services, and acquisitions. In addition, Management evaluates the Company's short-term operational performance through measures such as quality and on-time delivery.

The Company monitors a variety of factors and statistics to gauge market activity. The freight rail industry is largely driven by general economic conditions, which can cause fluctuations in rail traffic. Based on those fluctuations, railroads can increase or decrease purchases of new locomotive and freight cars.

In 2008, the Company expects conditions to remain generally favorable in its passenger transit rail markets and expects its freight rail markets to remain stable, but subject to changes in overall economic conditions. Through mid-July 2008, revenue ton-miles increased 1.6%, compared to the same period in 2007. Demand for new locomotives is expected to be slightly higher than in 2007, while demand for new freight cars is expected to be lower. Less than 20% of the Company's revenues are directly related to deliveries of new freight cars. At June 30, 2008, the industry backlog of freight cars ordered was 61,573, compared to 65,223 at the end of the prior quarter. In the passenger transit rail market, the Company believes that increases in ridership and federal funding will continue to have a positive effect on the demand for new equipment and aftermarket parts. In addition, the Company has a strong backlog of transit-related projects, some of which are expected to generate increased revenues in 2008.

In 2008 and beyond, we will continue to face many challenges, including increased costs for raw materials, higher costs for medical and insurance premiums, and foreign currency fluctuations. In addition, we face general economic risks, as well as the risk that our customers could curtail spending on new and existing equipment. Risks associated with our four-point growth strategy include the level of investment that customers are willing to make in new technologies developed by the industry and the Company, and risks inherent in global expansion. When necessary, we will modify our financial and operating strategies to reflect changes in market conditions and risks.

Since 2006, Wabtec has downsized its Canadian operations by moving certain products to lower-cost facilities and outsourcing. In the Freight segment, Wabtec recorded charges of \$3.1 million and \$3.0 million for

the three months ended June 30, 2008 and 2007, respectively. Wabtec recorded charges of \$3.1 million and \$4.0 million for the six months ended June 30, 2008 and 2007, respectively. Total expenses for restructuring and other expenses recorded since 2006 have been \$14.9 million, comprised of the \$4.8 million for employee severance costs for approximately 370 salaried and hourly employees; \$4.8 million of pension and postretirement benefit curtailment for those employees; \$4.8 million related to asset impairments for structures, machinery, and equipment; and \$541,000 for goodwill impairment specific to one location. The goodwill impairment was recorded as amortization expense and most of the other charges were recorded to cost of sales. Severance costs are contractual liabilities and payment is dependent on the waiver by or expiration of certain seniority rights of those employees. As of June 30, 2008, \$2.3 million of this amount had been paid. In the Transit segment, Wabtec recorded charges of \$1.1 million for the six months ended June 30, 2007. Total expenses for restructuring and other expenses recorded since 2006 have been \$1.5 million, comprised of severance, pension, and asset impairment. All charges were recorded to cost of sales. As of June 30, 2008, all but \$68,000 of these costs has been paid.

RESULTS OF OPERATIONS

The following table shows our Consolidated Statements of Operations for the periods indicated.

		Three Months Ended June 30,				hs Ended 2 30,
In millions	2008	2007	2008	2007		
Net sales	\$ 390.2	\$ 325.7	\$ 773.5	\$ 640.0		
Cost of sales	(281.7)	(234.9)	(559.8)	(462.6)		
Gross profit	108.5	90.8	213.7	177.4		
Selling, general and administrative expenses	(42.1)	(34.9)	(82.5)	(69.9)		
Engineering expenses	(9.6)	(9.0)	(19.6)	(17.8)		
Amortization expense	(0.9)	(1.0)	(1.8)	(1.5)		
Total operating expenses	(52.6)	(44.9)	(103.9)	(89.2)		
Income from operations	55.9	45.9	109.8	88.2		
Interest income (expense), net	(1.3)	(0.7)	(2.7)	(1.5)		
Other expense, net	(0.7)	(1.6)	(1.1)	(2.4)		
Income from continuing operations before income taxes	53.9	43.6	106.0	84.3		
Income tax expense	(20.1)	(15.5)	(39.7)	(30.6)		
Income from continuing operations	33.8	28.1	66.3	53.7		
Discontinued operations						
Net income	\$ 33.8	\$ 28.1	\$ 66.3	\$ 53.7		

SECOND QUARTER 2008 COMPARED TO SECOND QUARTER 2007

The following table summarizes the results of operations for the period:

	Three	Three months ended June 30,		
In thousands	2008	2007	Percent Change	
Net sales	\$390,194	\$325,722	19.8%	
Income from operations	55,922	45,917	21.8%	
Net income	33,762	28.130	20.0%	

Net sales increased by \$64.5 million to \$390.2 million from \$325.7 million for the three months ended June 30, 2008 and 2007, respectively. The increase is primarily due to internal growth from increased sales of \$13.7 million for combined freight and transit brake products, \$9.0 million for remanufacturing, overhaul and

build of cars and locomotives in the freight and transit groups, \$8.8 million for freight electronics and specialty products, \$5.2 million for other transit related products and \$5.2 million for other products. Net sales increased \$15.4 million from an acquisition completed in the second quarter of 2007. The Company also realized net sales improvements of \$7.2 million due to foreign exchange, but net earnings were generally not impacted by foreign exchange. Net income for the three months ended June 30, 2008 was \$33.8 million or \$0.69 per diluted share. Net income for the three months ended June 30, 2007 was \$28.1 million or \$0.57 per diluted share. Net income improved primarily due to sales increases and consistent operating costs.

Net sales by Segment The following table shows the Company's net sales by business segment:

	Three months	Three months ended	
	June 30	June 30,	
In thousands	2008	2007	
Freight Group	\$ 199,631	\$ 180,986	
Transit Group	190,563	144,736	
Net sales	\$ 390,194	\$ 325,722	

Freight Group sales increased by \$18.7 million or 10.3% primarily due to increased sales of \$8.8 million for freight electronics and specialty products, \$3.6 for brake products, \$5.4 for other products, and \$2.5 million due to foreign exchange. Offsetting those increases was a decrease of \$1.6 million for remanufacturing, overhaul and build of cars and locomotives. Transit Group sales increased by \$45.8 million or 31.7% primarily due to increased sales of \$10.6 million for remanufacturing, overhaul and build of cars and locomotives, \$10.1 for brake products, \$5.2 for other transit related products, \$15.4 million from an acquisition completed in the second quarter of 2007, and \$4.7 million due to foreign exchange, partially offset by a \$0.2 million decrease for other products.

Gross profit Gross profit increased to \$108.5 million in the second quarter of 2008 compared to \$90.8 million in the same period of 2007. Gross profit is dependent on a number of factors including pricing, sales volume and product mix. In the second quarter of 2008, gross profit, as a percentage of sales, was 27.8% compared to 27.9% the same period of 2007. The gross profit percentage was flat due to the changing mix of revenues from Freight to Transit as Transit margins tend to be lower than Freight. This was offset by ongoing efficiency and cost saving initiatives.

The provision for warranty expense was \$710,000 higher for the second quarter of 2008 compared to the same period of 2007, which had a negative impact on gross profit. These provisions, which are established based on historical claims as a percentage of revenue, were slightly higher due to increased sales resulting in higher reserve compared to the second quarter of 2007.

Operating expenses The following table shows our operating expenses:

	Three	Three months ended June 30,		
In thousands	2008	2007	Percent Change	
Selling, general and administrative expenses	\$42,036	\$34,915	20.4%	
Engineering expenses	9,631	9,026	6.7%	
Amortization expense	912	992	(8.1)%	
Total operating expenses	\$52,579	\$44,933	17.0%	

Selling, general, and administrative expenses increased \$7.1 million in the second quarter of 2008 compared to the same period of 2007 due to \$3.3 million from increased legal expenses and \$1.5 million from an acquisition completed during the second quarter of 2007. Engineering expenses increased by \$605,000 in the second quarter of 2008 compared to the same period of 2007 mostly due to the same acquisition. Total operating expenses were 13.5% and 13.8% of sales for the second quarters of 2008 and 2007, respectively.

Income from operations Income from operations totaled \$55.9 million (or 14.3% of sales) in the second quarter of 2008 compared with \$45.9 million (or 14.1% of sales) in the same period of 2007. Income from operations improved primarily due to sales increases and higher operating margins.

Interest expense, net Interest expense, net increased \$607,000 in the second quarter of 2008 compared to the same period of 2007 primarily due to the Company's overall lower cash balances, resulting in lower interest income.

Other expense, net The Company recorded foreign exchange expense of \$854,000 and \$1.3 million in the second quarter of 2008 and 2007, respectively, due to the effect of currency exchange rate changes on intercompany transactions that are non U.S. dollar denominated amounts and charged or credited to earnings.

Income taxes The overall effective income tax rate was 37.4% and 35.5% for second quarters of 2008 and 2007, respectively. The increase in effective tax rate is primarily due to approximately \$1.3 million of tax benefit recognized during the second quarter of 2007 related to the reversal of a state deferred tax valuation allowance.

Net income Net income for the second quarter of 2008 increased \$5.6 million, compared with the same period of 2007. Net income improved primarily due to sales increases and higher operating margins.

FIRST SIX MONTHS OF 2008 COMPARED TO FIRST SIX MONTHS OF 2007

The following table summarizes the results of operations for the period:

	Six	Six months ended June 3		
In thousands	2008	2007	Percent Change	
Net sales	\$773,521	\$639,986	20.9%	
Income from operations	109,808	88,181	24.5%	
Net income	66,272	53,652	23.5%	

Net sales increased by \$133.5 million to \$773.5 million from \$640.0 million for the six months ended June 30, 2008 and 2007, respectively. The increase is primarily due to internal growth from increased sales of \$31.8 million for remanufacturing, overhaul and build of cars and locomotives in the freight and transit groups, \$22.5 for combined freight and transit brake products, \$15.1 million for freight electronics and specialty products, \$9.2 million for other transit related products and \$6.6 million for other products. Net sales increased \$33.0 million from an acquisition completed in the second quarter of 2007. The Company also realized net sales improvements of \$15.3 million due to foreign exchange, but net earnings were generally not impacted by foreign exchange. Net income for the six months ended June 30, 2008 was \$66.3 million or \$1.35 per diluted share. Net income for the six months ended June 30, 2007 was \$53.7 million or \$1.09 per diluted share. Net income improved primarily due to sales increases and consistent operating costs.

Net sales by Segment The following table shows the Company's net sales by business segment:

	Six months end	Six months ended	
	June 30,		
In thousands	2008	2007	
Freight Group	\$ 391,397	365,653	
Transit Group	382,124	274,333	
Net sales	\$ 773,521	639,986	

Freight Group sales increased by \$25.7 million or 7.0% primarily due to increased sales of \$15.1 for freight electronics and specialty products, \$7.4 for brake products, \$6.8 for other products and \$5.2 million due to foreign exchange. Offsetting those increases was a decrease of \$8.8 million for remanufacturing, overhaul and build of

cars and locomotives. Transit Group sales increased by \$107.8 million or 39.3% primarily due to increased sales of \$40.6 million for remanufacturing, overhaul and build of cars and locomotives, \$15.1 for brake products, \$9.2 for other transit related products, \$33.0 million from an acquisition completed in the second quarter of 2007, and \$10.1 million due to foreign exchange, partially offset by a \$0.2 million decrease for other products.

Gross profit Gross profit increased to \$213.7 million for the first six months of 2008 compared to \$177.4 million in the same period of 2007. Gross profit is dependent on a number of factors including pricing, sales volume and product mix. For the first six months of 2008, gross profit, as a percentage of sales, was 27.6% compared to 27.7% the same period of 2007. The gross profit percentage was flat due to the changing mix of revenues from Freight to Transit as Transit margins tend to be lower than Freight. This was offset by ongoing efficiency and cost saving initiatives.

In addition, for the first six months of 2008 the provision for warranty expense was \$4.4 million higher compared to the same period of 2007, which had a negative impact on gross profit and margin. These increases relate to specific reserves. In addition, these provisions, which are established based on historical claims as a percentage of revenue, were slightly higher due to increased sales resulting in a higher reserve compared to the first six months of 2007.

Operating expenses The following table shows our operating expenses:

	Six m	,	
In thousands	2008	2007	Percent Change
Selling, general and administrative expenses	\$ 82,481	\$69,860	18.1%
Engineering expenses	19,612	17,842	9.9%
Amortization expense	1,815	1,533	18.4%
Total operating expenses	\$103,908	\$89,235	16.4%

Selling, general, and administrative expenses increased \$12.6 million for the first six months of 2008 compared to the same period of 2007 due to \$5.5 million from increased legal expenses and \$4.2 million from an acquisition completed during the second quarter of 2007 offset by \$2.5 million for expenses related to the Bombardier settlement that were recorded in the first quarter of 2007. Engineering expenses increased by \$1.8 million for the first six months of 2008 compared to the same period of 2007 mostly due to the same acquisition. Total operating expenses were 13.4% and 13.9% of sales for the first quarters of 2008 and 2007, respectively.

Income from operations Income from operations totaled \$109.8 million (or 14.2% of sales) for the first six months of 2008 compared with \$88.2 million (or 13.8% of sales) in the same period of 2007. Income from operations improved primarily due to sales increases and higher operating margins.

Interest expense, net Interest expense, net increased \$1.3 million for the first six months of 2008 compared to the same period of 2007 primarily due to the Company's overall lower cash balances, resulting in lower interest income.

Other expense, net The Company recorded foreign exchange expense of \$1.2 million and \$2.1 million during the first six months of 2008 and 2007, respectively, due to the effect of currency exchange rate changes on intercompany transactions that are non U.S. dollar denominated amounts and charged or credited to earnings.

Income taxes The overall effective income tax rate was 37.4% and 36.3% for the six months ended June 30, 2008 and 2007, respectively. The increase in effective tax rate is primarily due to approximately \$1.3 million of tax benefit recognized during the second quarter of 2007 related to the reversal of a state deferred tax valuation allowance.

Net income Net income for the first six months of 2008 increased \$12.6 million, compared with the same period of 2007. Net income improved primarily due to sales increases and higher operating margins.

Liquidity and Capital Resources

Liquidity is provided primarily by operating cash flow and borrowings under the Company's unsecured credit facility with a consortium of commercial banks ("credit agreement"). The following is a summary of selected cash flow information and other relevant data:

		Six months ended June 30,	
	June		
In thousands	2008	2007	
Cash provided by (used for):			
Operating activities	\$ 30,177	\$ 26,223	
Investing activities	(7,788)	(81,467)	
Financing activities	(22,804)	1,305	
Increase (decrease) in cash	\$ 6,991	\$(46,753)	

Operating activities Cash provided by operations in the first six months of 2008 was \$30.2 million as compared to cash provided by operations of \$26.2 million for the same period of 2007. This \$4.0 million increase was the result of increased earnings offset by certain changes in operating assets and liabilities. Net income for the Company increased \$12.6 million primarily as a result of increased sales. Accounts receivable decreased operating cash flows by \$46.2 million due to large customer billings for certain transit contracts. Accrued liabilities and customer deposits increased operating cash flows by \$48.2 million due to the progress of customer billing for certain contracts. Accrued taxes used cash of \$6.6 million primarily due to the adoption of FIN48 during the first quarter of 2007.

Investing activities In the first six months of 2008 and 2007, cash used in investing activities was \$7.8 million and \$81.5 million, respectively. During the second quarter of 2007, Wabtec acquired Ricon Corporation, a manufacturer of wheelchair lifts and ramps, for \$73.3 million. Capital expenditures were \$7.9 million and \$8.3 million in the first six months of 2008 and 2007, respectively.

On June 30, 2008, subsequent to the close of our four-four-five accounting quarter, the Company acquired POLI S.p.A., a European based manufacturer of rail braking equipment, for a net price of €53 million (about \$80 million USD) in cash. POLI's products include brake discs for high speed applications, as well as tread brake units and pneumatic brake valves that meet International Union of Railways (UIC) standards. POLI will operate as a business of Wabtec's Transit Group.

Financing activities In the first six months of 2008, cash used by financing activities was \$22.8 million, which included \$2.9 million of proceeds from the exercise of stock options and other benefit plan activity, offset by \$969,000 of dividend payments and \$24.7 million for the repurchase of 718,100 shares of stock. In the first six months of 2007, cash provided by financing activities was \$1.3 million, which included \$5.7 million of proceeds from the exercise of stock options and other benefit plans, offset by \$986,000 of dividend payments and \$3.4 million for the repurchase of 92,700 shares of stock.

The following table shows outstanding indebtedness at June 30, 2008 and December 31, 2007.

In thousands	June 30, 2008	December 31, 2007
6.875% senior notes, due 2013	\$150,000	\$ 150,000
Capital Leases	203	250
Total	150,203	150,250
Less—current portion	68	73
Long-term portion	\$150,135	\$ 150,177

Cash balance at June 30, 2008 and December 31, 2007 was \$241.7 million and \$234.7 million, respectively.

Refinancing Credit Agreement

In January 2004, the Company refinanced its existing unsecured revolving credit agreement with a consortium of commercial banks. This "Refinancing Credit Agreement" provided a \$175 million five-year revolving credit facility expiring in January 2009. In November 2005, the Company entered into an amendment to the Refinancing Credit Agreement which, among other things, extended the expiration of the agreement until December 2010. The Company entered into an amendment to its Refinancing Credit Agreement in February 2007 which permits the Company to complete any acquisitions without prior approval of the bank consortium as long as certain financial parameters and ratios are met. At June 30, 2008, the Company had available bank borrowing capacity, net of \$58.7 million of letters of credit, of approximately \$116.3 million, subject to certain financial covenant restrictions.

Refinancing Credit Agreement borrowings bear variable interest rates indexed to the indices described below. The Company did not borrow under the Refinancing Credit Agreement during the six months ended June 30, 2008 or during the year ended December 31, 2007.

Under the Refinancing Credit Agreement, the Company may elect a base interest rate or an interest rate based on the London Interbank Offered Rates of Interest ("LIBOR"). The base interest rate is the greater of LaSalle Bank National Association's prime rate or the federal funds effective rate plus 0.5% per annum. The LIBOR rate is based on LIBOR plus a margin that ranges from 62.5 to 175 basis points depending on the Company's consolidated total indebtedness to cash flow ratios. The current margin is 62.5 basis points.

The Refinancing Credit Agreement limits the Company's ability to declare or pay cash dividends and prohibits the Company from declaring or making other distributions, subject to certain exceptions. The Refinancing Credit Agreement contains various other covenants and restrictions including the following limitations: incurrence of additional indebtedness; mergers, consolidations and sales of assets and acquisitions; additional liens; sale and leasebacks; permissible investments, loans and advances; certain debt payments; capital expenditures; and imposes a minimum interest expense coverage ratio and a maximum debt to cash flow ratio.

The Refinancing Credit Agreement contains customary events of default, including payment defaults, failure of representations or warranties to be true in any material respect, covenant defaults, defaults with respect to other indebtedness of the Company, bankruptcy, certain judgments against the Company, ERISA defaults and "change of control" of the Company. The Refinancing Credit Agreement includes the following covenants: a minimum interest coverage ratio of 3, maximum debt to cash flow ratio of 3.25 and a minimum net worth of \$180 million plus 50% of consolidated net income since September 30, 2003. The Company is in compliance with these measurements and covenants and expects that these measurements will not be any type of limiting factor in executing our operating activities.

6.875% Senior Notes Due August 2013

In August 2003, the Company issued \$150 million of Senior Notes due in 2013 ("the Notes"). The Notes were issued at par. Interest on the Notes accrues at a rate of 6.875% per annum and is payable semi-annually on January 31 and July 31 of each year. The proceeds were used to repay debt outstanding under the Company's existing credit agreement, and for general corporate purposes. The principal balance is due in full at maturity.

The Company believes, based on current levels of operations and forecasted earnings, cash flow and liquidity will be sufficient to fund its working capital and capital equipment needs as well as to meet its debt service requirements. If the Company's sources of funds were to fail to satisfy the Company's cash requirements, the Company may need to refinance its existing debt or obtain additional financing. There is no assurance that such new financing alternatives would be available, and, in any case, such new financing, if available, would be expected to be more costly and burdensome than the debt agreements currently in place.

On July 31, 2006, the Board of Directors authorized the repurchase of up to \$50 million of the Company's outstanding shares. On February 20, 2008, the Board of Directors authorized the repurchase of up to an additional \$100 million of the Company's outstanding shares. During the first quarter of 2008, the Company completed the \$50 million authorization made in 2006. Cumulative purchases under both plans have totaled \$61.4 million, leaving \$88.6 million under the authorization.

The Company intends to purchase these shares on the open market or in negotiated or block trades. No time limit was set for the completion of the program which qualifies under the Refinancing Credit Agreement, as well as the 6.7/8% Senior Notes currently outstanding.

During 2007, 509,800 shares were repurchased at an average price of \$35.06 per share. During the first quarter of 2008, the Company repurchased 712,900 shares of Wabtec stock at an average price of \$34.29 per share. During the second quarter of 2008, the Company repurchased 5,200 shares of Wabtec stock at an average price of \$35.97 per share. All purchases were on the open market.

Contractual Obligations and Off-Balance Sheet Arrangements

Since the adoption of FIN 48, the Company has recognized a total liability of \$17.7 million for unrecognized tax benefits. The Company estimates that \$3.0 million to \$4.0 million of the total unrecognized tax benefits relate to uncertain tax positions in various taxing jurisdictions that may be resolved within the next 12 months. At this time, the Company is unable to make a reasonably reliable estimate of the timing of cash settlement for the remaining balances due to the uncertainty of the timing and outcome of its audits and other factors.

Since December 31, 2007, there have been no other significant changes in the total amount of the Company's contractual obligations or the timing of cash flows in accordance with those obligations, as reported in the Company's Annual Report on Form 10-K for the year ended December 31, 2007.

Forward Looking Statements

We believe that all statements other than statements of historical facts included in this report, including certain statements under "Business" and "Management's Discussion and Analysis of Financial Condition and Results of Operations," may constitute forward-looking statements. We have based these forward-looking statements on our current expectations and projections about future events. Although we believe that our assumptions made in connection with the forward-looking statements are reasonable, we cannot assure you that our assumptions and expectations are correct.

These forward-looking statements are subject to various risks, uncertainties and assumptions about us, including, among other things:

Economic and industry conditions

- materially adverse changes in economic or industry conditions generally or in the markets served by us, including North America, South America, Europe, Australia and Asia;
- demand for freight cars, locomotives, passenger transit cars, buses and related products and services;
- reliance on major original equipment manufacturer customers;
- original equipment manufacturers' program delays;
- · demand for services in the freight and passenger rail industry;
- demand for our products and services;
- orders either being delayed, canceled, not returning to historical levels, or reduced or any combination of the foregoing;

- · consolidations in the rail industry;
- · continued outsourcing by our customers; industry demand for faster and more efficient braking equipment; or
- fluctuations in interest rates and foreign currency exchange rates;
- · availability of credit;

Operating factors

- · supply disruptions;
- · technical difficulties;
- · changes in operating conditions and costs;
- increases in raw material costs;
- successful introduction of new products;
- performance under material long-term contracts;
- · labor relations;
- completion and integration of acquisitions;
- the development and use of new technology; or
- the integration of recently completed or future acquisitions.

Competitive factors

the actions of competitors;

Political/governmental factors

- political stability in relevant areas of the world;
- future regulation/deregulation of our customers and/or the rail industry;
- levels of governmental funding on transit projects, including for some of our customers;
- political developments and laws and regulations; or
- the outcome of our existing or any future legal proceedings, including litigation involving our principal customers and any litigation with respect to environmental, asbestos-related matters and pension liabilities; and

Transaction or commercial factors

• the outcome of negotiations with partners, governments, suppliers, customers or others.

Statements in this 10-Q apply only as of the date on which such statements are made, and we undertake no obligation to update any statement to reflect events or circumstances after the date on which the statement is made or to reflect the occurrence of unanticipated events.

Critical Accounting Policies

A summary of critical accounting policies is included in the Company's Annual Report on Form 10-K for the year ended December 31, 2007. In particular, judgment is used in areas such as accounts receivable and the allowance for doubtful accounts, inventories, goodwill and indefinite-lived intangibles, warranty reserves, pensions and postretirement benefits, income taxes and revenue recognition. There have been no significant changes in accounting policies since December 31, 2007.

Recent Accounting Pronouncements

See Note 2 of "Notes to Condensed Consolidated Financial Statements" included elsewhere in this report.

Item 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

Interest Rate Risk

In the ordinary course of business, Wabtec is exposed to risks that increases in interest rates may adversely affect funding costs associated with its variable-rate debt. There was no outstanding variable-rate debt at June 30, 2008.

Foreign Currency Exchange Risk

The Company has entered into foreign currency forward contracts to reduce the impact of changes in currency exchange rates. Forward contracts are agreements with a counterparty to exchange two distinct currencies at a set exchange rate for delivery on a set date at some point in the future. There is no exchange of funds until the delivery date. At the delivery date the Company can either take delivery of the currency or settle on a net basis.

At June 30, 2008, the Company had forward contracts for the sale of USD and the purchase of Euro with a notional value of €1.2 million Euro (or \$1.6 million USD), with an average exchange rate of \$1.33 USD per €1 Euro. These forward contracts are used to mitigate the variability in cash flows from the payment of liabilities denominated in currencies other than the USD. Since the Company does not treat these derivatives as hedges, the change in fair value of both the forward contracts and the related liabilities are recorded in the income statement. For the six months ended June 30, 2008, the Company recorded a fair value gain in the amount of \$295,000.

At June 30, 2008, the Company had forward contracts for the sale of USD and the purchase of South African Rand (ZAR). The Company concluded that these foreign currency forward contracts qualify for cash flow hedge accounting which permits the recording of the fair value of the forward contract and corresponding adjustment to other comprehensive income (loss), net of tax, on the balance sheet. As of June 30, 2008, the Company had forward contracts with a notional value of R18.7 million ZAR (or \$2.4 million USD), with an average exchange rate of R7.78 ZAR per USD, resulting in the recording of a current asset of \$68,000 and a corresponding offset in accumulated other comprehensive loss of \$43,000, net of tax.

At December 31, 2006, the Company had forward contracts for the sale of U.S. Dollars (USD) and the purchase of Canadian Dollars (CAD). The Company concluded that these foreign currency forward contracts qualify for cash flow hedge accounting which permits the recording of the fair value of the forward contract and corresponding adjustment to other comprehensive income (loss), net of tax, on the balance sheet. As of December 31, 2006, the Company had forward contracts with a notional value of \$48 million CAD (or \$42.7 million U.S.) with an average exchange rate of \$0.89 USD per \$1 CAD, resulting in the recording of a current liability of \$1.3 million and a corresponding offset in accumulated other comprehensive loss of \$825,000, net of tax. During 2007, these foreign currency forward contracts were settled.

We are also subject to certain risks associated with changes in foreign currency exchange rates to the extent our operations are conducted in currencies other than the U.S. dollar. For the six months of 2008, approximately 58% of Wabtec's net sales were to customers in the United States, 10% in Canada, 3% in Mexico, 5% in Australia, 2% in Germany, 12% in the United Kingdom, and 10% in other international locations.

Item 4. CONTROLS AND PROCEDURES

Wabtec's principal executive officer and its principal financial officer have evaluated the effectiveness of Wabtec's "disclosure controls and procedures," (as defined in Exchange Act Rule 13a-15(e)) as of June 30, 2008. Based upon their evaluation, the principal executive officer and principal financial officer concluded that Wabtec's disclosure controls and procedures are effective to provide reasonable assurance that information required to be disclosed by Wabtec in the reports filed or submitted by it under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the SEC's rules and forms, and to provide reasonable assurance that information required to be disclosed by Wabtec in such reports is accumulated and communicated to Wabtec's Management, including its principal executive officer and principal financial officer, as appropriate to allow timely decisions regarding required disclosure.

There was no change in Wabtec's "internal control over financial reporting" (as defined in Rule 13a-15(f) under the Exchange Act) that occurred during the quarter ended June 30, 2008, that has materially affected, or is reasonably likely to materially affect, Wabtec's internal control over financial reporting.

PART II—OTHER INFORMATION

Item 1. LEGAL PROCEEDINGS

There have been no material changes regarding the Company's commitments and contingencies as described in Note 18 of the Company's Annual Report on Form 10-K for the Year Ended December 31, 2007.

Item 1A. RISK FACTORS

There have been no material changes in our risk factors from those disclosed in our 2007 Annual Report on Form 10-K.

Item 2. UNREGISTERED SALES OF EQUITY SECURITIES AND USE OF PROCEEDS

On July 31, 2006, the Board of Directors authorized the repurchase of up to \$50 million of the Company's outstanding shares. On February 20, 2008, the Board of Directors authorized the repurchase of up to an additional \$100 million of the Company's outstanding shares. During the first quarter of 2008, the Company's purchases exhausted the \$50 million authorization made in 2006.

The Company intends to purchase these shares on the open market or in negotiated or block trades. No time limit was set for the completion of the program which qualifies under the Refinancing Credit Agreement, as well as the 6.7/8% Senior Notes currently outstanding.

During 2007, 509,800 shares were repurchased at an average price of \$35.06 per share. During the first quarter of 2008, the Company repurchased 712,900 shares of Wabtec stock at an average price of \$34.29 per share. During the second quarter of 2008, the Company repurchased 5,200 shares of Wabtec stock at an average price of \$35.97 per share. All purchases were on the open market.

Period	Total Number of Shares Purchased	Average Price Paid per Share	Number of Shares Purchased for Announced Program	Approximate Dollar Value of Shares that May Yet Be Purchased
March 30, 2008 to April 26, 2008	5,200	\$35.97	5,200	\$88,584,329
April 27, 2008 to May 24, 2008	_	_	_	\$88,584,329
May 25, 2008 to June 28, 2008	_	_	_	\$88,584,329
Total	5,200	\$35.97	5,200	\$88,584,329

Item 4. SUBMISSION OF MATTERS TO A VOTE OF SECURITY HOLDERS

The Annual Meeting of Stockholders of Wabtec was held May 14, 2008. One matter was considered and voted upon at the Annual Meeting: the election of three persons to serve as Directors.

Election of Directors Nominations of Emilio A. Fernandez, Lee B. Foster II, and James V. Napier to serve as Directors for a term expiring in 2011 were considered and each nominee was elected.

The voting was as follows:

			Votes	Votes
Nominee	<u>Title</u>	Votes For	Against	Withheld
Emilio A. Fernandez	Director and Vice Chairman	41,124,193	_	2,972,451
Lee B. Foster II	Director	40,831,715	_	3,264,929
James V. Napier	Director	41,532,897		2,563,747

The terms of office of Robert J. Brooks, Brian P. Hehir, Michael W. D. Howell, William E. Kassling, Albert J. Neupaver, Nickolas W. Vande Steeg, and Gary C. Valade continued after the Annual Meeting. They will serve as directors until their terms expire and until their successors have been duly elected and qualified.

Item 6. EXHIBITS

The following exhibits are being filed with this report:

- 3.1 Restated Certificate of Incorporation of the Company dated January 30, 1995, as amended March 30, 1995.
- 3.2 Amended and Restated By-Laws of the Company, effective December 13, 2007.
- 10.1 Stock Purchase Agreement dated May 16, 2008 between the Company and Polinvest S.r.l.
- 31.1 Rule 13a-14(a) Certification of Chief Executive Officer.
- 31.2 Rule 13a-14(a) Certification of Chief Financial Officer.
- 32.1 Section 1350 Certification of Chief Executive Officer and Chief Financial Officer.

SIGNATURES

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, the Company has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

WESTINGHOUSE AIR BRAKE TECHNOLOGIES CORPORATION

By: /s/ ALVARO GARCIA-TUNON

Alvaro Garcia-Tunon, Senior Vice President, Chief Financial Officer and Secretary

DATE: August 8, 2008

EXHIBIT INDEX

Exhibit Number 3.1	<u>Description and Method of Filing</u> Restated Certificate of Incorporation of the Company dated January 30, 1995, as amended March 30, 1995, filed as an exhibit to the Company's Registration Statement on Form S-1 (No. 33-90866), and incorporated herein by reference.
3.2	Amended and Restated By-Laws of the Company, effective December 13, 2007, filed as Exhibit 3.1 to Form 8-K filed on December 14, 2007, and incorporated herein by reference.
10.1	Stock Purchase Agreement dated May 16, 2008 between the Company and Polinvest S.r.l., filed herewith.
31.1	Rule 13a-14(a) Certification of Chief Executive Officer, filed herewith.
31.2	Rule 13a-14(a) Certification of Chief Financial Officer, filed herewith.
32.1	Section 1350 Certification of Chief Executive Officer and Chief Financial Officer, filed herewith.

STOCK PURCHASE AGREEMENT

This STOCK PURCHASE AGREEMENT (this "**Agreement**") is entered into this 16th day of May, 2008 (hereinafter referred to as the "**Signing Date**") by and between

<u>WESTINGHOUSE AIR BRAKE TECHNOLOGIES CORPORATION</u>, a company duly organized and existing under the laws of the State of Delaware (U.S.A.), with registered office in 1001 Air Brake Avenue, Wilmerding, PA 15148, U.S.A. (hereinafter referred to as "**Purchaser**"), hereby represented by Mr. R. Mark Cox, duly authorized pursuant to the certificate of the Assistant Secretary of Purchaser dated May 13, 2008, a copy of which is attached hereto under **Exhibit 1**;

and

Polinvest S.r.l., a company duly organized and existing under the laws of the Republic of Italy, with registered office in Crema, at Via Riva Fredda, 3, tax code and registration number at the Companies' Register of Cremona 01444090193 (hereinafter referred to as "**Seller**"), hereby represented by Mr. Giuseppe Alfredo Poli, duly authorized pursuant to the resolution of the Shareholders' Meeting of Seller dated May 16, 2008, a copy of which is attached hereto under **Exhibit 2**.

WHEREAS

- A. Seller is the full and exclusive registered owner of 100% of the issued and outstanding stock capital of Poli S.p.A., a company duly organized and existing under the laws of the Republic of Italy, with registered office in Via Fontanella 11, 26010 Camisano (Italy), having a stock capital of Euro 6,000,000.00 fully paid-in, tax code and registration number in the Companies' Register of Cremona 00104570197 (the "Company").
- **B.** The Company and the Group Companies (as defined hereunder) are engaged in the business of developing, manufacturing, distributing and servicing rail brakes mainly in Europe (the "**Business**").
- C. The Company is the full and exclusive beneficial and registered owner of an equity interest representing 79.58% of the stock capital of the company MZT Hepos AD, a company duly organized and existing under the laws of Macedonia, with registered office in Skopje, Pero Nakov b.b., having a stock capital of Euro 4,105,483.35, registered with the Central Register of Republic of Macedonia, under no. 4137787, listed on the Stock Exchange of Skopje (hereinafter referred to as the "MZT Hepos"). The holding structure of the Company and of the Group Companies is set forth in <u>Exhibit A-1</u> attached hereto
- **D.** MZT Hepos, in turn, is the full and exclusive beneficial and registered owner of:
 - (i) an equity interest corresponding to 100% of the stock capital of the company MZT HEPOS POLSKA Sp.Zo.o, a company duly organized and existing under the laws of Poland, with registered office in Warszawa, Volodiovskiego no. 51, loc10, no. 02-724, registered with the Court Register of Warszawa under no. 0000267042 (hereinafter referred to as "**Hepos Polska**");

- (ii) an equity interest corresponding to 90% of the stock capital of the company MZT HEPOS KTB d.oo. Beograd, a company duly organized and existing under the laws of Serbia, with registered office in Belgrade, Gavrilo Princip no. 29, registered with the Corporate Court of Belgrade under no. III Fi 9125/99 (hereinafter referred to as "**Hepos KTB**"); and
- (iii) an equity interest corresponding to 100% of the stock capital of the company MZT HEPOS Inzenering, a company duly organized and existing under the laws of Macedonia, with registered office in Skopje, Pero Nakov b.b., registered with the Basic Court Skopje II under bankruptcy procedure number III. St. 722/06 (hereinafter referred to as "Hepos Inzenering");
- E. Mr. Giuseppe Alfredo Poli, Mr. Cristiano Poli, Mr. Alessandro Poli, Mr. Paolo Poli, Mr. Giancarlo Poli, Mr. Alberto Poli, Mr. Francesco Poli, Mr. Mario Poli, Mrs. Elena Poli, Mrs. Emma Severgnini, Mrs. Gianfranca Tinti, Mrs. Lucia Boffelli and Mrs. Giuseppina Enrica Virtuani are the exclusive, beneficial and, as the case may be, usufructuary owners, of the entire capital stock of Seller.
- F. Purchaser has conducted a due diligence investigation on the Company, MZT Hepos and their respective Business by means of (i) access to a data room containing the documents listed in <u>Exhibit A-2</u> ("Data Room Documents") attached hereto; (ii) management interviews; (iii) site visits and (iv) soil samplings on Camisano and Skopje sites ("Due Diligence"). The Data Room Documents include also the business documentation sent or delivered directly to Purchaser as listed in <u>Exhibit A-2</u>.
- G. The interest of the Parties to enter into this Agreement is also based on the circumstance that Mr. Giuseppe Alfredo Poli will assume a role within the Company after the implementation of the transaction contemplated herein, pursuant to the terms and conditions set forth in the Term Sheet for Mr. G. A. Poli (as defined below).
- **H.** Upon the terms and subject to the conditions set forth in this Agreement, Purchaser has agreed to purchase from Seller, and Seller has agreed to sell and transfer to Purchaser, at Closing on the Closing Date (as those terms are hereinafter defined), an equity interest equal to 100% of the corporate capital of the Company (hereinafter referred to as the "**Equity Interest**"), free and clear of any Lien.
- I. Upon the terms and subject to the conditions set forth in this Agreement, Purchaser has agreed to the sale and transfer to Seller, and Seller has agreed to purchase from Purchaser the real estate properties as listed in <u>Exhibit A-3</u> attached hereto (hereinafter referred to as the "Real Estate Property") on the Closing Date or immediately after the Closing.
- L. The Parties acknowledge that Purchaser is willing to proceed with the transaction contemplated herein for the purposes of entering into the European railway market and creating new synergies with the Company and the Group Companies.

NOW, THEREFORE, in consideration of the premises, mutual promises, covenants, representations and warranties contained in this Agreement, the Parties hereby agree as follows:

ARTICLE 1

Recitals, Exhibits and Schedules

- 1.1 The above Recitals, as well as the Exhibits and Schedules annexed to this Agreement and referred herein shall form an integral and substantial part hereof.
- 1.2 The following are the Exhibits and Schedules attached hereto and incorporated herein as an integral and substantial part of this Agreement:

Exhibit 1 - Powers of Purchaser;

Exhibit 2 – Powers of Seller;

Exhibit A-1 – Holding Structure of the Company and of Group Companies;

Exhibit A-2 – Data Room Documents;

Exhibit A-3 – Real Estate Property;

Exhibit 2.1.1 – Example of Calculation of the Closing Date Working Capital and Target Working Capital;

Exhibit 2.1.2 – Accounts Receivables;

Exhibit 2.1.3 – Financial Statements;

Exhibit 2.1.4 – Notes for the Real Estate Property;

Exhibit 2.1.5(a)—Term Sheet for Mr. G. A. Poli;

Exhibit 2.1.5(b) – Term Sheet for Poli Family's Members;

Exhibit 2.1.6 – Termination Letter;

Exhibit 3.2.1 – Modalities for the Calculation of the Net Debt;

Exhibit 3.2.2 –Escrow Agreement;

Exhibit 3.3(g) – Mandate Letter;

Exhibit 5.2(d) – Certificates;

Exhibit 5.2(e) – Letters of Resignation;

Exhibit 5.2(g) – Form of Lease;

Exhibit 6.6(i) – MZT Hepos Land;

Exhibit 6.6(ii) - MZT Hepos Buildings; and

Exhibit 8.1(g) – Tax Matters.

ARTICLE 2

Certain Definitions

- 2.1 In addition to the terms defined in other parts of this Agreement, the following terms shall have, for the purposes of this Agreement, the meanings set forth below unless otherwise expressly provided or unless the context clearly requires otherwise:
 - "Accounting Principles" shall mean for the Company the accounting principles set forth in the Italian Civil Code and those established by the Italian Accountants' Organization ("Principi Contabili predisposti dai Consigli Nazionali dei Dottori Commercialisti e dei Ragionieri"), as amended and interpreted by the Italian Accounting Organization ("Organismo Italiano di Contabilità") and for the Group Companies the accounting principles generally accepted in the respective jurisdiction of incorporation.

- "Accounts Receivables" shall mean the receivables shown in the Financial Statements and listed in Exhibit 2.1.2.
- "Affiliate" shall mean with respect to any specified Person (as hereinafter defined) at the time of determination, any other Person that directly, or indirectly through one or more intermediaries, controls, is controlled by or is under common control with, such specified Party pursuant to Article 2359, Section 1, no.1 of the ICC.
- "Business" shall have the meaning ascribed to such term in Recital (B).
- "Business Day" shall mean each calendar day (excluding Saturdays and Sundays and public holidays of each year) on which credit institutions are open in Milan, Skopje and in New York markets, for the exercise of their activity.
- "Claim Notice" shall have the meaning ascribed to such term in Section 8.3(i).
- "Closing" shall mean the sale and purchase of the Equity Interest, the payment of the Purchase Price (as hereinafter defined) and, in general, the execution and exchange of all documents and the performance and consummation of all obligations, respectively required to be executed, exchanged, performed and consummated on the Closing Date (as hereinafter defined) pursuant to this Agreement.
- "Closing Date" shall mean the date provided under Section 5.1 of this Agreement.
- "Closing Date Net Debt" shall mean the actual Net Debt of the Company on the Closing Date as calculated in accordance with the modalities and the Accounting Principles set forth in Exhibit 3.2.1. It is intended that the cash paid by Seller to the Company for the VAT relating to the Real Estate Transaction shall not be part of the Closing Date Net Debt.
- "Closing Date Working Capital" of the Company and of MZT Hepos shall mean, as of the Closing Date, the amount of the Company's and MZT Hepos' current assets excluding cash and cash equivalents (as calculated in accordance with the Accounting Principles consistently applied for the calculation of the Target Working Capital) less the Company's and MZT Hepos' current liabilities excluding financial short term liabilities (as calculated in accordance with the Accounting Principles consistently applied for the calculation of the Target Working Capital) an example of which is attached hereto under Exhibit 2.1.1. The Closing Date Working Capital excludes also Tax assets and Tax liabilities other than "ritenute d'acconto".
- "Company" shall have the meaning ascribed to such term in Recital (A).
- "Designated Purchaser" shall have the meaning ascribed under the definition of the term "Purchaser" in this Section 2.1 below.
- "Disputed Accounting Matters" shall have the meaning ascribed to such term in Section 3.3(e).

- "Environmental Laws" means any and all applicable national or local laws and regulations concerning air pollution control, water pollution control, resource conservation and recovery, solid waste disposal, toxic substance control and any other national or local laws, regulations or other requirements regulating or otherwise concerning Hazardous Materials or the environment or their negative impacts on individual health.
- "Equity Interest" shall have the meaning ascribed to such term in Recital (H).
- "Escrow Account" shall have the meaning ascribed to such term in Section 3.2.2.
- "Escrow Agent" shall have the meaning ascribed to such term in Section 3.2.2.
- "Escrow Agreement" shall have the meaning ascribed to such term in Section 3.2.2.
- "Escrow Amount" shall have the meaning ascribed to such term in Section 3.2.2.
- "Estimation of Real Estate Taxes" shall mean the estimation of the Real Estate Taxes made by Seller equal to Euro 893,058.00.
- **"Financial Statements"** shall mean the financial statements audited or unaudited as of December 31, 2007, indicated with respect to each set of financial statements in **Exhibit 2.1.3** of the Company, MZT Hepos, Hepos KTB and Hepos Polska.
- "Governmental Authority" shall mean any legislative, executive or judicial unit of any governmental entity (foreign, national, state or local) or any department, commission, board, agency, bureau, official or other regulatory, administrative or judicial authority thereof.
- "Group Companies" shall mean MZT Hepos, Hepos Polska, Hepos KTB and Hepos Inzenering.
- "Hazardous Materials" mean any hazardous, toxic or dangerous substance, pollutant, contaminant, waste or other material regulated under Environmental Laws; asbestos or asbestos containing materials; oil and petroleum products and natural gas, natural gas liquids, liquefied natural gas, and synthetic gas usable for fuel subject to Environmental Laws; chemicals subject to Environmental Laws; and industrial process and pollution control wastes.
- "Hepos Polska" shall have the meaning ascribed to such term in Recital (D).
- "Hepos KTB" shall have the meaning ascribed to such term in Recital (D).
- "Hepos Inzenering" shall have the meaning ascribed to such term in Recital (D).
- "ICC" shall mean the Italian Civil Code.
- "Independent Accountant" shall have the meaning ascribed it under Section 3.3(e).

"Law(s)" shall mean any applicable national federal, state, provincial or local law, statute, ordinance rule, regulation, code, order, judgment, injunction or decree.

"Lien" shall mean any security interest, mortgage, lien, usufruct, claim, charge, pledge, encumbrance, assessments, third party's rights, defects of title or restriction of any kind, including, without limitation, any restriction on the use, voting, transfer or receipt of income or other attributes of ownership or rights of set-off, including but not limited to rights of first refusal, or other adverse interest of any kind and nature whatsoever and other encumbrances of any kind. This definition, however, shall not include the following: (i) Liens reflected in the Financial Statements, (ii) Liens consisting of zoning or planning restrictions, easements, permits and other restrictions or limitations on the use of real property or irregularities in title thereto which do not materially detract from the value of, or materially impair the use of, such property by the Company and the Group Companies, (iii) Liens for Taxes, assessments or governmental charges or levies on property not yet due, or due but not delinquent and (iv) Liens incurred or deposits made in the ordinary course of business in connection with workers' compensation, unemployment insurance and other types of social security or to secure the performance of tenders, statutory obligations, surety, bids or performance of similar obligations.

"Losses" shall have the meaning ascribed to such term in Section 8.1(a).

"Management Agreement" shall mean the management agreement to be entered into, on or before Closing, by and between the Company, on the one side, and Mr. Giuseppe Alfredo Poli, on the other side, on the basis of the terms and conditions of the Term Sheet for Mr. G.A. Poli,

"Mandate Letter" shall have the meaning ascribed to such term in Section 3.3(g).

"Material Adverse Effect" shall mean a material adverse change in the financial conditions, results, operations, business, prospects, rights, licenses, properties or assets of the Company and the Group Companies, excluding such changes which (after taking into reasonable account any warranty, indemnity or other right or recourse contained in this Agreement with respect to the relevant event or circumstance) result from (A) any terrorist attacks, (B) a change in, or enhancement of, any provision of Law or any regulation, issuance, order or similar provisions by any competent authorities, after the date of execution of this Agreement and (C) the public announcement of the transactions contemplated in this Agreement. Any Material Adverse Effect shall be deemed as such if, individually or in the aggregate, such effect involves costs, expenses or other losses for the Company and the Group Companies of Euro 5,000,000.00 (five million) or more (excluding any costs and expenses of investigating and defending such item, such as attorneys' and accountants' fees).

" $MZT\ Hepos$ " shall have the meaning ascribed to such term in Recital (C).

"Net Debt" shall mean the aggregate amount of (i)(a) 25% of Trattamento di Fine Rapporto (also identified as the "employee leaving indemnity" in the Financial Statements) <u>plus</u> (b) any assumed financial debt <u>plus</u> (c) any capital lease obligations

calculated as net present value (*valore attuale dei canoni*) of the Company and any Group Company <u>less</u> (ii) the cash and cash equivalents on hand of the Company and any Group Company, each as of the Closing Date, as estimated by the Company and calculated in accordance with the modalities and the Accounting Principles set forth in <u>Exhibit 3.2.1</u> and provided to Purchaser 2 (two) Business Days prior to Closing.

- "Notes" shall mean the zero percent interest and unsecured notes issued by Purchaser in favor of Seller in relation to the payment of the Real Estate Property located in a) Camisano, b) Romanengo, c) Milan and d) Warsaw—substantially in the text attached hereto under Exhibit 2.1.4. The amount of the Notes shall be equal to the value attributed to the relevant Real Estate Property as communicated by Seller to Purchaser 2 (two) Business Days prior to Closing or the envisaged date of transfer, as applicable under Section 5.4 (*Sale of Real Estate Property*).
- "Notice of Disagreement" shall have the meaning ascribed to such term in Section 3.3(b).
- "Order" shall have the meaning ascribed to such term in Section 7.2.14(a).
- "Parties" shall mean Purchaser and Seller.
- "Person" shall mean any individual or natural person, foundation, company, corporation, partnership, firm, association, unincorporated organization or any other entity.
- "Present Value of Estimation of Real Estate Taxes": shall mean the amount of the Estimation of Real Estate Taxes divided equally over five years and discounted to the present value at a four percent annual rate equal to Euro 795,147.00.
- "Price Adjustment" shall have the meaning ascribed to such term in Section 3.4.1.
- "Purchase Price" shall have the meaning ascribed to such term in Section 3.2.
- "Purchaser" shall have the meaning ascribed to such term in the preamble to this Agreement. The "Designated Purchaser" shall be a 100% wholly owned Affiliate of Purchaser designated by Purchaser to purchase the Equity Interest upon a 3-Business Day prior written notice sent to Seller in accordance with article 1401 *et seq.* of the ICC. Upon such designation and relevant acceptance by the Designated Purchaser, all and any of Purchaser's rights and all and any of Purchaser's obligations deriving from this Agreement shall be respectively automatically transferred and assigned to the Designated Purchaser, effective as of the Signing Date, as if the Designated Purchaser originally executed this Agreement in place of Purchaser. The Designated Purchaser shall be appointed pursuant to the provisions of Section 10.1 (*Right to Designate*) of this Agreement.
- "Purchaser Conditions Precedent" shall have the meaning ascribed to such term in Section 4.1.
- "Real Estate Property" shall have the meaning ascribed to such term in the Recitals.

- "Real Estate Transaction" shall have the meaning ascribed to such term in Section 5.4(a).
- "Seller" shall have the meaning ascribed to such term in the preamble.
- "Seller Conditions Precedent" shall have the meaning ascribed to such term in Section 4.2.
- "Tax" shall mean all applicable taxes and duties imposed by any national or local taxing authorities, including income, property, excise, sale, use, value added and franchise taxes.
- "Target Working Capital" shall mean the amount equal to Euro 8,082,000.00 as stated and explained in Exhibit 2.1.1.
- "Term Sheet for Mr. G. A. Poli" shall mean the term sheet containing the main terms and conditions relating to the role that Mr. Giuseppe Alfredo Poli will assume within the Company after the implementation of the transaction contemplated herein. The Term Sheet for Mr. G. A. Poli is attached hereto as **Exhibit 2.1.5 (a)**.
- "Term Sheet for Poli Family Members" shall mean the term sheet containing the main terms and conditions relating to the roles that Mr. Francesco Poli, Mr. Mario Poli, Mr. Cristiano Poli, Mr. Alessandro Poli, Mr. Giancarlo Poli, Mr. Alberto Poli and Mr. Paolo Poli will assume within the Company after the implementation of the transaction contemplated herein. The Term Sheet for Poli Family Members is attached hereto as **Exhibit 2.1.5** (b).
- "Termination Letter" shall mean the letter in the form attached to this Agreement as <u>Exhibit 2.1.6</u> to be sent by Seller to the Iranian Sale Representatives to the address included in the Iranian Agreement, for the purposes of terminating the Iranian Agreement, in accordance with the terms and conditions of Section 6.9 (*Covenant relating to Iranian Agreement*).
- 2.2 All references to Articles, Sections, Recitals, Exhibits and Schedules are intended as references to the Articles, Sections, Recitals, Exhibits and Schedules relating to this Agreement. All terms in the singular shall have the same meaning as the same terms in the plural and vice versa. The definitions of the terms importing the masculine gender shall be identical to those of the same terms importing the feminine gender and vice versa. The headings in this Agreement shall be for convenience of reference only and shall not affect the construction hereof. The terms with their initials in capital letters shall have the meaning of the same terms as defined in this Agreement.

ARTICLE 3

<u>Purchase and Sale of the Equity Interest</u> <u>Purchase Price and Other Related Matters</u>

3.1 **Purchase and Sale of the Equity Interest.**

- 3.1.1 Upon the terms and subject to the conditions of this Agreement, at Closing on the Closing Date, Seller shall sell, assign, convey, transfer and deliver to Purchaser, and Purchaser shall purchase from Seller the Equity Interest, free and clear of any Lien, against the payment of the Purchase Price.
- 3.1.2 The Equity Interest shall be sold and transferred together with all rights and obligations attaching to it as at Closing (including the right to receive all dividends or distributions paid on or after the Closing Date) free and clear of any Lien.

3.2 Purchase Price and Escrow Amount.

- 3.2.1 In consideration for the sale and purchase of the Equity Interest, Purchaser shall pay to Seller on the Closing Date a purchase price (hereinafter referred to as the "**Purchase Price**") which shall be calculated as follows:
 - Euro 53,000,000.00 plus the Notes minus the Net Debt minus the Present Value of Estimation of the Real Estate Taxes.
 - The Parties agree that, on the Closing Date, Purchaser shall pay to Seller the Purchase Price less the Escrow Amount which shall be deposited by Purchaser to the Escrow Account as per Section 3.2.2 below.
- 3.2.2 The Parties agree that for the purpose of Section 8.1 (*Indemnification obligation of Seller*), 9.5 (*Environmental Liability*) and 9.4.1 (*Allocation of Tax Liabilities*) of this Agreement on the Closing Date, simultaneously with the payment by Purchaser of the Purchase Price as per Section 3.2.1 above, Purchaser shall deposit an amount equal to Euro 7,500,000.00 (hereinafter referred to as the "Escrow Amount"), to an escrow account opened in the name of both Seller and Purchaser (hereinafter referred to as the "Escrow Account") at the bank, mutually agreed upon the Parties (hereinafter referred to as the "Escrow Agent") pursuant to the terms and conditions of an escrow agreement substantially in the text attached hereto as Exhibit 3.2.2 (hereinafter referred to as the "Escrow Agreement"). The Escrow Agreement will direct the Escrow Agent to pay subject to the terms and conditions of the Escrow Agreement to Seller the amount of any funds in the Escrow Account on the first year anniversary of the Closing Date in excess of Euro 5,000,000.00 (and not subject to any indemnification claim under this Agreement) including any accrued interest on the Escrow Agent to pay subject to the terms and conditions of the Escrow Agreement to Seller the balance of any funds held in the Escrow Account (and not subject to any indemnification claim under this Agreement) including all accrued interest since the first anniversary date.
- 3.3 <u>Calculation of Closing Date Working Capital and Closing Date Net Debt</u>. The Parties agree to the following policies and procedures relating to the calculation of the

amount of the Closing Date Working Capital and the Closing Date Net Debt:

- (a) Within 30 (thirty) Business Days following the Closing Date, Purchaser shall provide Seller with a calculation of the amount of the actual Closing Date Working Capital and the Closing Date Net Debt. In quantifying the Closing Date Working Capital, the obsolete, not usable, not saleable inventory as of the Closing Date, will be the greater of the actual value or Euro 600,000.00.
- (b) Seller shall have reasonable access, upon prior notice, during normal business hours, to any books and records relating to the Company and the Group Companies and the related calculation of the Closing Date Working Capital and the Closing Date Net Debt that Purchaser may possess. Seller shall have 30 (thirty) Business Days from the date of Seller's receipt of Purchaser's Closing Date Working Capital and Closing Date Net Debt calculation to object in writing to part or all of Purchaser's calculation and amount of the Closing Date Working Capital and the Closing Date Net Debt. If Seller's elects to object, Seller shall deliver a written notice of disagreement (hereinafter referred to as the "Notice of Disagreement"), specifying the nature and reasons of such disagreement.
- (c) In the event that Seller fails to deliver a Notice of Disagreement within the 30 Business Days period, Purchaser's Closing Date Working Capital and the Closing Date Net Debt calculation shall be final and binding upon the Parties for the purposes of Section 3.3 (*Calculation of Closing Date Working Capital and Closing Date Net Debt*) and Section 3.4 (*Purchase Price Adjustment*).
- (d) If Seller submits a Notice of Disagreement, during a period of ten (10) Business Days following the delivery of such notice, Seller and Purchaser shall attempt to resolve any disagreement which they may have with respect to any matter specified in such Notice of Disagreement.
- (e) If, at the end of such period, Seller and Purchaser fail to reach an agreement in writing with respect to any and all of the matters which are the subject matter of the Notice of Disagreement then all of such matters to which agreement is not so reached and which have an accounting nature (hereinafter referred to as the "**Disputed Accounting Matters**") will be submitted by the more diligent Party to Deloitte or failing acceptance by such independent accountant to Price WaterhouseCoopers (hereinafter referred to as the "**Independent Accountant**").
- (f) If for any reason the Independent Accountant named in accordance with sub-paragraph (e) above is not able to act or informs the Parties that it is no longer willing or able to act, the Parties shall attempt to agree on a replacement of the Independent Accountant, to be selected among primary independent international accounting firms. If after 5 (five) Business Days from the date of notification of the fact that the Independent Accountant is unable or unwilling to act, the Parties have not agreed on a new Independent Accountant, then the Independent Accountant shall be selected (on the application of the more diligent Party) by the Chairman of the Court of Milan among primary independent international

accounting firms (and shall thereafter for all purposes relating to this Agreement be deemed to be the Independent Accountant).

- (g) The Independent Accountant shall: (i) act in accordance with terms and conditions of a mandate letter which shall have substantially the form attached hereto as **Exhibit 3.3(g)** (hereinafter referred to as the "**Mandate Letter**"), subject to any amendments requested by the Independent Accountant in accordance with the market practice that the Parties hereby undertake to accept; (ii) act as an arbitrator pursuant to Article 1349 (*Determinazione dell'oggetto*) and Article 1437 (*Determinazione del prezzo affidata ad un terzo*) of the ICC, to the extent strictly required to make the determinations contemplated by this Section 3.3 (*Calculation of Closing Date Working Capital and Closing Date Net Debt*); and (iii) be bound by the rules set forth in Article 1703 *et seq.* (*Mandato*) of the ICC and not act on the basis of its *arbitrium merum*.
- (h) The Independent Accountant shall formally accept in writing the Mandate Letter in order to settle and determine the Disputed Accounting Matters within the time frame set forth in such Mandate Letter and in accepting such mandate shall expressly undertake in writing to:
 - (i) consider only the Disputed Accounting Matters;
 - determine the Disputed Accounting Matters by redrafting, to the extent necessary, the calculation of the Closing Date Working Capital and the Closing Date Net Debt. The calculation of individual items for the purposes of determining the Closing Date Working Capital will be done consistently in accordance with historical practice (that is with methods and criteria used in the Financial Statements of the Company and MZT Hepos). In ensuring the consistency, the Independent Accountant will take as his/her basis for the comparison each single value included in the Financial Statements whether or not the values are calculated in accordance with the Accounting Principles and/or the principles set forth in the notes to the Financial Statements. It is intended that tax assets and tax liabilities will not constitute elements of the Closing Date Working Capital for the purpose of this provision, with the only exception of the taxes withheld (*ritenute d'acconto*) from the compensation of the personnel and other individuals, whether single or in association (*associazione di professionisti*);
 - (iii) be required to summarily justify its determinations with respect to each of the Disputed Accounting Matters; and
 - (iv) resolve all Disputed Accounting Matters within the time frame set forth in the Mandate Letter.
- (i) The Independent Accountant shall have reasonable access to the books, records, personnel and any other information of the Parties to the maximum extent

- required, in its reasonable judgment, to perform the services contemplated hereby.
- (j) Upon resolution of all Disputed Accounting Matters, the Independent Accountant shall deliver to the Parties a final calculation of the Closing Date Working Capital and the Closing Date Net Debt.
- (k) The final calculation of the Closing Date Working Capital and the Closing Date Net Debt prepared and delivered in accordance with paragraph (g) preceding shall be final and binding upon the Parties for purposes of Section 3.4 below (*Purchase Price Adjustment*) and shall not be subject to appeal. Any disputes not finally determined by the Independent Accountant shall be determined in accordance with Section 10.10 (*Arbitration*).
- (l) All fees and disbursements of the Independent Accountant due in connection with the resolution of the Disputed Accounting Matters pursuant hereto and with the provision of the services contemplated hereby shall be equally shared between the Parties.

3.4 Purchase Price Adjustment.

- 3.4.1 In the event that the Closing Date Working Capital is less than the Target Working Capital, Seller shall pay Purchaser the difference between the Closing Date Working Capital and the Target Working Capital; provided, that if the difference is less than Euro 100,000.00 (one hundred thousand), no adjustment amount will be due. In the event that the Closing Date Working Capital is more than the Target Working Capital, Purchaser shall pay Seller the amount by which the Closing Date Working Capital exceeds the Target Working Capital; provided, that if such excess amount is less than Euro 100,000.00 (one hundred thousand) no adjustment amount will be due. The Purchase Price shall be subject to a downward or, as the case may be, upward adjustment made Euro by Euro equal to the difference between the Net Debt and the Closing Date Net Debt which shall be paid by Seller or Purchaser, as applicable.
 - The amounts paid pursuant to this Section 3.4 (Purchase Price Adjustment) shall be the "Price Adjustment".
- 3.4.2 The Parties expressly agree that all payment operations carried out at Closing shall be considered neutral for the purposes of the Price Adjustment.
- Payments. Save as otherwise provided for in this Agreement, the Purchase Price, the Price Adjustment and any and all other sums payable by one Party to the other pursuant to this Agreement shall be paid (with value date as at the date on which the payment is due) by wire transfer to the bank account designated by either Party and communicated to the other Party at least 10 (ten) Business Days prior to the date on which the payment is due.

ARTICLE 4

Conditions Precedent to Closing

- 4.1 **Conditions Precedent in favor of Purchaser**. The obligations of Purchaser to proceed with the Closing and to consummate the transactions contemplated by this Agreement is subject to the satisfaction, unless waived in writing by the Purchaser, on or before the Closing, of the conditions precedent (hereinafter referred to as "**Purchaser Conditions Precedent**"):
 - (a) The representations and warranties made by Seller to Purchaser set forth in Sections 7.2.1 (*Organization, Qualification and Corporate Power*); 7.2.2 (*Power and Authority of Seller*), 7.2.4 (*Capitalization of the Company and the Group Companies*), 7.2.10 (*Assets and Inventory*), 7.2.11 (*Real Property*), 7.2.12 (*Intellectual Property*), 7.2.13 (*Contracts*) and 7.2.18 (*Licenses and Permits*), whose breach may prevent the Company or any of the Group Companies from conducting the Business as conducted as of the Signing Date, shall be true and correct in all material respects on and as of the Closing Date with the same effect as if such representations and warranties had been made on and as the Closing Date; provided, however, that this condition to Closing will be deemed to have been satisfied so long as any and all breaches of or inaccuracies in the representations and warranties above as of the Closing Date could not, individually or in the aggregate, be reasonably expected to result in an adverse effect on the Company and the Group Companies involving losses of Euro 5,000,000.00 or more.
 - (b) No suit or proceeding not disclosed in any Schedule or Exhibit attached to this Agreement by any governmental or regulatory commission (other than the Antitrust Authority) or by any shareholder or creditor of Seller shall be pending on the Closing Date that materially challenges the transactions to be consummated pursuant to this Agreement.
 - (c) Prior to or at the date of the Closing Date, the purchase and sale of the Equity Interest pursuant to this Agreement shall have been approved, cleared or granted an exemption by the Antitrust Authorities in Macedonia and Serbia, to the extent that any such approval, clearance or exemption is required, or all relevant waiting periods having expired without the purchase and sale of the Equity Interest contemplated herein being prohibited by the Macedonian and Serbian Antitrust Authorities.
 - (d) No Material Adverse Effect shall have occurred.
 - (e) The Management Agreement shall have been executed on or prior to Closing.
 - (f) Prior to or at the date of the Closing Date, the Termination Letter shall have been sent, if so requested by Purchaser, in accordance with the terms and conditions of Section 6.9 (*Covenant relating to Iranian Agreement*).
- 4.2 **Conditions Precedent in favor of Seller.** The obligations of Seller to proceed with the Closing and to consummate the transactions contemplated by this Agreement is subject to

the satisfaction, unless waived in writing by Seller, on or before the Closing, of the conditions precedent (hereinafter referred to as "Seller Conditions Precedent"):

- (a) The representations and warranties made by Purchaser to Seller pursuant to Sections 7.1.1 (*Organization, Qualification and Corporate Power*); and 7.1.4 (*Financial Capability*) of this Agreement shall be true and correct in all material respects on and as of the Closing Date with the same effect as if such representations and warranties had been made on and as the Closing Date.
- (b) No suit or proceeding by any governmental or regulatory commission (other than the Antitrust Authority) by any shareholder or creditor of the Purchaser shall be pending on the Closing Date that materially challenges the transactions to be consummated pursuant this Agreement.
- (c) The Management Agreement shall have been executed on or prior to Closing.
- 4.3 **Effects**. If any of Purchaser Conditions Precedent or Seller Conditions Precedent are not satisfied or waived in accordance with the above provisions on or prior to July 31, 2008 or October 31, 2008, as the case may be, *then* (unless otherwise agreed in writing by the Parties) Section 10.8 (*Termination*) shall govern.

ARTICLE 5

Closing

- 5.1 Closing and Closing Date. The Closing shall take place at the offices of Gianni, Origoni, Grippo & Partners in Milan, Italy, at Piazza Belgioioso 2, or at such other place mutually agreed upon in writing on: (i) June 30, 2008 if all the conditions precedent to Closing under Sections 4.1 and 4.2 are satisfied or waived prior to June 30, 2008 (other than those which are to be satisfied or waived at Closing), or (ii) the last day of the month in which all the conditions precedent to Closing under Sections 4.1 and 4.2 are satisfied or waived, if all such conditions precedent are satisfied or waived after June 30, 2008 (other than those which are to be satisfied or waived at Closing), or, in any event, within the deadline set forth in Section 10.8 (*Termination*) (hereinafter referred to as the "Closing Date").
- 5.2 Closing Date Actions and Deliveries by Seller. In addition to any other action to be taken and to any other instrument to be executed or delivered pursuant to this Agreement on or prior to the Closing Date, on the Closing Date, simultaneously with the receipt by Seller of satisfactory evidence that irrevocable instructions for the payment of the Purchase Price to the bank account to be specified by Seller in accordance with Section 3.5 (*Payments*) have been given by Purchaser and confirmation by the relevant bank that such irrevocable instructions of Purchaser have been duly processed, Seller shall take the following actions and shall deliver to Purchaser the following documentation:
 - (a) endorsement in favor of Purchaser of the share certificates representing the Equity Interest by way of *girata* certified by a Notary public, bank officer or other person under and for purposes of article 2355, 3rd paragraph, ICC, or execute such

instruments or documents and do such other things as are necessary, under Italian law, to transfer to Purchaser good and marketable title to the Equity Interest free and clear of any Liens and to cause Purchaser to be registered in the Shareholders' Book of the Company as owner of the Equity Interest;

- (b) deliver to Purchaser copies of the resolutions of the relevant corporate body (certified to be in conformity by a representative deemed to be fit to do so) of Seller, authorizing Seller to execute and perform its obligations under this Agreement, as well as all other documents under this Agreement requiring signature by Seller;
- (c) deliver to Purchaser a copy of the current by-laws of the Company and each Group Company certified by an appropriate officer of the relevant company and the corporate books of the Company and;
- (d) deliver the certificates individually and specifically listed in **Exhibit 5.2(d)** dated as of a date as close to the Closing Date as is reasonably practicable;
- (e) cause all members of the board of directors and use its reasonable efforts to cause all statutory auditors (both permanent and alternate) of the Company and of the Group Companies to submit and deliver their resignation letters (in the form attached hereto as **Exhibit 5.2(e)** and copies of which shall be delivered to Purchaser by Seller), effective as of the Closing Date, whereby they resign or otherwise cease from office with effect from Closing and confirm that they have no claims for compensation for termination, loss of office, unpaid remuneration or otherwise;
- (f) cause the shareholders' meeting of the Company and of the Group Companies to resolve upon (1) the appointment of new members of the board of directors and new statutory auditors (where required) of the Company and of the Group Companies *in lieu* of those who shall have resigned from their office pursuant to point (e) above, all in accordance with the Purchaser's instructions to be notified in writing to Seller at least three (3) Business Days prior to the Closing Date; and (2) if requested by Purchaser, the approval of new by-laws, a draft of which shall be delivered by Purchaser to Seller at least three (3) Business Days prior to the Closing Date;
- deliver to Purchaser an executed copy of the lease agreement between Seller, as lessor, and the Company, as lessee having as its subject matter part of the Real Estate Property located in Camisano (Italy) namely the industrial shed and offices (including other appurtenances) all identified at the Borough New Land Register of Camisano, Sheet 8, Parcel 186, Sub. 504 (hereinafter referred to as the "Leased Property") substantially in the form attached as Exhibit 5.2(g), which shall automatically become effective immediately following completion of the Real Estate Transaction in relation to the site of Camisano. To this respect, Purchaser acknowledges that before the date of this Agreement the Company started the actions including the execution of a contract (appalto) with

Precompressi S.r.l. (hereinafter referred to as the "Contract") – necessary to carry out certain building works in the industrial land (hereinafter referred to as the "Construction") – identified at the Borough New Land Register of Camisano, Sheet 8, Parcel 236 and 238 – neighboring to the Leased Property and that the Construction shall be continued by Seller. Therefore, Purchaser undertakes to cause the Company to perform any action requested or appropriate to transfer (*voltura*) to Seller the existing building permits (as well as any other possible administrative documents) currently registered in the name of Company and to assign (*cessione*) the Contract – prior to the obtainment of the consent of Precompressi s.r.l. – and other possible agreements related thereto to the extent permitted by law and pursuant to the terms and conditions provided therein; and

- (h) execute the Escrow Agreement.
- 5.3 Closing Date Actions and Deliveries by Purchaser. In addition to any other action to be taken and to any other instrument to be executed or delivered pursuant to this Agreement on or prior to the Closing Date, on the Closing Date, Purchaser shall, concurrently with the actions taken by Seller pursuant to Section 5.2 (Closing Date Actions and Deliveries by Seller) above, take the following actions and shall deliver to Seller the following documentation:
 - (a) pay, or cause to be paid, to Seller the Purchase Price in accordance with Sections 3.2.1 (*Purchase Price –Escrow*) and 3.5 (*Payments*) above to the bank account to be specified by Seller pursuant to Section 3.5 (*Payments*) in accordance with the applicable provisions of this Agreement pursuant to the modalities set forth in Section 3.5 (*Payments*);
 - (b) pay (or cause to be paid) immediately at Closing or within 5 (five) Business Days from Closing by the appropriate local entities or persons and in the appropriate manner any stamp, transfer or similar taxes or charges however levied by any Governmental Authority on the transfer of the Equity Interest, to be incurred in relation to such transfer and the notarial fees relating thereto;
 - (c) deliver to Seller certified copies of the resolutions of the Board of Directors of the Purchaser, authorizing Purchaser to execute and perform its obligations under this Agreement, as well as all other documents under this Agreement requiring signature by Purchaser; and
 - (d) execute the Escrow Agreement and deposit the Escrow Amount to the Escrow Account.

5.4 **Sale of Real Estate Property**.

(a) At Closing, or as soon as practicable after Closing, as the case may be, Purchaser shall cause the Company to sell to Seller, and Seller shall purchase, all right, title and interest in the Real Estate Property free and clear of any Lien in exchange for the cancellation of the Notes. In connection with the transfer, Seller and the Company shall execute such instruments or documents and take such actions as are necessary, under any applicable

- Law, to transfer to Seller good and marketable title, to the Real Estate Property (hereinafter referred to as the "Real Estate Transaction").
- (b) At Closing, Purchaser shall cause the Company to sell to Seller, and Seller shall purchase, all right, title and interest in the Real Estate Property located in Camisano as indicated in <u>Exhibit A-3</u> attached hereto, free and clear of any Lien in exchange for the cancellation of the relevant Notes.
- (c) As soon as practicable after Closing, Purchaser shall cause the Company to sell to Seller, and Seller shall purchase, all right, title and interest in the Real Estate Properties located in Warsaw, Milan and Romanengo as indicated in Exhibit A-3 attached hereto, free and clear of any Lien in exchange for the cancellation of the relevant Notes.
- (d) Seller undertakes to provide Purchaser with all the information and, upon Purchaser's request, the copy of the documentation concerning the transfer of the Real Estate Property and the cancellation of the Notes at least 7 (seven) Business Days prior to Closing, with respect to the Real Estate Property located in Camisano; and at least 3 (three) Business Days prior to Closing or the envisaged date of the transfer, with respect to Real Estate Properties located in Warsaw, Milan and Romanengo-.
- (e) All costs and expenses relating to the Real Estate Transaction shall be borne by Seller including transfer fees, registration tax, cadastral tax, mortgage tax and VAT.
- 5.5 <u>Additional Information and Documents</u>. Seller and Purchaser shall, on request, on and after the Closing Date, cooperate with one another by providing any additional information, executing and delivering any additional documents and/or instruments and doing any and all such other things as may be reasonably required by the Parties or their counsels to consummate or otherwise implement the transactions contemplated by this Agreement.
- Single Transaction and Timing of Real Estate Transaction. All of the actions, executions, productions, remittances and deliveries under the above Sections 5.1 (Closing and Closing Date), 5.2 (Closing Date Actions and Deliveries by Seller) and 5.3 (Closing Date Actions and Deliveries by Purchaser) provided to be taken and made at Closing on the Closing Date shall be regarded for the purpose of the Closing as a single transaction, so that, at the option of the Party having an interest in carrying out the relevant action, no action, execution, production, remittance and delivery constituting the Closing shall be effective unless all other actions, executions, productions, remittances and deliveries constituting the Closing shall be fully and regularly performed in accordance with the provisions of this Agreement. For the purposes, the Parties hereby already express their agreement in relation to the granting, if necessary, of joint irrevocable mandates in their mutual interest to a third party of their mutual satisfaction, in order to perform, in their name and behalf, the actions, executions, productions, remittances and deliveries provided under the above Sections 5.1 (Closing and Closing Date), 5.2 (Closing Date Actions and Deliveries by Seller) and 5.3 (Closing Date Actions and Deliveries by Purchaser). The actions contemplated in Section 5.4 (Sale of the Real Estate Property) shall occur on the Closing Date or, only with respect to the Real Estate Properties located

in Warsaw, Milan and Romanengo, as soon as possible after Closing.

ARTICLE 6

Covenants

- 6.1 Access. Seller agrees to cause the Company and each of the Group Companies to allow Purchaser and its directors, officers, employees, representatives, counsels and/or advisors (hereinafter referred to as the "Purchaser Representatives") reasonable access during normal business hours on Business Days throughout the period prior to the Closing, as Purchaser and Purchaser Representatives may reasonably request, to the premises, facilities, plants, machinery, equipment, inventories, directors, officers, personnel (limited to the executives and middle managers), books, records, data and other documents and information of the Company and of the Group Companies. It is expressly agreed upon by the Parties that the Purchaser shall have no access to and shall not contact in any manner whatsoever customers of the Company and of the Group Companies unless such contacts have been previously expressly approved in writing by Seller.
- 6.2 <u>Conduct of the Company and the Group Companies</u>. From the Signing Date to the Closing Date, except as may be expressly agreed by Purchaser and except as is expressly permitted under this Agreement, Seller shall and/or, as applicable, shall cause the Company and the Group Companies to:
 - (a) conduct their business in the ordinary and usual course, consistent with past practices;
 - (b) maintain the current insurance polices on all their properties, real and personal, owned or leased; and use, operate, maintain and repair all such properties in the ordinary course of business;
 - (c) refrain from doing any act or omitting to do any act, or permitting any act or omission, that shall cause a breach of any material agreement, contract, commitment of the Company or of the Group Companies, having a value exceeding Euro 100,000.00 (one hundred thousand) in the aggregate;
 - (d) not solicit, encourage, any inquiries or proposals (other than the transaction contemplated hereby) for the acquisition of (i) part or all of the capital stock or other equity interest in the Company or in any of the Group Companies owned by Seller, the Company or any other Group Company as of the Signing Date; or (ii) part or all of the assets of the Company or of the Group Companies except for sales of inventory in the ordinary course of business consistent with past practice;
 - (e) maintain and keep the facilities, machinery and equipment of the Company and of the Group Companies that are necessary to the operation of the business being conducted by the Company and the Group Companies in normal operating condition and repair, except for ordinary wear and tear;

- (f) not permit or, not consent or promote, the merger or consolidation of the Company and any Group Company with any other Person;
- (g) consult with Purchaser regarding all significant transactions relating to the business of the Company or of the Group Companies, for an amount in excess of Euro 500,000.00 (five hundred thousand);
- (h) not allow the Company and any of the Group Companies, or not consent or promote, to:
 - (i) amend any of its by-laws;
 - (ii) incur any indebtedness in excess of Euro 150,000.00 (one hundred fifty thousand) except for any trade payables in the ordinary course of business;
 - (iii) incur or extend the timing of its payment of any payables outside of the ordinary course of business;
 - (iv) change any payment terms of, or change any payment practices with respect to, any payable or receivable; or
 - (v) enter into any contract or agreement involving any capital expenditure or undertaking of obligations in excess of Euro 25,000.00 (twenty-five thousand) except for purchase orders and supply orders in the ordinary course of business; however, in the event such purchase or supply orders shall have a duration exceeding one (1) year they shall have to be preauthorized by Purchaser within five Business Days, which shall not be unreasonably withheld;
- (i) not grant any options, warrants or other rights to purchase or obtain any of the capital stock or equity interest of the Company or of the Group Companies, or issue, sell or otherwise dispose of the capital stock of the Company or of the Group Companies;
- (j) not declare, set aside or pay any dividend or distribution with respect to the capital stock of the Company or of the Group Companies, or, except as expressly directed by Purchaser, redeem, repurchase or otherwise acquire shares/quotas of the capital stock of the Company or of the Group Companies;
- (k) not grant any Lien upon any of the assets of the Company or of the Group Companies, except in the ordinary course of Business (including the fulfillment of obligations already assumed);
- (l) not make any loan in excess of Euro 10,000.00 (ten thousand);
- (m) not make any investment in, or acquire any non marketable securities or assets of any other Person in an amount in excess of Euro 50,000.00 (fifty thousand) other than purchase of inventory and supplies in the ordinary course of business of the

Company and of the Group Companies;

- (n) not increase the rate of compensation or materially increase the benefits payable or to become payable to any of their directors, officers, employees or self-employed persons (other than raises made in the ordinary course of business or unless such raises are due under applicable labor Laws and regulations or Collective Bargaining Agreements) or make any material change any of the terms of employment of any of their directors, officers or employees; and
- (o) promptly notify Purchaser in writing:
 - (i) of the occurrence of any event that has resulted in, or could be reasonably expected to result in a Material Adverse Effect;
 - (ii) if the Company or any of Group Companies is clearly advised that any of the customers of the Company or of the Group Companies intends to cease doing business or reduce the amount of goods or services purchased from the Company or any of the Group Companies or intends to decrease the prices applied because of the transactions contemplated by this Agreement or the announcement thereof: or
 - (iii) if the Company or any of the Group Companies is clearly advised that any supplier to the Company or the Group Companies intends to cease doing business with the Company or any of the Group Companies or materially reduce the amount of goods sold to the Company or the Group Companies or intends to increase the prices applied because of the transactions contemplated by this Agreement or the announcement thereof.
- (p) not initiate any new litigation or claim against any customer or pursue any existing litigation or claim against Trenitalia, without Purchaser's prior consent.

Solely for the purposes of this Section 6.2 (*Conduct of the Company and the Group Companies*), Purchaser shall appoint Mr. Chuck Kovac as Purchaser's representative who shall be promptly available to consider and either approve or deny any actions requested by Sellers under provided for by this Section 6.2 (hereinafter, the "Interim Period Purchaser Representative"). Seller shall send a written notice to the Interim Purchaser Representative (copy to Mr. R. Mark Cox) requesting approval for Seller, the Company or any Group Company to take any action during the period from the date of this Agreement until the Closing that are not permitted under Section 6.2. For purposes of this Section 6.2, written notice may be provided by email. In the event that the Interim Period Purchaser Representative fails to respond to the request within 5 (five) Business Days from the date of the receipt of the relevant written notification by Seller, then such request shall be deemed to have been irrevocably and fully accepted and approved by Purchaser.

6.3 Consents.

(a) Purchaser and Seller covenant and agree to pursue diligently all consents or

approvals required from any Person or Governmental Authority as may be necessary to consummate the transactions contemplated by this Agreement.

- (b) As soon as is reasonably practicable after the date of this Agreement (but in any event no later than 5 (five) Business Days, with reference to both Serbia and Macedonia—after the date of this Agreement, Purchaser shall duly and properly file with the relevant Antitrust Authority the notification required by the relevant Serbian and Macedonian Antitrust Law respectively, the Serbian Law on Protection of Competition, adopted on 14 September 2005 (Official Gazette of the Republic of Serbia no. 79/2005), and the Macedonian Law on Protection of Competition, adopted on 11 January 2005 (Official Gazette of the Republic of Macedonia no. 04/05). Purchaser shall provide copies of all such documents to Seller prior to filing and, if requested, shall accept all reasonable additions, deletions or changes suggested in connection therewith. Seller shall furnish to Purchaser all information required for any such filing. Seller and Purchaser will consult and cooperate with one another, and consider in good faith the views of one another, in connection with any analyses, appearances, presentations, memoranda, briefs, arguments, opinions and proposals made or submitted by or on behalf of any party hereto in connection with the filings above or any additional inquiries by the relevant Antitrust Authority. Each of Seller and Purchaser agree to consider in good faith any actions necessary to resolve objections, if any, as may be asserted by any Antitrust Authority with respect to the transactions contemplated by this Agreement.
- (c) Purchaser agrees to indemnify Seller for any costs, expenses, losses or liabilities that the latter should sustain to the extent the Antitrust Authority has imposed remedies to be undertaken by Seller in connection with the purchase and sale of the Equity Interest.
- (d) In case the Antitrust Authority has imposed remedies to be undertaken by Purchaser in connection with the purchase and sale of the Equity Interest, Purchaser shall be entitled to either (i) not close this Agreement, or (ii) close this Agreement pursuant to the same terms and conditions hereof (including the Purchase Price).
- (e) With particular reference to Serbia, should the relevant clearance be not granted by the relevant Antitrust Authority, the Parties shall agree to negotiate in good faith any necessary amendment to the terms and conditions of this Agreement in order to consummate the transactions contemplated hereby, provided that such amendments shall not imply a material change of arrangements herein contained or any material expenditure or an increase of the risk of either Party.
- (f) Seller is entitled to cause the sale/disposal by MZT Hepos of its participation in Hepos KTB before the Closing at its own costs and expenses.
- (g) Should the sale/disposal under letter (e) above occur, upon completion of the relevant sale/disposal process (which shall be proved by MZT Hepos with

evidence satisfactory to the Purchaser), Purchaser shall take all necessary steps and actions (including the filing of the relevant documentation) in order to withdraw Purchaser's request for clearance filed with the Serbian Antitrust Authority. Upon receipt by Purchaser of a written confirmation by the Serbian Antitrust Authority approving the Purchaser's request of withdrawal, then, the condition precedent under Section 4.1.(c) (*Serbian Antitrust Clearance*) shall be deemed fully and irrevocably satisfied.

6.4 **Representations and Warranties**. Each of Seller and Purchaser shall not take any action that would result in any of its respective representations and warranties not being true in all material respects on the Closing Date. Seller and Purchaser shall use best efforts to cause the fulfillment at the earliest practicable date of all the conditions to its respective obligations to consummate the sale and purchase under this Agreement. Seller and Purchaser shall promptly notify to each other in writing should they become aware of the existence or occurrence of any condition which would make any representation or warranty in this Agreement untrue, any breach of any covenant of Seller or Purchaser in this Agreement or which might prevent or delay the Closing.

6.5 **Duty of Confidentiality**.

- 6.5.1 Until Closing, each Party undertakes: (i) to keep confidential all information delivered or disclosed (whether orally or in writing) to the other Party in connection herewith with a level of care and attention which is not less than that used to protect confidential information and documentation relating to such Party's own companies; and (ii) not to disclose any of the foregoing to any Person without the previous written consent of the other Party, except and only to the extent required by applicable Law or by any Governmental Authority (including without limitation any regulatory or stock exchange authority having jurisdiction on any of the Parties) or such information has become public through no fault of the Party who receives it.
- 6.5.2 For a period of 1 (one) year from the date hereof, Seller shall keep strictly confidential and shall not use, directly or indirectly, any of the information concerning Purchaser that it has received or receives in its capacity as Party to this Agreement and shall cause its officers, employees and consultants to whom such information has been disclosed for the purposes of this Agreement to comply with such commitment. Seller shall exercise all necessary precautions to safeguard the confidentiality and secrecy of such information and to prevent the disclosure thereof.
- 6.5.3 For a period of 1 (one) year from the date hereof, Purchaser shall keep secret and confidential any confidential information concerning the personal business and affairs relating to the Poli family that Purchaser has received or receives as a result of this Agreement or of any investigations made in connection herewith or otherwise and shall cause its officers, employees and consultants to whom such information has been disclosed for the purposes of this Agreement to comply with

- such commitment. Purchaser shall exercise all necessary precautions to safeguard the confidentiality and secrecy of such information and to prevent the disclosure thereof.
- 6.5.4 The Parties shall not be deemed in breach of Section 6.4 (*Representations and Warranties*) by virtue of any disclosure made pursuant to the provisions or requirements of any applicable Law or any rule issued by any regulatory or stock exchange authority having jurisdiction on any of the Parties.
- 6.6 Covenants relating to MZT Hepos. All fees, expenses and charges related to the privatization process of the land where MZT Hepos is currently conducting its business (as defined under Exhibit 6.6(i) (MZT Hepos Land)) in accordance with the Law on Privatization and Lease of State-owned Construction Land (Official Gazette of the Republic of Macedonia no. 4/05 and no. 13/07) and the decree on the amount of fees for privatization of construction land from July 1, 2007 and those deriving from the registration in the "possession list" and the "property list" at the Cadastral register as user of the land as well as those connected with the procedure for determining the full ownership of the buildings and recording of the buildings in the property list in the Cadastral register (as identified under Exhibit 6.6(ii) (MZT Hepos Buildings)) (including the payment of any taxes, including the communal tax) in accordance with Law on Construction Land ("Official Gazette of the Republic of Macedonia no. 53/2001) and the Rulebook for calculating the costs for organizing construction land from October 2, 2001 shall be entirely borne by Purchaser. Similarly, all costs and expenses connected to the law suit initiated by MZT Hepos in connection with the determination of its right to use the above land and the denationalization procedure shall be entirely borne by Purchaser.

6.7 Covenants relating to Mandatory Tender Offer.

- (a) The Parties hereby acknowledge and agree that—in the event that any transactions contemplated by this Agreement trigger, in any manner whatsoever, a mandatory tender offer on the outstanding shares of MZT Hepos, regardless of the effective modalities pursuant to which said tender offer should be promoted under applicable law(s), any debt, cost, loss, amount incurred and/or suffered by the Company and any Group Company arising from and/or in relation to the purchase of any shares of MZT Hepos pursuant to the above tender offer (i) shall not be taken into account and shall be disregarded for the purpose of calculation of the Closing Date Net Debt and the Net Debt and that, as a consequence (ii) shall not have any impact on the calculation of the Purchase Price.
- (b) In the event under letter (a) above, the Parties further acknowledge and agree that any action to be implemented, any authorization, consent or approval to be required, any document to be filed by the Seller, the Company and/or any Group Company in connection with the above tender offer, regardless of the effective modalities pursuant to which said tender offer should be promoted under applicable law(s), shall be the sole responsibility of Purchaser and any fees, charges and expenses in connection with the above shall be exclusively borne by

Purchaser.

6.8 Covenant relating to Camisano and Skopje Site. [INTENTIONALLY DELETED]

6.9 Covenant relating to Iranian Agreement.

- (a) Immediately after the execution of this Agreement, Seller shall cause MZT Hepos to use its best efforts to terminate before the Closing Date the sale representative agreement entered into with Mr. Nader Ghalandari Safavi, Mr. Shahram Sanayei Lotfabadi and M. Abdulrahman Samadi, as sales representatives (hereinafter, the "Iranian Sale Representatives"), on May 10, 2006, relating to the promotion and sale of the MZT Hepos products in Iran (hereinafter referred to as the "Iranian Agreement"). In this respect, after the execution of this Agreement, and prior to the Closing Date, Seller agrees to pay, or cause the Company or any Group Company, to pay any amounts payable under the terms of the Iranian Agreement related to services or products provided under Iranian Agreement and undertakes to complete any outstanding orders and to refuse any new orders from the Iranian Sale Representatives.
- (b) For the purposes of achieving the above termination, Seller shall consult with and inform, and shall cause MZT Hepos to consult with and inform, Purchaser on the actions to be taken in order to terminate the Iranian Agreement before the Closing Date.
- (c) In the event that Seller does not receive written evidence of termination of the Iranian Agreement by June 15, 2008, <u>then</u> Purchaser shall be entitled to ask Seller to send the Termination Letter prior to June 25, 2008. Should no request in writing of delivering the Termination Letter be received by Seller within the above June 25, 2008 deadline, the condition precedent under Section 4.1(f) of this Agreement (*Delivery of the Termination Letter*) shall be deemed fully and irrevocably satisfied.
- (d) For the sake of clarity, it is understood that the condition precedent under Section 4.1(f) of this Agreement (*Delivery of the Termination Letter*) shall be deemed fully and irrevocably satisfied upon the delivery by Seller of the Termination Letter, regardless of the circumstance that the Iranian Agreement is not terminated before Closing.

The Parties expressly agree that, except for the amount to be paid by Seller, the Company or any Group Company under Section 6.9(a) above, all costs, fees and disbursements due in connection with the termination of the Iranian Agreement (including, without limitation, any possible amount to be paid to the Iranian Sale Representatives and those connected to any possible claim or proceeding by the Iranian Sale Representatives in connection with the termination of the Iranian Agreement) shall be entirely borne by Purchaser.

6.10 Term Sheet for Mr. G.A. Poli and Term Sheet for Poli Family Members

- (a) After the execution of this Agreement, Purchaser shall cause the Company, before or at Closing, to execute the Management Agreement with Mr. Giuseppe Alfredo Poli on the basis of the terms and conditions of the Term Sheet for Mr. G.A. Poli. It is understood among the Parties that the Management Agreement shall not contain any further provisions other than the ones provided by the Term Sheet for Mr. G.A. Poli, with the exception of those provisions mandatory required by the National Collective Bargaining Agreement for Dirigenti in the industrial sector (hereinafter referred to as the "NCBA Dirigenti") or by the applicable laws. In particular, the Management Agreement shall not contain provisions which (i) worsen the provisions of the Term Sheet for Mr. G.A. Poli or (ii) are in breach of the NCBA Dirigenti or (iii) increase the obligations to be borne by Mr. Giuseppe Alfredo Poli or (iv) in any way limit his rights and entitlements as provided in the Term Sheet for Mr. G.A. Poli.
- (b) After the execution of this Agreement, Purchaser shall cause the Company, before or at Closing, to grant Mr. Francesco Poli, Mr. Mario Poli, Mr. Cristiano Poli, Mr. Alessandro Poli, Mr. Giancarlo Poli, Mr. Alberto Poli and Mr. Paolo Poli the qualification and job title (or function), compensation and benefits, described, for each of them, under the Term Sheet for Poli Family Members. Such terms and conditions can not be changed.

ARTICLE 7

Representations and Warranties

Representations and Warranties by Purchaser. Purchaser hereby makes the following representations and warranties to Seller.

7.1

- 7.1.1 Organization, Qualification and Corporate Power. Purchaser is a corporation duly organized, validly existing and in good standing under the Laws of the State of Delaware, with corporate powers and authority to own properties and to conduct its business as currently conducted.
- 7.1.2 <u>Power and Authority of Purchaser</u>. Purchaser has full power and authority to enter into this Agreement and any other agreement or instrument contemplated herein and shall have taken all requisite corporate action to enter into this Agreement and to consummate the transactions contemplated herein.
- 7.1.3 <u>Binding Effect</u>. This Agreement has been duly executed by Purchaser and constitutes the legal, valid and binding obligation of Purchaser, enforceable in accordance with its terms. Purchaser is not required to give any notice to, make any filing with, or obtain any authorization, consent or approval of any government or governmental agency in order to consummate the transactions contemplated by this Agreement, except as contemplated by this Agreement.
- 7.1.4 <u>Financial capability</u>. Purchaser (i) has adequate and sufficient funds available to pay the Purchase Price and any expenses incurred by Purchaser in connection with the transaction contemplated by this Agreement; (ii) has the resources and

- capabilities (financial or otherwise) to perform its obligations hereunder; and (iii) has not incurred in any obligation, restriction or liability of any kind, absolute or contingent, which would materially impair or adversely affect such resources and capabilities.
- 7.1.5 No conflict. Neither the execution and delivery by Purchaser of this Agreement nor the consummation of the transaction contemplated herein will conflict with, or result in a breach of, any provision of the articles of incorporation or by-laws of Purchaser, or any material agreement or instrument by which is bound, nor will violate any order, judgment, award, writ, injunction, decree, statute, rule or regulation applicable to Purchaser.
- 7.1.6 Accuracy as of the Closing Date. The representations and warranties of Purchaser contained herein shall be true and correct also as of, and as though made on, the Closing Date.
- 7.2 Representations and Warranties by Seller. Seller hereby makes the following representations and warranties on which Purchaser has relied in executing this Agreement and entering into the transactions contemplated herein. The Parties agree that the representation and warranties are independent contractual warranties and are not subject to the provisions of article 1495 ICC. Seller shall not be liable under the representations and warranties set forth in this Section 7.2 in respect of any act, contract, action, event or circumstance disclosed in, or referred to, or resulting from any Schedule or Exhibit attached to this Agreement irrespective of whether the relevant Schedule or Exhibit is expressly referred to by the relevant representation and warranty.
 - 7.2.1 <u>Organization, Qualification and Corporate Power</u>. Except for Hepos Inzenering, each of Seller, the Company and the Group Companies is a corporation duly organized, validly existing and in good standing under the Laws of the jurisdiction in which it was incorporated. Each of Seller, the Company and the Group Companies is duly authorized to conduct business as presently conducted and is in good standing under the Laws of each jurisdiction in which the nature of its business or the ownership or leasing of its properties requires such qualification.
 - 7.2.2 <u>Power and Authority of Seller</u>. Seller has full power and authority to enter into this Agreement and any other agreement or instrument contemplated herein and has taken all requisite corporate action to enter into this Agreement and to consummate the transaction contemplated herein.
 - 7.2.3 <u>Binding Effect; No Conflict</u>. This Agreement has been duly executed by Seller and constitutes the legal, valid and binding obligation of Seller, enforceable in accordance with its terms. None of Seller, the Company or any Group Company is required to give any notice to, make any filing with, or obtain any authorization, consent or approval of any government or Governmental Authority in order to consummate the transactions contemplated by this Agreement, and no consent or approval of any Person is required for Seller to enter into this Agreement or any

other agreement or instrument contemplated in this Agreement or to consummate the transactions contemplated herein or therein.

- 7.2.4 <u>Capitalization of the Company and of the Group Companies.</u>
 - (a) The holding structure of the Company and of the Group Companies is set forth on **Exhibit A-1**; none of the Company nor any of the Group Companies own any capital stock or other equity interest in, or have the right or obligation to purchase or pay up any capital stock or other equity interest in, any other Person.
 - (b) Except for the pre-emption right provided for by the Law on Business Companies (Published in the Official Gazette of the Republic of Serbia no. 125/04) with respect to Hepos KTB, the stock capital of the Company and each Group Company is validly issued, fully paid and is not subject to any rights of first refusal, pre-emption right, option right or prohibitions or limitations upon transfer and there are no voting trust agreements or other contracts, agreements or arrangements restricting voting, management or dividends rights or transferability.
 - (c) Seller owns and has, and on Closing Date shall own and shall have, good and valid title to the Equity Interest free and clear of any Lien. On the Closing Date, Purchaser shall own the Equity Interest free and clear of any Lien.
- 7.2.5 <u>Financial Statements</u>. Except as set forth in <u>Schedule 7.2.5</u> hereof, the Financial Statements (i) are true and correct in all material respects, (ii) have been prepared in accordance with the relevant books and records, (iii) have been prepared in accordance with the Accounting Principles consistently applied, and (iv) fairly present the financial position of the Company and of the Group Companies as of the dates indicated and the results of operations, shareholders' equity and cash flow of the Company and of the Group Companies for the periods indicated. Except as set forth in <u>Schedule 7.2.5</u> hereof, the books and records of the Company and of the Group Companies accurately and fairly reflect, in all material respects, all the transactions to be recorded therein in accordance with the applicable Laws and the Accounting Principles.
- 7.2.6 <u>Absence of Undisclosed Liabilities</u>. Except as disclosed in <u>Schedule 7.2.6</u>, as of December 31, 2007, the Company and the Group Companies shall have no debts or liabilities which have been incurred on or prior to December 31, 2007, and which should have been reflected in the Financial Statements referred to such date according to the Accounting Principles and were not so reflected, other than any liabilities incurred after the date of the Financial Statements as a result of the ordinary course of business.
- 7.2.7 Receivables. Except as disclosed in Schedule 7.2.7, no fact or circumstance exists which would result in the in-collectability of any Accounts Receivables. With

reference to the Accounts Receivables, there is no contest, claim, defence or right of setoff pending, other than returns in the ordinary course of business, under any contract with any account debtor of an account receivable relating to the amount or validity of such account receivable.

7.2.8 No Change. Since December 31, 2007, the Business has been operated in the ordinary course of business consistent with past practice and no change has occurred with respect to the Business that has resulted in or could reasonably be expected to result in a Material Adverse Effect.

7.2.9 <u>Taxes</u>.

- (a) The Company and each of the Group Companies has, in a timely manner (within any applicable extension periods, including retroactive extensions such as "ravvedimento operoso" and, where applicable, tax amnesties), paid all Taxes, due pursuant to tax Law applicable to the Company and the Group Companies (hereinafter referred to as the "Tax Laws"). The Company and each of the Group Companies have made all withholdings required by the Tax Laws, and has paid over to the appropriate Governmental Authorities the amounts due within the time periods provided for by the Tax Laws.
- (b) The Company and each of the Group Companies have filed, in a timely manner (within any applicable extension periods, including retroactive extensions such as "ravvedimento operoso" and, where applicable tax amnesties), all annual corporate tax returns (hereinafter referred to as the "Tax Returns") prescribed by applicable Tax Laws. The Tax Returns have been prepared in accordance with the Tax Laws, are correct and complete in all material respects, and accurately state taxes due by the Company and the Group Companies. All taxes shown as due on the Tax Returns have been paid.
- (c) Except as disclosed under Schedule 7.2.9 (c), there is no pending or threatened litigation, nor have assessments been issued, nor are there any inspections or other actions being conducted by tax authorities or by any other governmental authority regarding asserted violations of Tax Laws by the Company and the Group Companies, which, in the event of an adverse result, could lead to the application of sanctions for the Company and the Group Companies.

7.2.10 <u>Assets and Inventory.</u>

(a) The Company and the Group Companies own or have the right to lease, use or license all of the assets, tangible and intangible, and properties listed in Schedule 7.2.10(a)(i), and such assets and properties are sufficient to operate the Business in the manner presently operated by the Company. Except as disclosed under Schedule 7.2.10(a)(ii), the assets are sufficient

for the purposes for which such assets are currently used or are held for use, and are in reasonably good repair and operating condition (subject to normal wear and tear). Except for the leases and licensed assets listed on Schedule 7.2.10(a)(iii), the Company and the Group Companies own good and transferable title to all of the assets included in Schedule 7.2.10(a)(i) free and clear of any Liens.

(b) The inventories of the Company and of the Group Companies as reflected in the Financial Statements consist of a quality and quantity usable and, with respect to finished goods, saleable in the ordinary course of business, and do not include obsolete, non saleable or non usable items for an amount exceeding Euro 600,000.00. For purposes of the foregoing and of the Target Working Capital, "obsolete" items shall mean any items that have not moved within the past 10 (ten) years.

7.2.11 Real Property.

- (a) The Group Companies have possession over the real estate property (hereinafter referred to as the "Possessed Real Property") listed in <u>Schedule 7.2.11(a)</u> attached hereto.
- (b) The Company and the Group Companies have entered into the lease agreements (hereinafter referred to as the "**Leases**") listed in **Schedule 7.2.11(b)** attached hereto for certain real estate property (hereinafter referred to as the "**Leased Real Property**").
- (c) Except as disclosed under Schedule 7.2.11(c), neither the Company nor any of the Group Companies is party to, or under any agreement or commitment to become party to, any lease with respect to real property used in the relevant businesses, other than the Leases. With respect to each Lease, with respect to the Company and the Group Companies: (i) all rents have been paid in accordance with the terms and conditions of each Lease; (ii) neither the Company, nor the Group Companies nor the lessors under any of the Lease has brought any actions vis-à-vis the relevant contractual relationships referred to the leased properties.
- (d) Except as disclosed under <u>Schedule 7.2.11(d)</u>, the Possessed Real Property is free and clear of any Lien and there are no outstanding options or rights of first refusal to purchase the Possessed Real Property.
- (e) The Company is the owner of the Real Estate Property listed in **Exhibit A-3** free and clear of any Lien.

7.2.12 <u>Intellectual Property</u>

(a) The Company and the Group Companies own and are licensed to use only the patents, patent applications, trademarks, trademark applications, trade names, know-how, trade secrets, technologies and all computer software

(including data bases and related documentation) (the "Intellectual Property Rights") listed in Schedule 7.2.12(a) attached hereto. Schedule 7.2.12(a)(i) contains a list of the Intellectual Property Rights owned by third parties and used by the Company and the Group Companies in the conduct of the Business and a list of the applicable license or contract providing for such license. The Intellectual Rights listed on Schedule 7.2.12(a) are in full force and effect and shall constitute all of the Intellectual Property Rights sufficient to conduct the Business as it is currently conducted.

- (b) All owned Intellectual Property is free and clear from any and all Liens.
- (c) Neither the Company nor the Group Companies have received written notices alleging that neither the conduct of the Company nor of any Group Company has infringed or currently is infringing any intellectual property rights of third parties.
- (d) Except as disclosed under <u>Schedule 7.2.12(d)</u>, to the knowledge of Seller, no third party has infringed or misappropriated (or currently is infringing or misappropriated) any Intellectual Property Rights owned by the Company or any Group Company which relates to the Business, and, to Seller's knowledge, there are no events which could give rise to such infringements.
- (e) No third parties have any rights to receive from the Company or the Group Companies any royalty, nor are there any other royalty free licenses to which the Company or the Group Companies is a party with respect to the patents (except for Software) or trademarks.
- (f) Neither the Company nor the Group Companies are party to any agreement with any third party limiting or restricting the Company's and the Group Companies' right to use the patents or trademarks.
- (g) The Company and the Group Companies own or are licensed to use, on a time basis, the computer software (including data bases) (hereinafter, collectively, referred to as the "Software") which is used to the conduct of the Business and except as disclosed under Schedule 7.2.12(g) and also except for non customized software readily available from multiple sources, neither the Company nor the Group Companies have to pay any consideration for the use of the Software. To Seller's knowledge, no Person is interfering with or infringing, and no entity has misappropriated any of the Software. None of the Software infringes upon, is in misappropriation of or otherwise conflicts with any patent or copyright of any third parties.

7.2.13 Contracts.

(a) All material contracts relating to the Company and the Group Companies

- (including any contract with suppliers and customers) are listed on (i) **Schedule 7.2.13(a)(i)** with respect to real property leases, (ii) **Schedule 7.2.13(a)(ii)** with respect to third party licenses and (iii) **Schedule 7.2.13(a)(iii)** with respect to any other material contract of the Company and the Group Companies having a global annual consideration exceeding Euro 100,000.00 (one hundred thousand) (collectively, the "**Material Contracts**").
- (b) Each Material Contract is in full force and effect and constitutes the legal, valid and binding obligation of the Company and any Group Companies which is party thereto and the other parties thereto, enforceable in accordance with its terms. Neither the Company nor any Group Company has received, as of the date hereof, any notice that it is in default under any Material Contract.
- (c) Except as disclosed under <u>Schedule 7.2.13(c)</u>, no Material Contract may be terminated as a result of the execution of this Agreement.
- (d) Neither the Company nor any Group Company is in default under any Material Contract and no event has occurred which, with the passage of time, would constitute such a default; and, to Seller's knowledge, no other Person is in default under any Material Contract.

7.2.14 <u>Litigation</u>.

- (a) Except as disclosed under <u>Schedule 7.2.14(a)</u> neither the Company nor any Group Company is subject to any injunction, judgment, order, decree or ruling issued by any court, arbitrators or other Governmental Authority ("**Order**").
- (b) Except as disclosed under <u>Schedule 7.2.14(b)</u>, there is no pending or, to Seller's knowledge threatened, action, arbitration, governmental investigation, litigation or suit (whether civil, criminal, administrative, judicial or investigative) commenced, brought, conducted or heard by or before any Governmental Authority or arbitrator (hereinafter referred to as the "**Proceeding**") by or against the Company or any Group Company that relates to or may affect the Business or any Group Company. No event has occurred or circumstance exists that could clearly give rise to or serve as a basis for the commencement of any such Proceeding.
- 7.2.15 Product Liability. In the last (3) three years, no claims or demands in writing have been made against the Company or any of the Group Companies alleging injury to individuals or property as a result of the faults or defects of any product manufactured, sold, owned or possessed by the Company or any Group Company. No event has occurred or circumstance exists that could clearly give rise to or serve as a basis for the commencement of any action, suit, proceeding, hearing, charge, complaint against the Company or any Group Company or otherwise

relating to the Business giving rise to any liability arising out of any injury to individuals or property as a result of the faults or defects of any product manufactured, sold, owned or possessed by the Company or any Group Company. In the last (3) three years, the Company and the Group Companies have not received any order from any Governmental Authority to recall any products manufactured and delivered by the Company or any Group Company, and there is no pending investigation by any Governmental Authority relating to any of the products manufactured by the Company or any Group Company.

- 7.2.16 <u>Insurance</u>. <u>Schedule 7.2.16</u> hereof lists all insurance policies (including policies providing casualty, liability and other forms of insurance) which relate to the Company and any Group Company. The policies of fire, casualty, liability and other forms of insurance in connection with the conduct of the Business are in full force and effect in accordance with their respective terms.
- 7.2.17 <u>Compliance with Laws</u>. The Business has been and is currently being conducted, and the Company and the Group Companies have been and currently are, in full compliance with applicable (i) contract tender Laws and (ii) anti-mafia Laws.
- 7.2.18 <u>Licenses and Permits</u>. The Company and the Group Companies have in full force and effect all licenses, permits and other governmental authorizations required for, and that are material to, the operation of the Business. All of such licenses and permits are in full force and effect and shall remain in full force and effect immediately after giving effect to the transactions contemplated in this Agreement.
- 7.2.19 Environmental Matters
 - (a) Except as indicated in <u>Schedule 7.2.19(a)</u> hereof, the Company and each of the Group Companies have, in all material respects, complied with applicable Environmental Laws. Neither the Company nor any Group Companies has received written notice from any Governmental Authority charged with the enforcement of Environmental Laws of a violation of any requirements of any Environmental Laws.
- (a-bis) Seller hereby recognizes and acknowledges that Purchaser carried out a environmental audit, assessment and investigation on the Camisano site (Italy) ("Acquisition Due Diligence Environmental Health and Safety Assessment of Poli SpA, Italy, 12th May 2008, Final, Issue no. 3, 49313957/LORP0002" and "Acquisition Due Diligence, Phase II, Environmental Due Diligence of Poli SpA, Italy, 12th May 2008, Final, Issue no. 2, 49313957/LORP0004") and in the Skopje site (Macedonia) ("Acquisition Due Diligence Environmental Health and Safety Assessment of MZT Hepos, Skopje, Macedonia, 12th May 2008, Final, Issue no. 3, 49313957/LORP0001" and "Phase II, Environmental Site Assessment MZT Hepos, Skopje, Macedonia, 9 th May 2008, Final, Issue no. 2, 49313957/LORP0003") (jointly and collectively referred to as the "URS Report"), whose findings have been disclosed to the Seller.

- (i) The findings of the URS Report (hereinafter referred to as the "**Findings**"), may not be completely conclusive for the Purchaser and exhaustive for the Company and / or any other Group Company.
- (ii) Purchaser acknowledges Seller's assertion that the disclosures listed in <u>Schedule 7.2.19(a)</u> and the Findings do not constitute an admission of fault and/or liability by the Company and/or any other Group Company vis-à-vis the environmental situation at the Camisano (Italy) and Skopje (Macedonia) sites. However, in making this acknowledgement, Purchaser does not waive its rights under Section 8.1 (*Indemnification Obligation of Seller*) and Section 9.5 (*Environmental Liability*).
- (iii) Without prejudice to points (i) and (ii) above, the Findings and the disclosures contained in <u>Schedule 7.2.19(a)</u> shall constitute the baseline for the purposes of Section 9.5 below (the "Status Quo Ante").
- (b) There has been no release of any Hazardous Materials, pollutants or contaminants in, on or affecting any properties owned, leased or operated by the Company and the Group Companies other than in accordance with applicable Environmental Laws.
- (c) None of the Company or the Group Companies has received any written notice that any property now owned, operated or leased by the Company or the Group Companies is listed on any list of sites requiring investigation or cleanup; and no lien has been filed against either the personal or real property of the Company or any of the Group Companies under any Environmental Law, regulation promulgated thereunder or order issued with respect thereto.
- (d) Neither the Company nor the Group Companies has produced, distributed or sold any asbestos containing materials nor has it assumed, voluntarily or otherwise, any liability for asbestos containing materials from any predecessor company or other third party.

7.2.20 <u>Employees—Other Self-Employed Persons</u>.

- (a) All labor relationships with the Company and the Group Companies are in compliance with the applicable Law and Collective Bargaining Agreements (including Company Agreements). All obligations deriving from the applicable Collective Bargaining Agreements in force have been duly fulfilled. There are no economic or legal conditions which are not expressed in this Agreement or in contracts or in Collective Bargaining Agreements ruling such labor relationships (in particular, but with no limitation, severance arrangements and fringe benefits).
- (b) Except as disclosed under Schedule <u>7.2.20(b)</u>, the Company and the Group Companies have made all filings and taken all actions required to be made or taken in respect of their respective employees under labor, safety at work and social security Laws.

- (c) All social security and other compulsory contributions due with respect to the employees and self-employed persons have been regularly paid by the Company and the Group Companies.
- (d) The severance indemnity and other indemnities due to the employees by the Company and the Group Companies have been regularly set aside and are properly entered in the respective financial statements.
- (e) No bonus or severance shall become due and payable under any existing agreement between the Company and the Group Companies and any of their employees and self-employed persons as a result of this Agreement or the consummation of the transaction contemplated herein.
- 7.2.21 <u>Labor Relations</u>. No allegation, charge or complaint of age, disability, sex or race discrimination or similar charge has been made or threatened against the Company or the Group Companies. With respect to the Company and the Group Companies, there is no reasonable basis upon which any such allegation, charge or complaint could be made.
- 7.2.22 <u>Employee Benefit Plans</u>. Except as disclosed under <u>Schedule 7.2.22</u>, as of the date hereof and at Closing on Closing Date, no employee benefit plans other than statutory employee benefit plans have been, and are, maintained or contributed to by the Company or the Group Companies.
- 7.2.23 <u>Customers.</u> Except as indicated in <u>Schedule 7.2.23</u> hereof, the Company and the Group Companies do not have any contract or agreement in effect (directly or through third party agents or distributors), nor accounts receivables, which relate to sales or other contracts with Iran, Sudan, North Korea, Cuba or Siria.
- 7.2.24 <u>Powers of Attorney</u>. Except as set forth in <u>Schedule 7.2.24</u>, there are no outstanding powers of attorney executed on behalf of the Company or the Group Companies authorizing any Person to assume obligation for the Company or any of the Group Companies having a value in excess of Euro 10,000.00 (ten thousand).
- 7.2.25 <u>Letters of Intent</u>. During 1 (one) year preceding the execution of this Agreement, neither the Company nor the Group Companies have executed a written letter of intent or written agreement in principle regarding the sale of the assets or of the Business of the Company or of the Group Companies outside the ordinary course of business.
- 7.2.26 Product Warranty. Each product manufactured, sold or delivered by, and/or in connection with the operation of the Business has been manufactured, sold or delivered in conformity, in all material respect, with the relevant applicable contracts on the basis of which such products have been manufactured, sold or delivered. Except as set forth on Schedule 7.2.26, in the last three (3) years, neither the Company nor any of the Group Companies have received claims or demands in writing concerning any request for replacement or repair of the above

products for an amount in excess of Euro 70,000.00 for each year.

- 7.2.27 Banks. Schedule 7.2.27 hereof contains a true, correct and complete schedule setting forth the names and addresses of all banks, trust companies, savings and loan associations and other financial institutions and post offices at which the Company and the Group Companies maintain any account. Schedule 7.2.27 includes all the account numbers and the names of every persons authorized to draw on such accounts, redeem such certificates or have access to such boxes.
- 7.2.28 Accuracy as of the Closing Date. The representations and warranties of Seller contained herein shall be true, accurate and correct in all material respects also as of, and as though made on, the Closing Date, except for those representations and warranties that are given as of a certain date, which shall be true and correct as of such date, and except as affected by actions or transactions contemplated herein or otherwise approved in writing by Purchaser or taken or made in the ordinary course of the Company's and Group Companies' business. For sake of clarity, upon sale/disposal of Hepos KTB in accordance with Section 6.3 letters (f) and (g), the representations and warranties related to Hepos KTB shall be no longer applicable and therefore Purchaser shall not be entitled to seek any indemnification for any breach of the above representations and warranties, in accordance with Section 8.1 of this Agreement (*Indemnification obligation of Seller*).

7.2.29 <u>Qualification on Knowledge of Seller.</u>

- (a) Except as expressly provided therein, none of the representations and warranties of Seller set forth in Sections 7.2.1 (*Organizations*, *Qualifications and Corporate Power*), 7.2.2 (*Power and Authority of Seller*), 7.2.3 (*Binding Effect; No Conflict*); 7.2.4 (*Capitalization of the Company and of the Group Companies*); 7.2.5 (*Financial Statements*); 7.2.6 (*Absence of Undisclosed Liabilities*); 7.2.9 (*Taxes*); 7.2.10(a) (*Assets*); 7.2.11 (*Real Property*); 7.2.12 (*Intellectual Property*) excluding letter (d) (*Infringement by third party of the Intellectual Property*) and (g) (*Software*); 7.2.13 (*Contracts*); 7.2.14 (*Litigation*), 7.2.15 (*Product Liability*), 7.2.17 (*Compliance with Laws*), 7.2.19 (*Environmental Matters*), 7.2.20 (*Employees Other Self-Employed Persons*), excluding letter (b); 7.2.21 (*Labor Relations*); 7.2.22 (*Employee Benefit Plans*); 7.2.23 (*Customers*); 7.2.24 (*Powers of Attorney*); 7.2.25 (*Letters of Intent*); 7.2.26 (*Product Warranty*), 7.2.27 (*Banks*) and 7.2.28 (*Accuracy as of the Closing Date*) are limited or otherwise qualified by the knowledge of Seller.
- (b) The representations and warranties of Seller with respect to the Group Companies set forth in Sections 7.2.7 (*Receivables*), 7.2.10(b) (*Assets and Inventory*), 7.2.12(d) (*Infringement by third party of the Intellectual Property*) and (g) (*Software*), 7.2.17 (*Compliance with Laws*) but only with respect to contract tender Laws, 7.2.18 (*Licenses and Permits*) and

- 7.2.20(b) (*Employees*) are expressly made solely to the knowledge of the Seller.
- (c) All other representations and warranties (other than those referenced in (a) and (b) above) set forth in this Section 7.2 (*Representations and Warranties by Seller*) with reference to the Company and the Group Companies are expressly made solely to the knowledge of the Seller.
- (d) The Parties further acknowledge and agree, where any representation or warranty contained in this Agreement is qualified by reference to "Seller's knowledge" and/or to the fact that Seller "is not aware" this shall mean the actual knowledge and/or non-knowledge, as the case may be—after reasonable due inquiry—as of the date hereof—of Mr. Giuseppe Alfredo Poli.
- No other Representations and Warranties. The representations and warranties of Seller contained in this Section 7.2 (*Representations and Warranties by Seller*) are *in lieu* of all other representations and warranties however provided under applicable Law and constitute all of the representations and warranties made by Seller in connection with the purchase and sale of the Equity Interest and any other transactions contemplated by this Agreement. In particular, except for the representations and warranties contained in this Section 7.2 (*Representations and Warranties by Seller*), Seller does not make any other representations or warranties and gives no warranty, express or implied, to Purchaser with respect to financial projections, budgets or management analysis and to the future prospects and profitability and financial performance of the Company or of the Group Companies, or with respect to any forecasts, projections or business plans prepared by or on behalf of Seller and delivered to Purchaser in connection with the negotiation and execution of this Agreement. Seller hereby disclaims any other representations or warranties, whether made by Seller, the Company, any Group Company or any of its or their respective officers, directors, employees, agents, advisors or representatives, with respect to the execution and delivery of this Agreement or the transactions contemplated hereby, notwithstanding the delivery or disclosure to Purchaser or its representatives of any documentation or other information with respect to any one or more of the foregoing.

ARTICLE 8

Indemnification

- 8.1 <u>Indemnification Obligation of Seller</u>. Subject to the consummation of the Closing, and in accordance with the provisions of this Article 8, Seller hereby agrees to indemnify and hold Purchaser harmless against and from:
 - (a) any and all costs, expenses (including, without limitation, reasonable professional fees and costs of investigation, judgement and litigation) losses, damages and liabilities (hereinafter referred to as the "Losses") incurred by Purchaser which

- would have not been so incurred or suffered had the representations and warranties of Seller herein contained under Section 7.2 (*Representations and Warranties by Seller*) been true and correct;
- (b) any and all Losses incurred by the Company, which would have not been so incurred or suffered had the representations and warranties of Seller herein contained under Section 7.2 (*Representations and Warranties by Seller*) been true and correct; to the extent such Losses are not indemnifiable under point (a) above;
- (c) any and all Losses incurred by any of the Group Companies , which would have not been so incurred or suffered had the representations and warranties of Seller herein contained under Section 7.2 (*Representations and Warranties by Seller*) been true and correct; to the extent such Losses are not indemnifiable under points (a) and/or (b) above;
- (d) any Losses incurred by Purchaser arising out or as a result of any breach by Seller of its obligations and covenants provided by this Agreement;
- (e) any and all Losses incurred by Purchaser, the Company or any Group Company arising out of or relating to the application for obtaining an IPPC "A" Permit, as further described in <u>Schedule 7.2.19(a)</u>;
- (f) any and all Losses incurred by Purchaser, the Company or any Group Company arising out of or relating to Hepos Inzenering, including without limitation the bankruptcy proceeding of Hepos Inzenering;
- (g) any and all Losses incurred by Purchaser, the Company or any Group Company up to Euro 741,000.00 arising out of or relating to the Tax matters identified on **Exhibit 8.1(g)**; and
- (h) any and all Losses incurred by Purchaser or the Company for any and all Taxes in excess of the Estimation of Real Estate Taxes paid by Purchaser or the Company with respect to the Real Estate Transaction.
- (i) in the event that Seller causes the sale/disposal of Hepos KTB prior to Closing, any and all Losses incurred by Purchaser, the Company or any Group Company arising out of or relating to the process of sale/disposal of Hepos KTB.
- (j) any and all Losses incurred by Purchaser, the Company or any Group Company relating to any Real Estate Property not sold to Seller at Closing.

Purchaser shall not be entitled to indemnification for any Losses resulting from any breach or failure by Seller in the event that Seller can prove—in accordance with rules set forth in the Italian Civil Procedure Code—that (i) the Purchaser or any of URS Corporation Ltd, Gianni, Origoni, Grippo & Partners, KPMG LLP and Mercer LLC had actual knowledge prior to the Closing of facts which constitute a breach by the Company, the Group Companies or Seller of a representation and warranty given in this Agreement, including by reason of the facts that such circumstances had been disclosed

during the Due Diligence, and (ii) has an understanding prior to the Closing that such facts constitute a breach of a representation or warranty by the Company, the Group Companies or Seller under this Agreement; provided that no such disclosure by Purchaser's advisors will be deemed to waive any privilege between Purchaser and the advisor.

8.2 <u>Limitations of Seller's Liability</u>. Seller's liability under Sections 8.1 (*Indemnification Obligation of Seller*) shall be subject to the quantitative and time limitations set forth below:

8.2.1 <u>Quantitative Limitations</u>.

- (a) Seller shall not be liable to Purchaser under Section 8.1 (*Indemnification Obligation of Seller*) if (i) the amount due in connection with any single occurrence giving rise to liability pursuant thereto (after having taken into account any and all applicable exclusions provided for by Section 8.2 (*Limitations of Seller's Liability*)) does not exceed Euro 25,000.00; or (ii) until the aggregate of all amounts that would otherwise be due thereto exceeds Euro 350,000.00 provided that if such limit is exceeded, Seller's liability shall be limited to the excess, and provided that all amounts in respect of which Seller's liability is excluded pursuant to point (i) above shall be disregarded for the purposes of the threshold provided in this point (ii).
- (b) It is agreed that Seller shall be liable for Losses under Section 8.1 (*Indemnification Obligation of Seller*) for all representations and warranties up to a maximum amount of Euro 7,775,000.00.
- 8.2.2 <u>Time Limitations</u>. In no event shall Seller be liable to Purchaser under Section 8.1 (*Indemnification Obligation of Seller*) in respect of:
 - (i) any actual or alleged breach of the representations and warranties contained in this Agreement which is notified to Seller later than 24 (twenty-four) months after the Closing Date;
 - (ii) any actual or alleged breach of the representations and warranties contained in Section 7.2.19 (*Environmental Matters*) and Section 7.2.20 (*Employees-Other Self Employed Persons*) (except for clause 7.2.20(c), which relates to social security matters and which is addressed in sub-part (iii) below) which is notified to Seller later than 36 (thirty-six) months after the Closing Date; and
 - (iii) any actual or alleged breach of the representations and warranties under Section 7.2.9 (*Taxes*) or Section 7.2.20(c) (*Social Security Contributions*) which is notified to Seller 60 (sixty) days after the expiry of the relevant statute of limitation.
- 8.2.3 <u>Exception to Indemnity Limitations</u>.

- (a) Any and all limitations set forth in Sections 8.2.1 (*Quantitative Limitations*) and 8.2.2 (*Time Limitations*) shall not apply with reference to Sections 8.1(d) (*Breach of Obligations and Covenants*), 8.1(f) (*Hepos Inzenering Bankruptcy Proceedings*), 8.1(i) (*Hepos KTB indemnification*) 8.1(j) (*Real Estate Properties not sold at Closing*) and in the case of breach by Seller of the representations and warranties set forth in Sections 7.2.1 (*Organizations, Qualifications and Corporate Power*), 7.2.2 (*Power and Authority of Seller*), 7.2.3 (*Binding Effect; No Conflicts*) and 7.2.4 (*Capitalization of the Company and the Group Companies*).
- (b) The limitations provided for under Sections 8.2.1 (a) (*floor and threshold*) and 8.2.1(b) (*cap*) shall not apply in respect of any Losses related to the Tax matters identified on **Exhibit 8.1(g)** in which case: (A); no floor and no threshold shall apply; and (B) a separate cap on Seller's liability up to an amount of Euro 741,000.00 (seven hundred thousand forty one) shall apply and against such a cap solely Losses related to the Identified Tax Liabilities shall be counted.
- (c) Any and all limitations set forth in Sections 8.2.1 (*Quantitative Limitations*) shall not apply with reference to Section 8.1(h) (*Real Estate Transaction*).
- 8.2.4 Other exclusions and limitations. Notwithstanding any provision of this Agreement or of any applicable Law to the contrary:
 - (a) Seller shall not be liable to Purchaser under Section 8.1 (*Indemnification Obligation of Seller*) above:
 - (i) in the event and to the extent that any Loss to be indemnified by Seller under Section 8.1 (*Indemnification Obligation of Seller*) is covered by any insurance policy of Purchaser, the Company or of the Group Companies, whether or not the relevant insurer has effectively paid the amount due under said policies; provided, that Seller shall be liable for the payment of any deductible (*franchiqia*) to any third party insurer with respect to such coverage;
 - (ii) in the event and to the extent that any Loss to be indemnified by Seller under Section 8.1 (*Indemnification Obligation of Seller*) is paid to Purchaser, the Company or the Group Companies (as the case may be) by any third party, but only to the extent that the relevant right can not be assigned to the Seller;
 - (iii) for any indirect or consequential Losses, even if arising out of any breach of the representations and warranties of Seller provided for in Section 7.2 (*Representations and Warranties by Seller*) (For avoidance of doubt, indirect or consequential Losses paid to a third party will be deemed to be direct Losses under this Agreement and not included in this

- exception under 8.2.4(a)(ii) only to the extent that such Losses constitute an object of indemnification pursuant to Section 8.1 of this Agreement);
- (iv) for any contingent or potential Losses of Purchaser, the Company and/or any of the Group Companies unless and until such Losses has become actual and final and has been paid for by the Company or the Group Companies or has become the subject matter of a final and incontestable obligation to pay of the Company or of the Group Companies (or any of them).
- (b) In the event and to the extent that any Loss to be indemnified by Seller under Section 8.1 (*Indemnification Obligation of Seller*) is deductible by the Purchaser, the Company or by any of the Group Companies for income tax purposes in any given fiscal year, then, in such event and to such extent, the amount of any Loss to be indemnified by Seller under Section 8.1 (*Indemnification Obligation of Seller*) will be reduced by an amount equal to the tax benefit obtained by the Purchaser, the Company or any of the Group Companies by virtue of such deduction (even if such tax benefit applies in subsequent fiscal years).
- (c) The amount of all indemnities payable by Seller to Purchaser pursuant to Section 8.1 (*Indemnification Obligation of Seller*) shall be further reduced by any specific reserve relating to the event giving rise to indemnification recorded in the Financial Statements.
- (d) In the case of any tax assessment that only results in the shifting of the tax burden from one fiscal year to another, any indemnification relating thereto shall be limited to the actual net and final cash cost thereof incurred by the Company or the relevant Group Company.
- (e) The Parties further agree that:
 - (i) any event that may give rise to a request by Purchaser for indemnification for breach of the representations and warranties of Seller under Section 7.2 (*Representations and Warranties by Seller*) may be the subject of the request for indemnification only upon the specific representation or warranty which is more strictly connected to such event;
 - (ii) in no way Purchaser may recover for the same event more than once even if it refers to the breach of more than one representation and warranty contained in Section 7.2 (*Representations and Warranties by Seller*).
- (f) If Seller has made any payment to Purchaser according to this Article 8 (*Indemnification*) and subsequently Purchaser, the Company or any of the Group Companies either (i) recovers or (ii) has the right to recover from

- any third party any amount for the same facts or circumstances forming the basis of Seller's indemnification, then Purchaser will (in case under point (i) above) pay or (in case under point (ii) above)—to the maximum extent possible—assign, or procure the assignment of, that right to Seller or, if such assignment cannot be carried out, the Purchaser at direction of Seller—will pursue the said recovery and pay to Seller any money or property recovered (net of Taxes).
- (g) Any indemnification due by Seller to Purchaser, the Company or any Group Company shall be computed without regard to any multiple, price-earnings or equivalent ratio implicit in the negotiation of the Purchase Price. Any indemnification due by the Seller shall not include any decrease in value of the Equity Interest and/or the shares of the Group Companies.
- (h) In no event shall Seller be required to indemnify the Purchaser in respect of any Loss to the extent it has already been taken into account in the determination of the Purchase Price, the Closing Date Net Debt and/or the Closing Date Working Capital.
- 8.3 <u>Indemnification Procedure</u>. If any event occurs which could give rise to Seller's liability under Section 8.1 (*Indemnification Obligation of Seller*) above, the following provisions shall apply:
 - (i) Purchaser shall give written notice to Seller of such event (hereinafter referred to as the "Claim Notice") under penalty of forfeiture within 35 (thirty-five) Business Days from which Purchaser has had actual knowledge or notice that Purchaser has a claim under this Article 8 (*Indemnification*). The Claim Notice will provide reasonable details of the claim and, if possible, the amounts claimed by Purchaser. Seller agrees that the Purchaser will be deemed to have automatically satisfied the notice requirement under this Section 8.3(i) (*Claim Notice*) in the event that Mr. Giuseppe Alfredo Poli or Mr. Vlado Atanasovski (for so long as he reports directly to Mr. Giuseppe Alfredo Poli) has actual knowledge or notice of the right to make a claim.
 - (ii) Seller shall be entitled to challenge in writing the Claim Notice notified by Purchaser according to the preceding paragraph 8.3(i) (*Claim Notice*), within 15 (fifteen) Business Days of the day of receipt of the Claim Notice referred to in Paragraph 8.3(i) (*Claim Notice*) above.
 - (iii) If the Claim Notice sent by Purchaser according to Paragraph 8.3(i) (*Claim Notice*) is challenged by Seller according to Paragraph 8.3(ii) (*Challenge to Claim Notice*), then during a period of 20 (twenty) Business Days following the giving of the notice by Seller under Paragraph 8.3(ii) (*Challenge to Claim Notice*) preceding, Seller and Purchaser shall attempt to resolve any differences which they may have with respect to any matters constituting the subject matter of such Claim Notice. If, at the end of such period, Seller and Purchaser fail to reach an

- agreement in writing with respect to all such matters, then all matters as to which an agreement is not so reached may be submitted to arbitration pursuant to Section 10.10 (*Arbitration*) at any time following the expiration of the 20 Business Day period referred to in this Paragraph 8.3(iii).
- (iv) if, conversely, the Claim Notice sent by Purchaser according to Paragraph 8.3(i) (*Claim Notice*) is not challenged by Seller according to Paragraph 8.3(ii) (*Challenge to Claim Notice*), the claims raised by Purchaser shall be considered accepted by Seller.
- 8.4 **Handling of Third Party Claims**. If any event occurs which could give rise to Seller's liability under paragraph 8.1 (*Indemnification Obligation of Seller*) and which consists of any written claim, action, suit or proceeding by any third party (including any Governmental Authority) (hereinafter referred to as a "**Third Party Claim**"), the following provisions shall apply:
 - (a) Purchaser shall give the Claim Notice to Seller under penalty of forfeiture within 35 (thirty-five) Business Days after Purchaser has received written notice of such Third Party Claim. The Claim Notice will provide reasonable details of the claim and, if possible, the amounts claimed by Purchaser. Seller agrees that the Purchaser will be deemed to have automatically satisfied the notice requirement under this Section 8.4(a) (*Third Party Claim Notice*) in the event that Mr. Giuseppe Alfredo Poli or Mr. Vlado Atanasovski (for so long as he reports directly to Mr. Giuseppe Alfredo Poli) has actual knowledge or notice of the right to make a claim.
 - (b) Seller shall properly and diligently defend, and shall cause the Company or the affected Group Company to properly and diligently defend, any Third Party Claim with counsel of its own choosing. To this respect, Purchaser shall: (i) appoint and cause the Company and/or the affected Group Company to appoint the counsel(s) designated by Seller and (ii) cooperate and cause the Company and the Group Companies to cooperate with Seller in the preparation for and the prosecution of the defence of the Third Party Claim (including, without limitation, making available evidence within the control of the Company and the Group Companies and persons needed as witnesses who are employed by the Company and the Group Companies, in each case as needed for such defence).
 - (c) Purchaser shall have the right to join, at its own cost and expenses, by counsel of its own choosing, reasonably satisfactory to Seller, in the defense of such Third Party Claim constituting the subject matter of a notice to Seller of the kind referred to under paragraph (a) preceding.
 - (d) In the event that Seller does not undertake the defence of the Third Party Claim within a reasonable time after Purchaser has given a notice of the kind referred to under paragraph (a) above then, at the expense of Seller and after giving notice to Seller of such action, Purchaser shall properly and diligently defend (and shall cause the Company or affected Group Company to properly and diligently

- defend) any Third Party Claim and shall refrain from taking any action which could prejudice the defense of Seller's interest hereunder. In any case, Seller shall remain entitled to join, at its own cost and expenses, by counsel of its own choosing, in the defense of such Third Party Claim.
- (e) Purchaser shall not and shall cause the Company and the Group Companies not to make or accept any settlement of any Third Party Claim, nor shall make acquiescence thereto or, as the case may be, to any demand, assessment, judgment or order constituting the subject matter of a notice of the kind referred to under paragraph (a) above, or, as the case may be, having resulted from any Third Party Claim, without the prior written consent of Seller which consent shall not be unreasonably withheld.
- 8.5 **Exclusive Remedy**. Except for any rights and remedies relating to fraud or willful misconduct, the rights and remedies provided in this Article 8 (*Indemnification*), Sections 9.4. (*Tax Matters*) and 9.5 (*Environmental Liability*) shall be exclusive and *in lieu* of any other right, action, defence, claim or remedy of Purchaser, provided by Law or otherwise, however arising in connection with, or by virtue of, any breach of the representations and warranties, undertakings and covenants of Seller contained in this Agreement.
- 8.6 <u>Indemnification Obligation of Purchaser</u>. Purchaser shall indemnify and hold Seller harmless from, against and with respect to any and all damages suffered by the latter arising out, or as a result, of any inaccuracy or breach of any representation or warranty made by Purchaser in this Agreement or of any breach of any undertaking or covenant of the Purchaser under this Agreement.

ARTICLE 9

Additional Covenants

- 9.1 Non-Competition. Without prejudice to any of the provisions of this Agreement, for a period of 3 (three) years, save for point (b) below with respect of which the period shall be of 1 (one) year, from the Closing Date, none of Seller, Mr. Giuseppe Alfredo Poli, Mr. Alessandro Poli, Mr. Cristiano Poli, Mr. Giancarlo Poli, Mr. Alberto Poli and Mr. Paolo Poli, Mr. Francesco Poli or Mr. Mario Poli shall:
 - (a) directly or indirectly participate, acquire, establish, incorporate or in any way own interests (either in the position of entrepreneur or shareholder, quotaholder, associated, investor, partner and or other similar position) in any entity or activity or business, whether industrial or commercial, which is or may be in competition with the Business;
 - (b) offer employment or otherwise enjoy the services or solicit the services of individuals who have been managers or employees of Seller, the Company and/or the Group Companies before Closing; or
 - (c) solicit the business of any Person that is a customer of the Business (if such solicitation is in competition with the Business) or cause any customer or supplier

of the Business to cease doing business with the Business.

For the purposes of this Section 9.1, competing activity shall mean any activity of developing, manufacturing, distributing and servicing of rail brakes (and spare parts for rail brakes) carried our within the territory of World, excluding South America and Australia.

9.3 Waiver of Liability. Purchaser undertakes not to, and, pursuant to Article 1381 (*Promessa dell'obbligazione o del fatto del terzo*) of the ICC as long as the Company and the Group Companies are controlled – directly or indirectly—by Purchaser or by the Designated Purchaser, undertakes to cause the Company and the Group Companies not to, initiate any action, suit, claim or litigation—based on any applicable Law and on Articles 2393 (*Azione Sociale di Responsabilità*), 2395 (*Azione Individuale del socio e del terzo*) and 2407 (*Responsabilità*) of the ICC – against the members of the board of directors and/or statutory auditors of the Company and of any of the Group Companies who will have resigned from office pursuant to paragraph 5.2(e).

9.4 <u>Tax Matters</u>

9.4.1 <u>Allocation of Tax Liabilities</u>.

- (a) Subject to Section 9.4.2 (*Returns and Reports*), Seller shall bear the economic burden for Taxes that have not been paid at Closing Date of the Company and of any Group Company for: i) any taxable year or period subsequent to tax year 2007, that ends on or before the Closing Date or, for any taxable year or period that includes the Closing Date, that part of the taxable year or period that ends on the Closing Date; and ii) any Taxes relating to the Real Estate Transaction (excluding VAT) in excess of the Estimation of Real Estate Taxes.
- (b) Except as set forth in Sections 8.1(g), 8.1(h) and 9.4.1(a)(ii) above, Purchaser shall be liable for and shall hold Seller harmless against any liability for Taxes of the Company for any taxable year or other taxable period that begins after the close of the Closing Date and, in the case of any taxable year or other taxable period that includes the Closing Date, that part of the taxable year or other taxable period that begins after the close of the Closing Date.
- (c) Within 10 (ten) Business Days after Purchaser has received notice of any pending or threatened audits or assessments relating to Taxes for which Seller would be required to indemnify Purchaser pursuant to Sections 8.1 (*Indemnification obligation of Seller*) or 9.4.1(a) (*Tax borne by Seller*), Purchaser shall notify Seller in writing. Such notice shall provide reasonable details of the claim and the amounts claimed by Purchaser and enclose a copy of all the official documentation issued by the relevant tax authority that is competent for the audit or the assessment. Seller agrees that the Purchaser will be deemed to have automatically satisfied the

notice requirement under this Section 9.4.1(a) or Section 8.1 (*Indemnification obligation of Seller*) in the event that Mr. Giuseppe Alfredo Poli or Mr. Vlado Atanasovski (for so long as he reports directly to Mr. Giuseppe Alfredo Poli) has actual knowledge or notice of any pending or threatened audits or assessments for Taxes in accordance with the above deadline and conditions. Seller shall have the sole right to represent Seller's, Company's or such Group Company's interest in any audit or administrative or court proceeding relating to any such Tax and to employ counsel of its choice at its expense. To this respect, Purchaser shall: (i) appoint and cause the Company and/or the affected Group Company to appoint the counsel(s) designated by Seller and (ii) cooperate and cause the Company and the Group Companies to cooperate with Seller in the preparation for and the prosecution of the defence of the threatened audits or assessments (including, without limitation, making available evidence within the control of the Company and the Group Companies). Notwithstanding the foregoing, Seller shall not be entitled to settle, either administratively or after the commencement of litigation, any claim for such Taxes which would materially adversely affect the liability for such Taxes of Purchaser, the Company or such Group Company for any period after the Closing Date without the prior written consent of Purchaser, which consent shall not be unreasonably withheld. If Seller elects not to assume the defense of any claim for such Taxes, Seller shall be entitled to participate in such defense at its expense. Neither Purchaser, nor the Company nor any Group Company may agree to settle any claim for such Taxes without the prior written consent of Seller, which consent shall not be unreasonably withheld.

- (d) In the event that Seller is liable for any Taxes or Losses under Section 8.1 (*Indemnification obligation of Seller*), Purchaser agrees that it shall not make a claim under Section 9.4.1 (a) (*Taxes borne by Seller*) for the same Taxes and or Losses that would also be a Loss due to a breach of Seller's representation and warranty in Section 7.2.9.
- 9.4.2 Returns and Reports. Seller shall file when due all Tax returns with respect to Taxes that are required to be filed by it or with respect to the Company and any Group Company for taxable years or periods ending on or before the Closing Date and shall pay any Taxes due in respect of such Tax returns. Purchaser shall file when due all Tax returns with respect to Taxes that are required to be filed by or with respect to the Company and any Group Company for taxable years or periods ending after the Closing Date and shall pay any Taxes due in respect of such Tax returns.
- 9.4.3 <u>Cooperation; Access to Records.</u> After the Closing Date, each of Seller and Purchaser shall:
 - (a) assist the other party in preparing any Tax returns or reports which such other party is responsible for preparing and filing in accordance with

- Section 9.4.2 (*Returns and Reports*). With this respect, duty of assistance on Seller shall be confined solely to the filing of Tax returns pertaining to any open tax years until tax year 2008;
- (b) cooperate fully in preparing for and conducting any audits of, or disputes with taxing authorities regarding, any Tax returns of Seller, the Company or any Group Company. With this respect, duty of cooperation on Seller shall be confined solely with regard to audits of or disputes regarding Tax returns pertaining to any open tax years until tax year 2008;
- (c) make available to the other party and to any taxing authority as reasonably requested all records, documents, accounting data and other information relating to Taxes of Seller, the Company or any Group Company;
- (d) furnish the other party with copies of all correspondence received from any taxing authority in connection with any tax audit or information request with respect to any such taxable period for which the other party may have a liability under Sections 8.1 (*Indemnification obligation of Seller*) or 9.4.1 (*Allocation of Tax Liabilities*); and
- (e) execute and deliver such powers of attorney and other documents as are necessary to carry out the intent of this Section 9.4 (*Tax Matters*).

9.4.4 Payments and Refunds.

The Company shall pay the anticipation for the corporate taxes (*Imposta sul Reddito delle Società—IRES and Imposta Regionale sulle Attività Produttive—IRAP*) pertaining to tax year 2008.

Any refunds (including interest thereon) of Taxes paid, indemnified or borne by Seller pursuant to Section 9.4.1(a) (*Taxes borne by Seller*) and 8.1 shall be for the account of Seller. Any refunds (including interest thereon) of Taxes paid or indemnified or borne by Purchaser pursuant to Section 9.4.1(b) (*Tax Indemnification by Purchaser*) shall be for the account of Purchaser. Any amount of Taxes paid by the Company pursuant to this Section 9.4.1.(a) (*Tax borne by Seller*), that is in excess of the amounts in relation to which Seller shall bear the economic burden pursuant to Section 9.4.1.(a), shall be for the account of Seller. Purchaser agrees to promptly remit to Seller all refunds (including interest thereon) of Taxes to which Seller is entitled to hereunder and or which are received by Purchaser, the Company or any Group Company. Seller agrees to promptly remit to Purchaser all refunds (including interest thereon) of Taxes which Purchaser is entitled to hereunder and which are received by Seller.

9.4.6 <u>Survival</u>. Notwithstanding anything in this Agreement to the contrary, the provisions of this Section 9.4 (*Tax Matters*) shall survive through the expiration

of the applicable statute of limitations as the same may be extended.

9.5 Environmental Liability.

- 1) Within the limits provided by this Agreement, Seller shall indemnify and hold harmless the Purchaser in respect of any breach of Section 7.2.19 (*Environmental Matters*) for 3 (three) years from the Closing Date, in accordance with the following Triggering Events, by and within the limits of the Status Quo Ante, if applicable, or within the limitations set forth by Section 8.2 (*Limitation of Seller's Liability*) of this Agreement, if the Status Quo Ante is not applicable because the detected contaminant(s) is (are) not listed therein or because the relevant liability has not been evidenced in the Status Quo Ante itself (collectively, the "Environmental Liability Limits").
- 2) Conditions and events which may trigger Seller's liability, within the Environmental Liability Limits, vis-à-vis the Purchaser (hereinafter referred to as the "Triggering Events") will be:
- (A) an unsolicited order imposed on the Purchaser, the Company or any Group Company's by a Governmental Authority; or
- (B) Purchaser's response to a bona fide Third Party Claim asserting Purchaser's, the Company's or any Group Company's liability for the Release of Hazardous Materials at Camisano (Italy) and/or Skopje (Macedonia) sites, provided in this case, that any relevant investigation and/or sampling are in fact necessary to the Purchaser to defend in the relevant proceedings;
- (C) Purchaser's bona fide compulsory duty to investigate the consequences of an involuntary and not exacerbated contamination, following its discovery after the Closing, where there is obvious, significant and blatant evidence of it (*ictu oculi*).
- 3) For the purposes of points (A), (B) and (C) above, the Parties agree that: (i) where a finding is included in the Status Quo Ante, Seller shall be liable towards the Purchaser up to and within the limits provided therein; (ii) where a finding is not included (because it was not detected and/or discovered prior to the Closing) in the Status Quo Ante or where the relevant liability was not evidenced in the Status Quo Ante itself, Seller shall be liable towards the Purchaser up to and within the limits provided by Section 8.2 (*Limitation of Seller's Liability*) above, provided however, that Purchaser demonstrates and evidences that such finding and/or liability was present on and/or arose in respect of the relevant site (either Camisano or Skopje) prior to the Closing.

For sake of clarity, the amount of liability arising from the Status Quo Ante shall never be greater of, or otherwise exceed, the limit set forth by Section 8.2 (*Limitation of Seller's Liability*).

4) For the purpose of this Section 9.5, should any testing and/or sampling be necessary for the purposes of assessing the respective environmental liability, Seller shall be

anyhow entitled to appoint its environmental experts who have the right to participate in any test and/or sampling. Seller's experts shall be always entitled to make their own samplings and carry out their own independent assessment.

- 5) Seller shall not be obliged to indemnify the Purchaser under Sections 8.1 and 9.5 for any Loss suffered by the Purchaser, the Company or the Group Companies, to the extent that:
- (i) such Losses arise because the Purchaser or any director, employee or agent of them activated procedures and/or volunteered information concerning any actual or potential liability under any Environmental Laws to any Governmental Authority or other third party.
- (ii) such Losses arise because of any redevelopment, at the direction of the Purchaser, of either Camisano (Italy) or Skopje (Macedonia) sites following the Closing Date.
- 6) With specific and exclusive regard to manganese at the Camisano site, and as an express exception to Section 8.2 (*Limitation of Seller Liability*) and this Section 9.5, the Parties expressly agree that Seller shall not be liable to the Purchaser in any way whatsoever in respect of any disclosed and/or undisclosed exceeding of manganese, whether occurred and/or detected prior or after the Closing. Therefore, Purchaser hereby irrevocably waives its rights (including those under Section 8.1 above (*Indemnification obligation of Seller*) in respect of any issue which arises or may potentially arise or is relevant to manganese's presence and/or contamination, at any time and for whichever cause, before or after the Closing, expressly recognizing the manganese's exceeding at Camisano site as a non indemnified issue vis-à-vis the Seller.

ARTICLE 10

Miscellaneous

- 10.1 **Right to Designate**. Pursuant to the Article 1401 (*Riserva di nomina del contraente*) of the ICC, Purchaser may designate one or more companies to purchase the Equity Interest and to pay the Purchase Price according to the terms and conditions of this Agreement, provided that such designation is made in accordance with the following provisions:
 - (i) notwithstanding anything in article 1403 (*Forma e pubblicità*) of the ICC to the contrary, the designation shall be sufficiently made if notified in writing to Seller together with the written acceptance of the designee;
 - (ii) departing from Article 1402 (*Termine e modalità della dichiarazione di nomina*) of the ICC, any designation pursuant hereto shall be notified to Seller, under penalty of forfeiture, no later than 3 (three) Business Days before the Closing Date; and

- (iii) the designee shall be a 100% wholly owned Affiliate, directly or indirectly, of Purchaser.
- Upon designation of the Designated Purchaser in accordance with this Section 10.1 (*Right to Designate*), the Purchaser shall either: (i) provide Seller with appropriate evidence fully satisfactory to Seller of the adequate financial capability of the Designated Purchaser to fulfill the obligations deriving from this Agreement or (ii) remain jointly liable with the Designated Purchaser for any of such obligations.
- 10.2 **Entire Agreement**. This Agreement, the Schedules and Exhibits annexed hereto, and the documents executed and delivered pursuant hereto, constitute the entire agreement between the Parties with respect to the subject matter hereof and supersede all other prior agreements and understandings, both written and oral, between the Parties with respect to the subject matter hereof.
- 10.3 Severability. Should any provision, or provisions, of this Agreement for any reason be, or become, invalid or not capable of performance, in whole or in part, then the validity of the remaining provisions of this Agreement shall not be affected thereby. The same applies if this Agreement should fail to provide for any relevant matter. In lieu of the invalid or inoperable provision, or in order to provide for an omitted provision, this Agreement shall be applied in a reasonable manner, which, so far as legally permissible, comes as close as possible to the application of what the Parties intended or would have intended, according to the spirit and purpose of this Agreement, had they considered the matter.
- 10.4 <u>Assignment</u>. Neither Party may assign any of its rights, interests, obligations under this Agreement without the prior written consent of the other Party, save for assignment of Purchaser to its Affiliates.
- 10.5 **Amendments**. No amendments to this Agreement shall be valid and bounding unless agreed in writing by the Parties or approved in writing by the Party against which such amendment should be enforced.
- 10.6 Notices and Other Communications. All notices, demands or requests provided for or permitted to be given pursuant to this Agreement must be in writing. All notices, demands and requests shall be deemed to have been properly served if given by personal delivery, or if transmitted by facsimile, or if delivered to Federal Express or other reputable overnight carrier for next business day delivery, charges billed to or prepaid by shipper, or if sent by air mail, proper postage prepaid, addressed as follows:

If to Seller:

Polinvest S.r.l. Via Riva Fredda 3 Crema

Attention: Mr. Giuseppe Poli

with copy to:

Labruna, Mazziotti, Segni Via S. Maria Fulcorina No. 13 20123 Milan (Italy) Facsimile No. -39-02-57760400 Attention: Avv. Fabio Labruna

If to Purchaser:

Wabtec Corporation 1001 Airbrake Avenue Wilmerding, PA 15148 USA

Fax: +412-825-1305

Attention: Legal Department

with copy to:

Gianni, Origoni, Grippo & Partners Piazza Belgioiso No. 2 20121 Milan (Italy) Facsimile No. -39-02-76 00 96 28 Attention: Daniel Vonrufs

Reed Smith LLP 435 Sixth Avenue Pittsburgh, PA 15219 412.288.3131 Fax 412.288.3063

Attention: Hannah T. Frank

Each notice, demand or request shall be effective upon personal delivery, or upon confirmation of receipt of the applicable facsimile or 1 (one) Business Day after delivery to a reputable overnight carrier in accordance with the foregoing, or upon arrival at the recipient's address if sent by air mail in accordance with the foregoing. Rejection or other refusal to accept or the inability to deliver because of changed address of which no notice was given shall not adversely impact the effectiveness of any such notice, demand or request. Service by personal delivery upon Purchaser shall be valid only if delivered personally to an officer of Purchaser.

Any addressee may change its address for notices hereunder by giving written notice in accordance with this Section 10.6 (*Notices and other communications*).

10.7 **Counterparts**. This Agreement may be executed in any number of counterparts, and each counterpart shall constitute an original instrument, but all such separate counterparts

shall constitute one and the same agreement.

10.8 **Termination**.

This Agreement may be terminated:

- (a) by the mutual written consent of the Parties;
- (b) by Seller or Purchaser if the Closing does not occur by July 31, 2008, and provided further that if the Closing has not occurred by July 31, 2008, due solely to the failure of the condition set forth in Section 4.1.(c) (*Serbia Antitrust Clearance*) not been met or waived with respect to obtaining the consent or authorization from the Serbian Antitrust Authority and such condition has been diligently pursued by each Party, then, such date shall be extended to October 31, 2008; <u>provided</u>, however, that the right to terminate this Agreement under this Section 10.8 (*Termination*) shall not be available to any Party whose failure to fulfil any obligation under this Agreement shall have been the cause of, or shall have resulted in, the failure of the Closing to occur prior to such date;
- (c) in the event of termination of this Agreement pursuant to this Section, this Agreement shall forthwith become terminated and there shall be no liability on the part of either Party hereto except (i) for the provisions of Sections 6.5 (*Duty of Confidentiality*), 10.6 (*Notices and Other Communications*), 10.9 (*Governing Law*) or 10.10 (*Arbitration*) which shall survive termination of the Agreement, and (ii) that nothing herein shall relieve either Party from liability for any breach of this Agreement.
- 10.9 Governing Law. The validity, construction and enforceability of this Agreement shall be governed in all respects by the Laws of Italy.
- 10.10 **Arbitration**. Unless settled by mutual agreement, any dispute arising in connection with the performance or interpretation of this Agreement shall be finally settled by arbitration in accordance with Italian substantive Law and to the Rules of Arbitration of the Chamber of Commerce of Milan (Italy)—which the Parties acknowledge to know and accept—by three arbitrators appointed pursuant to such Rules. The arbitration shall be of a formal nature ("arbitrato rituale") and the award shall be made in accordance with the Laws of the Republic of Italy. The place of arbitration shall be Milan (Italy) and the proceedings shall be conducted in the English language; however, the parties shall be entitled to submit any document, pleading, deed and documentation to the arbitration panel and to the other party in the Italian language. The expenses of the arbitration proceedings referred in this Section 10.10 (Arbitration) shall be borne by the Parties in accordance with the applicable determinations of the arbitration tribunal.
- 10.11 Costs. Unless otherwise provided in this Agreement, each party hereof shall bear the costs and expenses of its own legal counsel, advisors and consultants retained in connection with the negotiations of this Agreement and the consummation of the transactions contemplated hereby. Stamp duty, registration, transfer, sales, use or similar taxes applicable to, imposed upon or arising as a result of the purchase and sale of the

Equity Interest pursuant to this Agreement shall be borne by Purchaser.

perso	IN WITNESS WHEREOF, each of Purchaser and Seller bonally, all on or as of the day and year first above written.	nas caused this Agreement to	nt to be executed on its behalf by its officer thereunto duly authoriz	ed o
	STINGHOUSE AIR BRAKE HNOLOGIES CORPORATION	POL	POLI S.p.A.	
By:	/s/ R. MARK COX	By:	By: /s/ Giuseppe Alfredo Poli	
Its:	Vice President, Corporate Development	Its:	ts: Sole Director	
Sole	ly for the purpose of being bound by Section 9.1 (Non compe	etition) of this Agreement:	t:	
By:	/s/ GIUSEPPE ALFREDO POLI			
3	Mr. Giuseppe Alfredo Poli			
	••			
By:	/s/ Alberto Poli			
	Mr. Alessandro Poli			
By:	/s/ Alberto Poli			
	Mr. Cristiano Poli			
Bv:	/s/ GIANCARLO POLI			
Dy.	Mr. Giancarlo Poli			
	ivii. Gialicario i dii			
By:	/s/ Alberto Poli			
5	Mr. Alberto Poli			
By:	/s/ Alberto Poli			
	Mr. Paolo Poli			

By: /s/ ALBERTO POLI
Mr. Francesco Poli

By: /s/ ALBERTO POLI
Mr. Mario Poli

CERTIFICATION

- I, Albert J. Neupaver, certify that:
- 1. I have reviewed this quarterly report on Form 10-Q of Westinghouse Air Brake Technologies Corporation.
- 2. Based on my knowledge, this quarterly report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this quarterly report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this quarterly report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this quarterly report;
- 4. The registrant's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and we have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this quarterly report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this quarterly report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this quarterly report based on such evaluation; and
 - (d) Disclosed in this quarterly report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officers and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 8, 2008

Title:

By: /s/ ALBERT J. NEUPAVER
Name: Albert J. Neupaver

President, Chief Executive Officer and Director

CERTIFICATION

- I, Alvaro Garcia-Tunon, certify that:
- 1. I have reviewed this quarterly report on Form 10-Q of Westinghouse Air Brake Technologies Corporation.
- 2. Based on my knowledge, this quarterly report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this quarterly report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this quarterly report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this quarterly report;
- 4. The registrant's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and we have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this quarterly report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this quarterly report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this quarterly report based on such evaluation; and
 - (d) Disclosed in this quarterly report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officers and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 8, 2008

By: /s/ ALVARO GARCIA-TUNON

Name: Alvaro Garcia-Tunon
Title: Senior Vice President,
Chief Financial Officer and Secretary

CERTIFICATION

Pursuant to 18 U.S.C. § 1350, the undersigned officers of Westinghouse Air Brake Technologies Corporation (the "*Company*"), hereby certify, to the best of their knowledge, that the Company's Quarterly Report on Form 10-Q for the quarter ended June 30, 2008 (the "*Report*") fully complies with the requirements of Section 13(a) or 15(d), as applicable, of the Securities Exchange Act of 1934 and that the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

By: /s/ ALBERT J. NEUPAVER

Albert J. Neupaver
President, Chief Executive Officer and Director

Date: August 8, 2008

By: /s/ ALVARO GARCIA-TUNON

Alvaro Garcia-Tunon,
Senior Vice President,
Chief Financial Officer and Secretary

Date: August 8, 2008